

### Testing for Parallel Carryover Effects in Redundant Systems

by

© Yongho Lim

A thesis submitted to the School of Graduate Studies in partial fulfillment of the requirements for the degree of Master of Science.

Department of Mathematics and Statistics Memorial University

October 2016

St. John's, Newfoundland and Labrador, Canada

### Abstract

Redundancy is an important approach to increase the reliability and availability of systems. There has been a recent interest in analyzing failure data from redundant systems to detect the effects of adverse events. A carryover effect is defined as an effect which may cause a temporary increase in the event intensity after the occurrence of a condition or an event. We consider a parallel type of carryover effect in which the event intensity of a process is temporarily increased after event occurrences in other processes. The main goal of this thesis is to develop formal tests for the assessment of parallel carryover effects in redundant systems with repairable components connected in parallel. We, therefore, develop partial score tests for the presence of parallel carryover effects, and discuss their asymptotic properties analytically as well as through simulations. A data set based on the information obtained from a power company is analyzed to illustrate the methods developed.

To Yuna

## Acknowledgements

First and above all, I would like to thank God for giving me this opportunity.

I am sincerely grateful to my supervisor Dr. Candemir Cigsar for his thoughtful guidance and constant patience. He never hesitated in supporting and helping me. He has broadened my eyes to area of recurrent analysis. Without his profound and insightful guidance, I would not be able to complete this thesis.

I would like to thank my thesis committee members Dr. Zhaozhi Fan and Dr. JC Loredo-Osti for their valuable comments, recommendations and suggestions.

Finally, I wish to give special thank to my beautiful wife Yuna Oh for her support and love. I would not be able to complete this thesis without her constant support, patience and love.

# Statement of contribution

Dr. Candemir Cigsar proposed the research question that was investigated throughout this thesis. Dr. Candemir Cigsar and Yongho Lim jointly designed the study. Yongho Lim implemented the algorithms, conducted the simulation study, and drafted the manuscript. Dr. Candemir Cigsar supervised the study and contributed to the final manuscript.

# Table of contents

Ti	tle p	age	i
Al	bstra	$\mathbf{ct}$	ii
A	cknov	wledgements	iv
$\mathbf{St}$	atem	nent of contribution	v
Ta	able o	of contents	vi
$\mathbf{Li}$	st of	tables	ix
$\mathbf{Li}$	st of	figures	xii
1	$\operatorname{Intr}$	oduction	1
	1.1	Reliability of Power Systems and Redundancy	1
		1.1.1 Example: Diesel Plants of a Power Company	3
	1.2	Main Goal and Outline of the Thesis	4
<b>2</b>	Bas	ic Concepts and Models	6
	2.1	Basic Concepts and Models	6
		2.1.1 Covariates in Recurrent Event Processes	11
	2.2	Likelihood Methods and Score Tests	13
	2.3	Simulation Procedures	17
		2.3.1 The Use of the Simulations	18
3	$\operatorname{Red}$	lundant Systems with Two Components	20
	3.1	Introduction	20
	3.2	Models and Tests for Parallel Carryover Effects	21

		3.2.1 Case 1: $m = 1, \Delta_A = 0 \dots 23$
		3.2.2 Case 2: $m > 1, \Delta_A = 0 \dots 2$
		3.2.3 Case 3: $m = 1, \Delta_A > 0 \dots 2$
		3.2.4 Case 4: $m > 1, \Delta_A > 0 \dots \dots$
	3.3	Asymptotic Properties of Test Statistics
	3.4	Simulation Studies
4	Rec	lundant Systems with Three Components 55
	4.1	Introduction
	4.2	Models and Tests for Parallel Carryover Effects
		4.2.1 Case 1: $m = 1, \Delta_A = 0, \Delta_B = 0, \Delta_C > 0 \dots \dots$
		4.2.2 Case 2: $m = 1, \Delta_A = 0, \Delta_B > 0, \Delta_C > 0 \dots \dots$
		4.2.3 Case 3: $m = 1, \Delta_A > 0, \Delta_B > 0, \Delta_C > 0 \dots \dots$
		4.2.4 Case 4: $m > 1, \Delta_A > 0, \Delta_B > 0, \Delta_C > 0$
	4.3	Simulation Studies
	4.4	Application
<b>5</b>	Rec	lundant Systems with Trends and Covariates 87
	5.1	Introduction
	5.2	Models and Tests for Parallel Carryover Effects with Trends 88
		5.2.1 Case 1: $m = 1, \Delta_A = 0, \Delta_B > 0 \dots \dots$
		5.2.2 Case 2: $m > 1, \Delta_A = 0, \Delta_B > 0 \dots \dots$
		5.2.3 Case 3: $m = 1, \Delta_A > 0, \Delta_B > 0 \dots \dots$
		5.2.4 Case 4: $m > 1, \Delta_A > 0, \Delta_B > 0 \dots \dots \dots \dots \dots \dots \dots \dots 98$
		5.2.5 Case 5: $m = 1, \Delta_A > 0, \Delta_B > 0, \Delta_C > 0 \dots \dots$
	5.3	Simulation Studies
	5.4	Redundant Systems with Trends and Covariates
	5.5	Application
6	Sun	nmary and Future Research 12
	6.1	Summary and Conclusions
	6.2	Alternating Two-State Processes
	6.3	Redundant Systems with Imperfect Repairs
Bi	ibliog	graphy 129

Α	Data	a Sets																									1	132	2
	A.1	Data 1					•			•	•		•				•	•	•									132	2
	A.2	Data 2			•				•	•	•	•	•		•	•	•	•	•	•	•	 • •	•	•		•	•	133	}
в	Mis	specific	ati	on	0	f ⊿	$\Delta_B$	3																			1	L <b>3</b> 4	Ł

# List of tables

3.1	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of <i>Z</i> computed from 10,000 samples when $m = 1$ . $\hat{Pr}(Z > Q_p)$ is the proportion of the values of <i>Z</i> in 10,000 samples which are larger than the <i>p</i> th quantile of a standard normal	
	distribution	. 42
3.2	Power of Z : $m = 1, \Delta_A = 0$	. 43
3.3	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of <i>Z</i> computed from 10,000 samples when $m > 1$ and $\tau = 100$ . $\hat{Pr}(Z > Q_p)$ is the proportion of the values of <i>Z</i> in 10,000 samples which are larger than the <i>p</i> th quantile of a	
	standard normal distribution	46
3.4	Power of Z : $m > 1, \Delta_A = 0$	. 47
3.5	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of <i>Z</i> computed from 10,000 samples when $m = 1$ and $\Delta_A > 0$ . $\hat{Pr}(Z > Q_p)$ is the proportion of the values of <i>Z</i> in 10,000 samples which are larger than the <i>p</i> th quantile of a	•
	standard normal distribution	. 50
3.6	Power of Z : $m = 1, \Delta_A > 0$	. 51
3.7	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of <i>Z</i> computed from 10,000 samples when $m > 1$ , $\Delta_A = 1$ and $\tau = 100$ . $\hat{Pr}(Z > Q_p)$ is the proportion of the values of <i>Z</i> in 10,000 samples which are larger than the <i>p</i> th quantile	
	of a standard normal distribution	. 54
3.8	Power of Z : $m > 1, \Delta_A = 1$	. 54
4.1	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of <i>Z</i> computed from 10,000 samples when $m = 1$ , $\Delta_A = 0$ , $\Delta_B > 0$ , $\Delta_C > 0$ . $\hat{Pr}(Z > Q_p)$ is the proportion of the values of <i>Z</i> in 10,000 samples which are larger than the <i>p</i> th	
	quantile of a standard normal distribution	. 72
4.2	Power of Z : $m = 1, \Delta_A = 0, \Delta_B > 0, \Delta_C > 0$	. 72

4.3	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of <i>Z</i> computed from 10,000 samples when $m = 1$ and $\hat{Q}_p = 1$ , $\hat{Q}_p = \hat{Q}_p = \hat$	
	when $m = 1$ and $\Delta_A = 1$ , $\Delta_B > 0$ , $\Delta_C > 0$ . $Pr(Z > Q_p)$ is the properties of the values of Z in 10,000 samples which are larger than	
	the <i>w</i> th quantile of a standard normal distribution	76
1 1	The pth quantile of a standard normal distribution $\dots \dots \dots \dots$	. 76
4.4 4.5	Fower of Z: $m = 1$ , $\Delta_A = 1$ , $\Delta_B > 0$ , $\Delta_C > 0$ $\hat{Q}_p$ is the empirical <i>p</i> th quantile of Z computed from 10,000 samples	. 70
	when $m > 1$ . $\hat{Pr}(Z > Q_p)$ is the proportion of the values of Z in 10,000 samples which are larger than the <i>p</i> th quantile of a standard normal	
	distribution	. 79
4.6	Power of Z : $m > 1$ , $\Delta_A = 1$ , $\Delta_B > 0$ , $\Delta_C > 0$	. 80
4.7	Estimates of $\alpha_A$ , $\beta_A$ and $\alpha_A$ when $\beta_A = 0$ . The numbers in parentheses	
	are the standard errors	. 85
4.8	The test statistic $Z$ , $\ell(\hat{\boldsymbol{\theta}})$ and $\ell(\tilde{\boldsymbol{\theta}})$	. 85
4.9	Estimates of $\alpha_A$ , $\beta_A$ , $\gamma$ and Wald type statistic $W$ and $p$ -value. The	
	numbers in parentheses are the standard errors	. 86
5.1	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of Z computed from 10,000 samples	
	when m=1. $\hat{Pr}(Z > Q_p)$ is the proportion of the values of Z in 10,000	
	samples which are larger than the $p$ th quantile of a standard normal	
	distribution $\ldots$	. 106
5.2	Power of Z : $m = 1, \Delta_A > 0, \Delta_B > 0$	. 107
5.3	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of Z computed from 10,000 samples	
	when $m > 1$ , $\Delta_A = 1$ , and $\tau = 100$ . $\hat{Pr}(Z > Q_p)$ is the proportion of	
	the values of $Z$ in 10,000 samples which are larger than the $p{\rm th}$ quantile	
	of a standard normal distribution $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	. 111
5.4	Power of Z : $m > 1$ , $\Delta_A = 1$ , $\Delta_B > 0$	. 111
5.5	$\hat{Q}_p$ is the empirical <i>p</i> th quantile of Z computed from 10,000 samples	
	when m=1. $\hat{Pr}(Z > Q_p)$ is the proportion of the values of Z in 10,000	
	samples which are larger than the $p$ th quantile of a standard normal	
	distribution	. 115
5.6	Power of Z : $m = 1, \Delta_A = 1$ with Stochastic Aging	. 115
5.7	Estimates of $\alpha_{Ai}$ , $\beta_{Ai}$ , $\gamma_i$ and $\alpha_{Ai}$ , $\gamma_i$ when $\beta_{Ai} = 0$ where $i = 3, 5$ ,	
	9, and 15. The numbers in parentheses are the standard errors of the	
	estimates	. 123
5.8	Statistic $Z$ and $p$ -values $\ldots \ldots \ldots$	. 123

5.9	Statistic LA and p-values	. 124
A.1	Data set used in Section 4.5, times (in days) are failure times in each component, repair times, $\Delta$ , are the same among the components	
	within a city	. 133
B.1	Results of the power study when $\Delta_{B_0}$ is misspecified, where $\Delta_B = 7$ is assumed	134
B.2	Results of the power study when $\Delta_{B_0}$ is misspecified, where $\Delta_B = 14$	. 101
	is assumed	. 135
B.3	Results of the power study when $\Delta_{B_0}$ is misspecified, where $\Delta_B = 7$ is	
	assumed $\ldots$	. 135
B.4	Results of the power study when $\Delta_{B_0}$ is misspecified, where $\Delta_B = 14$	
	is assumed	. 136

# List of figures

2.1	A system with two components (Component A and Component B)	
	connected in parallel	7
3.1	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4) $\tau = 1000$ .	40
3.2	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 3$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4) $\tau = 1000$ .	41
3.3	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 7$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4) $\tau = 1000$ .	41
3.4	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 14$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4) $\tau = 1000$	42
3.5	Normal Q-Q plots of 10,000 simulated values of the test statistic $Z$	
	when $\tau = 100, \Delta_B = 1$ , and (1) $m = 10$ , (2) $m = 20$ , (3) $m = 50$ , (4)	
	$m = 100 \dots \dots$	44
3.6	Normal Q-Q plots of 10,000 simulated values of the test statistic $Z$	
	when $\tau = 100, \Delta_B = 3$ , and (1) $m = 10$ , (2) $m = 20$ , (3) $m = 50$ , (4)	
	$m = 100 \dots \dots$	45
3.7	Normal Q-Q plots of 10,000 simulated values of the test statistic $Z$	
	when $\tau = 100, \Delta_B = 7$ , and (1) $m = 10$ , (2) $m = 20$ , (3) $m = 50$ , (4)	
0.0	$m = 100 \dots \dots$	45
3.8	Normal Q-Q plots of 10,000 simulated values of the test statistic Z 100  A = 14  1(1)  10(2)  20(2)  50(4)	
	when $\tau = 100, \Delta_B = 14$ , and (1) $m = 10$ , (2) $m = 20$ , (3) $m = 50$ , (4)	10
2.0	$m = 100 \dots \dots$	40
3.9	Normal Q-Q plots of 10,000 simulated values of the test statistic Z	
	when $\Delta_B = 1$ , $\Delta_A > 0$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4)	4.0
	$\tau = 1000 \dots $	48

3.10	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 3$ , $\Delta_A > 0$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4)	
	$\tau = 1000 \dots $	49
3.11	Normal Q-Q plots of 10,000 simulated values of the test statistic $Z$	
	when $\Delta_B = 7$ , $\Delta_A > 0$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4)	
	$\tau = 1000 \dots $	49
3.12	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 14$ , $\Delta_A > 0$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4)	
	$\tau = 1000 \dots $	50
3.13	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\tau = 100, \Delta_A = 1, \Delta_B = 1$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \dots$	52
3.14	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\tau = 100, \Delta_A = 1, \Delta_B = 3$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \$	52
3.15	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\tau = 100, \Delta_A = 1, \Delta_B = 7$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \dots$	53
3.16	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\tau = 100, \Delta_A = 1, \Delta_B = 14$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) m = 100 \dots \dots$	53
4.1	Normal Q-Q plots of 10,000 simulated values of the test statistic $Z$	
	when $\Delta_A = 0$ , $\Delta_B = 1$ , $\Delta_C = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \tau = 1000 \dots $	70
4.2	Normal Q-Q plots of 10,000 simulated values of the test statistic $Z$	
	when $\Delta_A = 0$ , $\Delta_B = 3$ , $\Delta_C = 3$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \tau = 1000 \dots $	70
4.3	Normal Q-Q plots of 10.000 simulated values of the test statistic $Z$	••
	when $\Delta_A = 0$ , $\Delta_B = 7$ , $\Delta_C = 7$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500.$ (4) $\tau = 1000$	71
4.4	Normal Q-Q plots of 10,000 simulated values of the test statistic $Z$	. =
	when $\Delta_A = 0$ , $\Delta_B = 14$ , $\Delta_C = 14$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \ \tau = 1000 \ \dots $	71

4.5	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_A = 1$ , $\Delta_B = 1$ , $\Delta_C = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \tau = 1000 \ldots \ldots$	74
4.6	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_A = 1$ , $\Delta_B = 3$ , $\Delta_C = 3$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \tau = 1000 \ldots \ldots$	74
4.7	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_A = 1$ , $\Delta_B = 7$ , $\Delta_C = 7$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \ \tau = 1000 \ \ldots $	75
4.8	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_A = 1$ , $\Delta_B = 14$ , $\Delta_C = 14$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \ \tau = 1000 \ \dots \dots$	75
4.9	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_A = 1$ , $\Delta_B = 1$ , $\Delta_C = 1$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \dots$	77
4.10	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_A = 1$ , $\Delta_B = 3$ , $\Delta_C = 3$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \dots$	78
4.11	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_A = 1$ , $\Delta_B = 7$ , $\Delta_C = 7$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \dots$	78
4.12	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_A = 1$ , $\Delta_B = 14$ , $\Delta_C = 14$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \dots$	79
4.13	Dot plots of failures of generators in communities	82
4.14	Cumulative failures of community 7	83
4.15	Cumulative failures of community 11	83
4.16	Cumulative failures of community 12	83
4.17	Cumulative failures of community 14	84
5.1	Normal Q-Q plots of 10,000 simulated values of the test statistic Z when $\Delta_B = 1$ , $\Delta_A = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4) $\tau = 1000$	104
		101

5.2	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 3$ , $\Delta_A = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4)	
	$\tau = 1000 \dots $	. 105
5.3	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 7$ , $\Delta_A = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4)	
	$\tau = 1000 \dots $	. 105
5.4	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = 14$ , $\Delta_A = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3) $\tau = 500$ , (4)	
	$\tau = 1000 \dots $	. 106
5.5	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\tau = 100, \Delta_B = 1, \Delta_A = 1$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \dots$	. 109
5.6	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\tau = 100, \Delta_B = 3, \Delta_A = 1, \text{ and } (1) \ m = 10, (2) \ m = 20, (3)$	
	$m = 50, (4) \ m = 100 \ \dots \dots$	. 109
5.7	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\tau = 100, \Delta_B = 7, \Delta_A = 1, \text{ and } (1) \ m = 10, (2) \ m = 20, (3)$	
	$m = 50, (4) \ m = 100 \ \dots \dots$	. 110
5.8	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\tau = 100, \Delta_B = 14, \Delta_A = 1$ , and (1) $m = 10$ , (2) $m = 20$ , (3)	
	$m = 50, (4) \ m = 100 \ \dots \dots$	. 110
5.9	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = \Delta_C = 1$ , $\Delta_A = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \ \tau = 1000 \dots $	. 113
5.10	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = \Delta_C = 3$ , $\Delta_A = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \ \tau = 1000 \dots $	. 113
5.11	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = \Delta_C = 7$ , $\Delta_A = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \ \tau = 1000 \dots $	. 114
5.12	Normal Q-Q plots of 10,000 simulated values of the test statistic ${\cal Z}$	
	when $\Delta_B = \Delta_C = 14$ , $\Delta_A = 1$ , and (1) $\tau = 100$ , (2) $\tau = 200$ , (3)	
	$\tau = 500, (4) \ \tau = 1000 \dots $	. 114

5.13	Dot plots of failures of the redundant generators operating in the com-	
	munity number 3, 5, 9 and 15 $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	120
5.14	Plots of cumulative failures of the redundant generators in the commu-	
	nity number 3, 5, 9 and 15 versus operating time	121

# Chapter 1

### Introduction

In this chapter, our main goal is to introduce some concepts related to the reliability of power systems and research topics. In Section 1.1, we first introduce fundamental concepts in reliability analysis of redundant systems, and next discuss the type of data sets. In Section 1.2, we give an outline of the thesis.

### 1.1 Reliability of Power Systems and Redundancy

The demand of consumers of manufacturers, systems or service providers on quality, productivity and availability of a product, system or service is getting higher than ever. Reliability attached to those characteristics is of critical importance. In basic terms, reliability is defined as the probability that a product, system or service operates under the operating conditions for some specified period of time. Improving reliability is, therefore, a very important issue for manufacturers and service providers to be competitive.

In this thesis, we consider the reliability of systems with multiple components. Our focus is on diesel operated power systems, which are often used to generate electricity in isolated, hard-to-reach communities to meet the power demand, but methods and models developed can be applied in other settings as well. The function of a power system is to supply electrical energy to its consumers on demand as reliably and safely as possible and in an economically justifiable manner (Billinton and Allan, 1984). Power generation companies may sell their power to utility, industrial, residential and commercial customers. The unavailability of a power system when it is needed can create serious problems for its consumers as well as resulting in severe financial loss to a power company. The expectation of customers from a power company is, therefore, a continuous supply of least-cost electricity whenever there is a demand for power. This is not always possible in reality. There are many events causing unwanted system stoppages, which are beyond the control of system engineers. Such events, to which we refer to as *failures*, are often in a recurrent nature. Modelling and analysis of recurrent failures can be useful for identifying opportunities for reliability improvements in power systems especially in the operation phase.

There are different ways of improving the reliability of a system. As Billinton and Allan (1992, Section 1.3) denoted, redundancy is an important approach to affect the reliability and availability of a system. In this approach, there is one or more back-ups of the components in a system so that the function of a failed component is absorbed by other unfailed components. There are two types of redundancy. The first type is called *standby redundancy*, in which the redundant component waits in a standby position until the failure of active components, and starts operating when one or more active components fail. In the second type of redundancy, called *active redundancy*, components operate and share a function together. In case of a failure of a component, remaining components absorb the load of the failed component (Billinton and Allan, 1992). In this thesis, we focus on the active redundancy as it fits a natural model for the type of data sets considered. A system having either standby or active redundancy is called a *redundant system*. An example of a redundant system with active redundancy is a (fully) parallel system with K components, in which all components are connected in parallel. In such systems, a system failure occurs if and only if all of the components fail. This type of systems are sometimes called *fully redundant.* A system is called a *series system*, if all of its components are connected in series. A series system is sometimes called *zero redundant*. There are also partially redundant systems, in which some components are connected in series and some components are connected in parallel. In this thesis, we focus on fully redundant systems consisting of K parallely connected components.

Systems or their components can be classified as repairable or nonrepairable (Rigdon and Basu, 2000). A *nonrepairable system* is a system that is discarded when a failure occurs. Cost of replacement is usually low in nonrepairable systems. A *repairable system* is a system that can be restored to an operating condition after some repair other than replacement of the entire system when a failure occurs. As discussed by Rigdon and Basu (2000), repairable systems are generally more complex than nonrepairable systems. Our focus in this thesis is on systems with repairable components. Statistical analysis of repairable systems usually needs assumptions on the nature of repairs, which plays an important role on the development of a model for failure data. Ascher (1968) introduces the concept of a minimal repair, in which a repair is assumed to bring a failed system or component to the operating condition just like before the failure. In this case, a repair is sometimes called *as-bad-as-old* (Ascher, 1968). A *perfect repair* is a repair that brings the condition of a repairable system to that of a completely new system after a failure. This type of repair is sometimes called *as-good-as-new* (Rigdon, 2007; Misra, 2012). As Lindqvist (2006) denoted, the assumption of a minimal repair is usually applied when only a minor part of the system is repaired or replaced after a failure. This type of repairs are better suited in our study, so we assume that a minimal repair occurs after each failure of components of a system. In Section 2.1, we introduce models corresponding to the assumption of minimal and perfect repairs. There are also *imperfect repair* models (Brown and Proschan, 1983; Baker, 2001; Lindqvist, 2006). In this thesis, we do not discuss such repairs, but state them as a future work in the final chapter.

In the next two subsections, we first introduce the motivating example of this research and then present the main target and outline of this thesis. It is well known that there are major problems in reliability when it comes to data collection (Lawless, 1983). For example, failure data are often incomplete or biased in the sense that not all types of failures are reported. We do not consider such difficulties in this thesis.

#### **1.1.1** Example: Diesel Plants of a Power Company

The framework of the data sets used in this thesis are obtained from a power company. Unexpected power outages in the remote, isolated communities in different regions is a major concern for this power company and its customers. There are 25 diesel plants operating in those isolated locations to provide electricity. Environmental conditions in these regions are usually extremely severe, especially in the winter times. Also, some of the power plants are 35-40 years old. Mainly because of these reasons, power plants operating in these communities fail frequently, and require increasing attention for maintenance, refurbishment and replacement. Most communities that receive their electricity from diesel plants have an operator living in the community who can respond to unexpected power outages that may occur on site and complete regular maintenance. Therefore, a failed power plant can be sometimes repaired in a short time. However, there are many occasions when a failed power plant requires long repair times, which may substantially increase the total cost of operation and create unpleasant situations for customers.

To improve the reliability of operations, the power company implements the redundancy approach. Most of the power plants operating in isolated communities have multiple diesel operated engines working together in order to supply electricity demand of a community. These engines are parallely connected to share the electricity demand. If one of the engines fails, the remaining engines in operating conditions share the load of the failed one and operate in an increased capacity. The power plant has an unwanted system stoppage if and only if all of the engines do not operate. This framework is, therefore, suitable for the analysis within the fully redundant systems. In the remainder of this thesis, we consider a power plant as a system and its engines as parallely connected components operating under redundancy.

### **1.2** Main Goal and Outline of the Thesis

Modelling and detection of the effects of adverse events on repairable systems has been a major research area in the analysis of reliability data. Redundancy is an important approach to increase the reliability. In redundant systems with active redundancy, the failure of a component may temporarily increase the probability of failures of other components. We refer to this phenomenon as a *parallel carryover effect*. In other words, a parallel carryover effect is a temporary adverse effect resulting in an increased risk of failures in redundant components of an active redundant system during the downtime of the failed components. The presence of parallel carryover effects causes temporary clustering of events (failures) together in the redundant components.

The main goal of this thesis is to provide a thorough discussion of parallel carryover effects, develop formal tests for the absence of them in various settings, and investigate their properties. The outcome of this thesis can be beneficial for power generation companies to improve their reliability programs and to determine their maintenance programs, which minimize the total cost of operation and maximize the availability of repairable systems.

The remainder of this thesis is arranged as follows. In Chapter 2, we introduce the notation used in the thesis, fundamental models for recurrent event processes as well as mathematical concepts and simulation procedures applied in the following chapters.

In Chapter 3, we discuss testing for the presence of parallel carryover effects in redundant systems with two components. We first consider testing for parallel carryover effects in redundant systems where repair times of the redundant component are negligible. Then, we consider testing for parallel carryover effects in redundant systems where repair times of the redundant component are not negligible. Asymptotic properties of the test statistics are discussed analytically as well as through simulations in two different setups; (i) when the observation period increases for a single system, and (ii) when the number of systems increases for multiple systems.

In Chapter 4, we discuss testing for parallel carryover effects in redundant systems with three components as an extension of Chapter 3. We first consider testing for parallel carryover effects in redundant systems with three components when repair times of the redundant component are negligible. Next, we discuss the same issue when repair times of the redundant component are not negligible. We discuss the asymptotic properties of test statistics within the same setups of Chapter 3 via simulations. Finally, we illustrate the methods developed in Chapter 4 by analyzing a simulated data set. The simulation of the data set is based on the information received from a power company on their diesel power plants.

In Chapter 5, we discuss testing for the presence of parallel carryover effects in redundant systems, which are subject to monotonically increasing time trends in the rate of event occurrences due to stochastic aging. This type of trends is often seen in repairable systems. In this chapter, we also extend our methodology to deal with external covariates in the models. We first consider systems with two components and then extended the methods to three components case. We also discuss the asymptotic properties of test statistics through simulations. Finally, we analyze a simulated failure data set in power systems to illustrate the methods.

In the final chapter, we give a summary of the results of the previous chapters and present our conclusions. Some important future research topics are also discussed.

### Chapter 2

### **Basic Concepts and Models**

In this chapter, we introduce basic concepts and models that are useful in this thesis. In Section 2.1, we introduce basic terminology and concepts in recurrent event processes. We focus mostly on the Poisson processes. In Section 2.2, we give the methodology based on the likelihood functions to test a hypothesis with score tests. We explain simulation procedures in Section 2.3.

### 2.1 Basic Concepts and Models

In this section, we introduce the basic notation and terminology frequently used in the remaining part of this thesis. We introduce the notation for a single process observed over a fixed time interval. Extensions to multiple processes and other observation schemes are given later in the thesis whenever it is needed.

As defined in the previous section, reliability is the probability that a product, system or service operates under the operating conditions for some specified period of time (Meeker and Escobar, 1998). In this thesis, we focus on systems with multiple components, in which components are parallely connected. Figure 2.1 shows the diagram of such a system with two parallelly connected components. We consider only binary components with operational (up) and nonoperational (down) states, in which either up or down state is possible for any component at any given time instant. We define any event that results in an unplanned stoppage in a component as a failure in the corresponding component. Furthermore, a system is in down state if and only if all of its components are down. We assume components are repairable and provide possibly recurrent failure (event) data.



Figure 2.1: A system with two components (Component A and Component B) connected in parallel.

The statistical analysis of repairable systems is usually constructed within the point process framework. Let random variables  $T_1, T_2, \ldots$ , with the property that  $0 < T_1 < T_2 < \cdots$  denote the occurrence times of a well-defined event along a time axis. The  $T_i$  are called event times or arrival times. We then define the *j*th gap time  $W_j$  as the interarrival time between the *j*th and *j* + 1st event times; that is,  $W_j = T_j - T_{j-1}, j = 1, 2, \ldots$ , where by mathematical convention  $T_0 = 0$ . We also let  $t_1, t_2, \ldots$  and  $w_1, w_2, \ldots$  denote realizations of  $T_1, T_2, \ldots$  and  $W_1, W_2, \ldots$ , respectively.

Let the random variable N(t) denote the number of event occurrences over a time interval (0, t], where t > 0. We also let N(s, t) denote the number of event occurrences over (s, t] so that N(s, t) = N(t) - N(s) for all  $0 \le s < t < \infty$ . We assume that N(0) = 0 and  $E\{N(t)\} < \infty$  for each t, where E denotes expectation. The stochastic process  $\{N(t), t > 0\}$  is then called a *counting process*. Many properties of counting processes and their related functions are given, for example, in Daley and Vere-Jones (2003).

We next define the intensity function of a counting process, but to do this we first define the history of a stochastic process. We let  $\mathcal{H}(t) = \{N(u), 0 \leq u < t\}$ denote the history of the process  $\{N(t), t > 0\}$  at time t. The history  $\mathcal{H}(t)$  includes all information about the counting process  $\{N(t), t > 0\}$  from time 0 to just prior to time t. In our simple setup, this information includes event occurrence times  $t_1$ ,  $t_2$ , ..., and the number of events at each time point in [0, t). More information on histories of stochastic processes can be found in Daley and Vere-Jones (2003, pp. 423–427). We let dt denote an infinitesimal positive valued real number and dN(t) be an infinitesimal increment in N(t); that is,  $dN(t) = N((t + dt)^-) - N(t^-)$ , which gives the number of events in [t, t + dt). We are now in a position to define the intensity function. Let  $\lambda(t|\mathcal{H}(t))$  denote the intensity function of a counting process  $\{N(t), t > 0\}$  (with respect to its history  $\mathcal{H}(t)$ ), which is mathematically defined as

$$\lambda(t|\mathcal{H}(t)) = \lim_{\Delta t \to 0} \frac{\Pr\left\{N((t+\Delta t)^{-}) - N(t^{-}) = 1 \,|\, \mathcal{H}(t)\right\}}{\Delta t},\tag{2.1}$$

where  $\Delta t > 0$ . The intensity function gives the instantaneous probability of an event occurring in [t, t + dt), given the process history  $\mathcal{H}(t)$ . Assuming that two or more events cannot occur together at the same instant, the intensity function completely specifies a recurrent event process (Cook and Lawless, 2007). In this case, since dN(t) is a 0-1 valued (binary) random variable, it can be shown that  $\lambda(t | \mathcal{H}(t)) dt = E\{dN(t) | \mathcal{H}(t)\}.$ 

Other important concepts include mean, rate and hazard functions. The mean function, denoted by  $\mu(t)$ , is a nondecreasing, right continuous function which gives the expected number of events up to time t; that is,  $\mu(t) = E\{N(t)\}$  for t > 0. We also use the notation  $\mu(s,t)$  to denote the expected number of events in any finite interval (s,t], where  $0 \le s < t < \infty$ . Thus,  $\mu(s,t) = E\{N(s,t)\}$ . We let  $\rho(t)$  denote the instantaneous rate of change of the expected number of events with respect to time. We refer to  $\rho(t)$  as the rate of occurrence of failures function (ROCOF) or, simply, the rate function. Since rate and mean functions are not conditioned on the history, they are called marginal properties of a point process. Assuming the derivative of  $\mu(t)$ exists for t > 0, by definition,  $\rho(t) = (d/dt)\mu(t)$ . The hazard function of a positive random variable W is defined by

$$h(w) = \lim_{\Delta t \to 0} \frac{\Pr\{W < w + \Delta t \,|\, W > w\}}{\Delta t}, \qquad w > 0.$$
(2.2)

If f(w), w > 0, denotes the probability density function (p.d.f.) of W and  $F(w) = \Pr\{W \le w\} = \int_0^w f(u) \, du$  is the cumulative density function (c.d.f.) of W, it is well known that h(w) = f(w)/[1 - F(w)] for w > 0.

The data generating mechanism of a counting process  $\{N(t), t > 0\}$  in the continuous case is governed by its intensity function. The specification of the intensity function (2.1), therefore, defines a statistical model for recurrent event processes. We now introduce some fundamental models for recurrent event processes through the specification of their intensity functions. We start with Poisson processes and then introduce renewal processes and some ramifications of them.

Poisson processes are usually useful if there is an interest in modelling the number of event occurrences. The process  $\{N(t), t > 0\}$  is called a Poisson process if its associated intensity function is given by

$$\lambda(t \mid \mathcal{H}(t)) = \rho(t), \qquad t > 0, \tag{2.3}$$

where  $\rho(t)$  is the rate function of the process (Cook and Lawless, 2007). It is clear from (2.3) that the intensity function of a Poisson process depends on t, but it is independent from the previous event occurrences over [0, t). Therefore, Poisson processes have the Markov property (see, e.g., Thompson, 1988). A Poisson process  $\{N(t), t > 0\}$  is called a *homogeneous Poisson process* (HPP) if the intensity function (2.3) is constant for any t > 0. Otherwise, it is called a *nonhomogeneous Poisson process* (NHPP).

In a reliability context, Poisson processes are canonical models for repairable systems if minimal repairs are applied after after each failure. The distinction between a HPP and a NHPP is an important modelling issue. As denoted by Thompson (1988, p. 22), homogeneous Poisson processes (HPPs) are often used for modelling recurrent event data mainly because of their simple properties, but their adequacy should be always investigated especially in reliability studies of repairable systems. For example, since HPPs have a constant rate function, they are not appropriate models when there is a time trend due to stochastic aging (Lai and Xie, 2006). This is a major limitation of their use in the reliability studies of repairable systems because many repairable systems are more prone to fail as they age. A constant rate function cannot be adequate in such cases. Nonetheless, HPPs have some applications in reliability studies, in particular, when the observation periods of systems are short. A NHPP is a canonical model for repairable system if there is stochastic aging in the system due to a wear-out phenomenon or due to reliability growth (Thompson, 1988, p. 53; Lai and Xie, 2006, p. 7).

Poisson processes and their generalizations are well discussed in the literature. Their properties can be found in many stochastic processes or point processes texts (see, e.g., Thompson, 1988; Kingman, 1993; Daley and Vere-Jones, 2003; Nakagawa, 2011). Here, we state only some of the useful properties, and refer to literature for their proofs. For example, let  $\{N(t), t > 0\}$  be a Poisson process with the intensity function (or equivalently, the rate function)  $\rho(t)$ , where t > 0. Then, the number of events in any finite interval (s, t], where  $0 \le s < t < \infty$ , has a Poisson distribution with the mean  $\mu(s, t)$  (Daley and Vere-Jones, 2003, p. 34). We have, therefore,

$$\Pr\{N(s,t)=n\} = \frac{[\mu(s,t)]^n}{n!} e^{-\mu(s,t)}, \qquad n=0,1,2,\dots,$$
(2.4)

where  $\mu(s,t) = \int_{s}^{t} \rho(u) du$ . Let V(t) denote the variance function. Since the mean and the variance of any Poisson distributed random variable are equal, the variance function of the number of events in a Poisson process  $\{N(t), t > 0\}$  is given by  $V(t) = Var\{N(t)\}$ , where Var stands for the variance and  $Var\{N(t)\} = \mu(t)$  for t > 0.

The following property is useful to simulate realizations of a HPP. The counting process  $\{N(t), t > 0\}$  is a HPP with a constant rate function  $\rho$ , where  $\rho > 0$ , if and only if the gap times  $W_j$ ,  $j = 1, 2, \ldots$ , are independent and identically distributed (i.i.d.) exponential variables with mean  $\rho^{-1}$ . A proof of this statement can be found, for example, in Rigdon and Basu (2000, pp. 45–49). Another important result which can be used to simulate realizations of a NHPP is given as follows. Let  $\{N(t), t > 0\}$ be a NHPP with mean function  $\mu(t)$  and  $\{N^*(s), s > 0\}$  be a HPP with mean function  $\mu^* = 1$ . By letting  $s = \mu(t)$ , we can show that  $N^*(s) = N(\mu^{-1}(s))$  for s > 0 (Daley and Vere-Jones, 2003, p. 258). We used these two results in simulation and data analysis sections of the next chapters. We explain the simulation procedure in Section 1.4.

Another important class of models for recurrent event processes can be based on *renewal processes*. A renewal process  $\{N(t), t > 0\}$  is a point process in which the gap times  $W_1, W_2, \ldots$ , are i.i.d. In this case, the intensity function (2.1) of  $\{N(t), t > 0\}$  takes the form of

$$\lambda(t \mid \mathcal{H}(t)) = h(t - t_{N(t^{-})}), \qquad t > 0,$$
(2.5)

where h is the hazard function defined in (2.2). In a reliability context, a renewal process implies that there is a perfect (i.e., as-good-as-new) repair after each failure of a repairable system, which brings the system to a brand new condition. In some cases, this can be a reasonable assumption; for example, if a complete overhaul is performed after each failure of a repairable system. However, the assumption of

i.i.d. gap times is a very strong one, and its validity needs to be carefully checked in each application. Many basic properties of the renewal processes can be found in stochastic processes texts. For example, see Nakagawa (2011, Chapter 3). As discussed by Thompson (1988, Section 5.2), a renewal process cannot model systems that is wearing out. Therefore, we do not provide a detailed background on renewal processes. However, one important relation is that, if the gap times  $W_j$ , j = 1, 2, ...,of renewal process  $\{N(t), t > 0\}$  are i.i.d. exponential random variables with mean  $E\{W_j\} = \rho^{-1}$ , where  $0 < \rho < \infty$ , then the process  $\{N(t), t > 0\}$  is a HPP with rate function  $\rho$ .

#### 2.1.1 Covariates in Recurrent Event Processes

In many studies, there are covariates of interest. In such cases, models can be extended to include covariates. An excellent discussion of this issue is given by Kalbfleisch and Prentice (2002); also, see Andersen et al. (1996), Daley and Vere-Jones (2003) and Cook and Lawless (2007). The basic idea is to consider the covariates as a vector of stochastic processes and then extend the history by including their path information. Following the notation of Daley and Vere-Jones (2003, pp. 237–238) and Cook and Lawless (2007, Section 2.2.2), this can be done as follows.

Suppose that we observe a *p*-dimensional vector of stochastic processes denoted by  $\{X(t), t > 0\}$ , where  $\{X(t), t > 0\} = \{X_1(t), \ldots, X_p(t); 0 < t < \infty\}$ . Let  $\mathcal{H}^X(t)$ denote the history of the process X(t) over the time interval [0, t]. Thus,  $\mathcal{H}^X(t)$ includes paths of covariate processes  $X_j(t)$ ,  $j = 1, \ldots, p$ , in [0,t]. Now suppose that  $\{N(t), 0 < t\}$  is a counting process with the intensity function  $\lambda_0(t \mid \mathcal{H}^N(t))$ , where  $\mathcal{H}^N(t) = \{N(u), 0 \le u < t\}$  is the history of the counting process. Then, a model of multiplicative form is given by the following intensity function.

$$\lambda(t|\mathcal{H}(t)) = \lambda_0(t|\mathcal{H}^N(t))\,\psi(X_1(t),\dots,X_p(t)), \qquad t > 0, \tag{2.6}$$

where  $\lambda_0$  is called the baseline intensity function,  $\psi$  is a nonnegative valued function and  $\mathcal{H}(t)$  is the extended history including information on both  $\mathcal{H}^N(t)$  and  $\mathcal{H}^X(t)$ . If we specify  $\lambda_0(t \mid \mathcal{H}^N(t)) = \rho(t)$  and  $\log \psi(X_1, \ldots, X_p) = \sum_{j=1}^p \beta_j X_j$  in (2.6), we obtain the intensity function of the multiplicative form

$$\rho(t) \exp[\beta_1 X_1(t) + \dots + \beta_p X_p(t)], \quad t > 0,$$
(2.7)

where  $\rho(t)$  is the rate function of  $\{N(t), t > 0\}$  and the  $\beta_j$  are regression parameters. The model (2.7) is called the *modulated Poisson process* (Cook and Lawless, 2007).

Instead of  $\rho(t)$ , if we specify the baseline rate function with  $h(t-t_{N(t^{-})})$ , we obtain the modulated renewal process with the intensity function of the form

$$h(t - t_{N(t^{-})}) \exp[\beta_1 X_1(t) + \dots + \beta_p X_p(t)], \quad t > 0,$$
 (2.8)

where h is the hazard function defined in (2.2). The model (2.8) is discussed by Cox (1972).

It should be noted that multiplicative models of the from (2.6) specify multiplicative effects of covariates on the intensity function. The validity of this assumption should be checked. Some methods for checking this assumption are discussed by Cook and Lawless (2007, Section 3.7.2). As an alternative to multiplicative models, additive models can also be used (Aalen et al., 2008). In this case, the general intensity function can be written as

$$\lambda(t|\mathcal{H}(t)) = \lambda_0(t|\mathcal{H}^N(t)) + \psi(X_1(t), \dots, X_p(t)), \qquad t > 0.$$
(2.9)

There is an important remark regarding to the inference with models involving covariates in recurrent event processes. The full likelihood based inference requires that the evolution of the covariate processes  $\{X_j(t), t > 0\}, j = 1, ..., p$ , should be independent from the counting process  $\{N(t), t > 0\}$ . Kalbfleisch and Prentice (2003, p. 196–198) refer to such covariates as *external*, which means, in their words, that the covariate process  $\{X(u), 0 \leq u \leq t\}$  may influence the probabilistic characteristics of event occurrences over time, but its future path up to any time t, where t > u, is not affected by the occurrence of an event at time u. In this case, as explained in the next section, a full likelihood approach can be based on the models of the multiplicative form (2.6). If a covariate is not external, the likelihood function should be considered for both  $\{X(t), t > 0\}$  and  $\{N(t), t > 0\}$  together. In general, such a likelihood function is too complicated and the treatment of covariates requires care. Unless otherwise stated, we restrict the discussion in this thesis only to external covariates, in which their values are known at time t and probability laws do not include the parameters in the event generating model under study. All the models and probabilities are conditional on the values of the covariates. For notational purposes, we use the notation  $\mathcal{H}(t) = \{N(u), X(s); 0 \le u < t, 0 \le s \le t\}$  to denote the history of the processes  $\{N(t), t > 0\}$  and  $\{X(t), t > 0\}$ .

### 2.2 Likelihood Methods and Score Tests

In this section, we give the likelihood function for a recurrent event process and develop partial score test procedures for testing composite hypothesis. Suppose that  $\{N(t), t > 0\}$  is a counting process with the associated intensity function  $\lambda(t|\mathcal{H}(t))$ . The likelihood function for the outcome that n events occur at times  $0 < t_1 < t_2 < \cdots < t_n < \tau$  in the time interval  $(0, \tau]$ , given the history of the process  $\mathcal{H}(t)$ , is of the form (Cook and Lawless, 2007, p. 30)

$$L(\boldsymbol{\theta}) = \prod_{i=1}^{n} \lambda(t_i | \mathcal{H}(t_i)) \exp\left\{-\int_0^\tau \lambda(s | \mathcal{H}(s)) \, ds\right\},\tag{2.10}$$

where  $\boldsymbol{\theta} = (\theta_1, \dots, \theta_p)'$  is a  $p \times 1$  vector of parameters. Let  $\hat{\boldsymbol{\theta}}$  be the maximum likelihood estimator of  $\boldsymbol{\theta}$  which maximizes  $L(\boldsymbol{\theta})$  and let  $\ell(\boldsymbol{\theta})$  be the log likelihood function; that is,  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$ . Let  $\boldsymbol{U}(\boldsymbol{\theta}) = (U_1(\boldsymbol{\theta}), \dots, U_p(\boldsymbol{\theta}))'$  be the  $p \times 1$  score vector with entries

$$U_j(\boldsymbol{\theta}) = \frac{\partial \ell(\boldsymbol{\theta})}{\partial \theta_j}, \qquad j = 1, \dots, p,$$
(2.11)

Usually the maximum likelihood estimator  $\hat{\boldsymbol{\theta}}$  can be obtained by solving  $\boldsymbol{U}(\boldsymbol{\theta}) = \boldsymbol{0}$ where  $\boldsymbol{0}$  is a  $p \times 1$  vector of zeros. Let  $\boldsymbol{I}(\boldsymbol{\theta})$  be the  $p \times p$  information matrix where the entries of  $\boldsymbol{I}(\boldsymbol{\theta})$  are defined by

$$I_{ij}(\boldsymbol{\theta}) = -\frac{\partial^2 \ell(\boldsymbol{\theta})}{\partial \theta_i \partial \theta_j}, \qquad i, j = 1, \dots, p,$$
(2.12)

Also, let  $J(\theta)$  be the  $p \times p$  expected information matrix with entries

$$J_{ij}(\boldsymbol{\theta}) = E\left(-\frac{\partial^2 \ell(\boldsymbol{\theta})}{\partial \theta_i \partial \theta_j}\right), \qquad i, j = 1, \dots, p.$$
(2.13)

Under mild regularity conditions,  $E(U(\theta)) = 0$  and variance-covariance matrix of  $U(\theta)$  is  $J(\theta)$ .

Assuming that the model is a regular model and inverse of  $J(\theta)$  exists, a test

statistic for testing  $H_0: \boldsymbol{\theta} = \boldsymbol{\theta}_0$  is of the form

$$\boldsymbol{U}(\boldsymbol{\theta}_{\mathbf{0}})' \boldsymbol{J}^{-1}(\boldsymbol{\theta}_{\mathbf{0}}) \boldsymbol{U}(\boldsymbol{\theta}_{\mathbf{0}})$$
(2.14)

which is called a score statistic and the test based on (2.14) is called a score test. Under regularity conditions, the test statistic (2.14) is asymptotically chi-squared distributed with p degrees of freedom under  $H_0$ , (Boos, 1992).

If we are interested in only a part of the parameters in  $\boldsymbol{\theta}$ , we can make a partition of it as  $\boldsymbol{\theta} = (\boldsymbol{\alpha}', \boldsymbol{\beta}')'$  where  $\boldsymbol{\alpha}$  is  $k \times 1$  vector of nuisance parameters and  $\boldsymbol{\beta}$  is  $q \times 1$ vector of parameters of interest. Then  $\boldsymbol{U}(\boldsymbol{\theta})$  can be partitioned accordingly into two parts denoted by  $\boldsymbol{U}_{\alpha}(\boldsymbol{\theta})$  and  $\boldsymbol{U}_{\beta}(\boldsymbol{\theta})$  where  $\boldsymbol{U}_{\alpha}(\boldsymbol{\theta})$  is a  $k \times 1$  vector of score functions with entries  $\boldsymbol{U}_{\alpha_j}(\boldsymbol{\theta}) = \partial \ell(\boldsymbol{\theta}) / \partial \alpha_j$ ,  $j = 1, \ldots, k$ , and  $\boldsymbol{U}_{\beta}(\boldsymbol{\theta})$  is a  $q \times 1$  vector of score functions with entries  $\boldsymbol{U}_{\beta_j}(\boldsymbol{\theta}) = \partial \ell(\boldsymbol{\theta}) / \partial \beta_j$ ,  $j = 1, \ldots, q$ . Similarly,  $\boldsymbol{J}(\boldsymbol{\theta})$  can be partitioned as follows.

$$\boldsymbol{J}(\boldsymbol{\theta}) = \begin{pmatrix} \boldsymbol{J}_{\alpha\alpha}(\boldsymbol{\theta}) & \boldsymbol{J}_{\alpha\beta}(\boldsymbol{\theta}) \\ \boldsymbol{J}_{\beta\alpha}(\boldsymbol{\theta}) & \boldsymbol{J}_{\beta\beta}(\boldsymbol{\theta}) \end{pmatrix}, \qquad (2.15)$$

where  $J_{\alpha\alpha}(\theta)$  is a  $k \times k$  matrix,  $J_{\alpha\beta}(\theta) = J_{\beta\alpha}(\theta)'$  is a  $k \times q$  matrix and  $J_{\beta\beta}(\theta)$  is a  $q \times q$  matrix. Observed information matrix  $I(\theta)$  can be partitioned in the same way. Assuming they exist, the inverse matrix of (2.15) can be written as

$$\boldsymbol{J}(\boldsymbol{\theta})^{-1} = \begin{pmatrix} \boldsymbol{J}^{\alpha\alpha}(\boldsymbol{\theta}) & \boldsymbol{J}^{\alpha\beta}(\boldsymbol{\theta}) \\ \boldsymbol{J}^{\beta\alpha}(\boldsymbol{\theta}) & \boldsymbol{J}^{\beta\beta}(\boldsymbol{\theta}) \end{pmatrix}.$$
 (2.16)

Let  $\boldsymbol{\beta} = \boldsymbol{\beta}_0$ , where  $\boldsymbol{\beta}_0$  is specified value of  $\boldsymbol{\beta}$ . An estimator of  $\boldsymbol{\alpha}$  under this specified case can be found by maximizing  $L(\boldsymbol{\alpha}, \boldsymbol{\beta}_0)$  or  $\ell(\boldsymbol{\alpha}, \boldsymbol{\beta}_0)$ . Such an estimator is called a restricted maximum likelihood estimator of  $\boldsymbol{\alpha}$  and denoted by  $\boldsymbol{\alpha}(\boldsymbol{\beta}_0)$  and if  $\boldsymbol{\beta}_0 = 0$ , then shortly  $\tilde{\boldsymbol{\alpha}}$ . In this case, the function  $L(\boldsymbol{\alpha}, \boldsymbol{\beta}_0)$  is called profile likelihood function for  $\boldsymbol{\beta}$  and  $\ell(\boldsymbol{\alpha}, \boldsymbol{\beta}_0)$  is called profile log likelihood function for  $\boldsymbol{\beta}$ . If we denote that  $\tilde{\boldsymbol{\theta}}_0 = (\tilde{\boldsymbol{\alpha}}(\boldsymbol{\beta}_0), \boldsymbol{\beta}_0)$  then under  $H_0 : \boldsymbol{\beta} = \boldsymbol{\beta}_0$ ,

$$\boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_{0})' \boldsymbol{J}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_{0}) \boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_{0})$$
(2.17)

is asymptotically a chi-squared distributed with q degrees of freedom. A test of  $H_0$ :  $\boldsymbol{\beta} = \boldsymbol{\beta}_0$  based on (2.17) is called *a partial score test*. We can replace  $\boldsymbol{J}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_0)$  with a consistent estimator of  $\boldsymbol{J}^{\beta\beta}(\boldsymbol{\theta}_0)$ , because this will also give the same asymptotic result (Boos, 1992).

Suppose that  $\{N(t), t > 0\}$  is a counting process and the associated intensity function is given by

$$\lambda(t|\mathcal{H}(t);\boldsymbol{\theta}) = \rho(t;\boldsymbol{\alpha}) \exp\{\boldsymbol{X}(t)'\boldsymbol{\beta}\}, \quad t > 0,$$
(2.18)

where  $\mathbf{X}(t)' = (X_1(t), \dots, X_q(t))'$  is a  $q \times 1$  vector,  $\rho$  is a baseline intensity function, and  $\boldsymbol{\alpha} = (\alpha_1, \dots, \alpha_k)'$  is a vector of unknown nuisance parameters, and  $\boldsymbol{\beta} = (\beta_1, \dots, \beta_q)'$  is a vector of unknown regression parameters that is of interest. Then the likelihood function is given as follows.

$$L(\boldsymbol{\theta}) = \prod_{i=1}^{n} \rho(t_i; \boldsymbol{\alpha}) \exp\{\boldsymbol{X}(t_i)'\boldsymbol{\beta}\} \exp\left\{-\int_0^{\tau} \rho(s; \boldsymbol{\alpha}) \exp\{\boldsymbol{X}(s)'\boldsymbol{\beta}\} ds\right\}.$$
 (2.19)

The likelihood (2.19) is a partial likelihood and it is discussed by Cook and Lawless (2007, pp. 47–49). The log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$\ell(\boldsymbol{\theta}) = \sum_{i=1}^{n} \log \rho(t_i; \boldsymbol{\alpha}) + \sum_{i=1}^{n} \boldsymbol{X}(t_i)' \boldsymbol{\beta} - \int_0^{\tau} \rho(s; \boldsymbol{\alpha}) \exp\{\boldsymbol{X}(s)' \boldsymbol{\beta}\} ds.$$
(2.20)

Then the score vector is  $\boldsymbol{U}(\boldsymbol{\theta}) = (\boldsymbol{U}_{\alpha}(\boldsymbol{\theta})', \boldsymbol{U}_{\beta}(\boldsymbol{\theta})')'$ , where  $\boldsymbol{U}_{\alpha}(\boldsymbol{\theta})$  is a  $k \times 1$  vector of score functions with entries  $\boldsymbol{U}_{\alpha_l}(\boldsymbol{\theta}) = \partial \ell(\boldsymbol{\theta}) / \partial \alpha_l$ ,  $l = 1, \ldots, k$ , and  $\boldsymbol{U}_{\beta}(\boldsymbol{\theta})$  is a  $q \times 1$ vector of score functions with entries  $\boldsymbol{U}_{\beta_j}(\boldsymbol{\theta}) = \partial \ell(\boldsymbol{\theta}) / \partial \beta_j$ ,  $j = 1, \ldots, q$ . Therefore, the components  $\boldsymbol{U}_{\alpha_l}(\boldsymbol{\theta})$  and  $\boldsymbol{U}_{\beta_j}(\boldsymbol{\theta})$  are given as follows.

$$\boldsymbol{U}_{\alpha_{l}}(\boldsymbol{\theta}) = \sum_{i=1}^{n} \left( \frac{\partial}{\partial \alpha_{l}} \log \rho(t_{i}; \boldsymbol{\alpha}) \right) - \int_{0}^{\tau} \left( \frac{\partial}{\partial \alpha_{l}} \rho(s; \boldsymbol{\alpha}) \right) \exp\{\boldsymbol{X}(s)' \boldsymbol{\beta}\} ds, \qquad (2.21)$$

and

$$\boldsymbol{U}_{\beta_j}(\boldsymbol{\theta}) = \sum_{i=1}^n X_j(t_i) - \int_0^\tau \rho(s; \boldsymbol{\alpha}) X_j(s) \exp\{\boldsymbol{X}(s)'\boldsymbol{\beta}\} ds.$$
(2.22)

The observed information matrix is partitioned as follows.

$$\boldsymbol{I}(\boldsymbol{\theta}) = \begin{pmatrix} \boldsymbol{I}_{\alpha\alpha}(\boldsymbol{\theta}) & \boldsymbol{I}_{\alpha\beta}(\boldsymbol{\theta}) \\ \boldsymbol{I}_{\beta\alpha}(\boldsymbol{\theta}) & \boldsymbol{I}_{\beta\beta}(\boldsymbol{\theta}) \end{pmatrix}, \qquad (2.23)$$

where  $I_{\alpha\alpha}(\theta)$  is a  $k \times k$  matrix with entries  $I_{\alpha_u\alpha_v} = -(\partial^2/\partial\alpha_u\partial\alpha_v)\ell(\theta), u, v = 1, \ldots, k$ ,

so that

$$I_{\alpha_u \alpha_v}(\boldsymbol{\theta}) = -\sum_{i=1}^n \left( \frac{\partial^2}{\partial \alpha_u \partial \alpha_v} \log \rho(t_i; \boldsymbol{\alpha}) \right) - \int_0^\tau \left( \frac{\partial^2}{\partial \alpha_u \partial \alpha_v} \rho(s; \boldsymbol{\alpha}) \right) \exp\{\boldsymbol{X}(s)' \boldsymbol{\beta}\} ds,$$
(2.24)

 $I_{\alpha\beta}(\boldsymbol{\theta}) = I_{\beta\alpha}(\boldsymbol{\theta})'$  is a  $k \times q$  matrix with entries  $I_{\alpha_u\beta_v} = -(\partial^2/\partial\alpha_u\partial\beta_v)\ell(\boldsymbol{\theta}), u = 1..., k, v = 1, ..., q$ , so that

$$I_{\alpha_{u}\beta_{v}}(\boldsymbol{\theta}) = \int_{0}^{\tau} X_{v}(s) \left(\frac{\partial}{\partial \alpha_{u}} \rho(s; \boldsymbol{\alpha})\right) \exp\{\boldsymbol{X}(s)'\boldsymbol{\beta}\} ds, \qquad (2.25)$$

 $I_{\beta\beta}(\boldsymbol{\theta})$  is a  $q \times q$  matrix with entries  $I_{\beta_u\beta_v} = -(\partial^2/\partial\beta_u\partial\beta_v)\ell(\boldsymbol{\theta}), u, v = 1, \dots, q$ , is given by

$$I_{\beta_{u}\beta_{v}}(\boldsymbol{\theta}) = \int_{0}^{\tau} \rho(s;\boldsymbol{\alpha}) X_{u}(s) X_{v}(s) \exp\{\boldsymbol{X}(s)'\boldsymbol{\beta}\} ds.$$
(2.26)

For testing the null hypothesis  $H_0: \boldsymbol{\beta} = \boldsymbol{\beta}_0$  against the alternative hypothesis  $H_a$ :  $\boldsymbol{\beta} \neq \boldsymbol{\beta}_0$ , a partial score test can be used to test  $H_0$ . If we let  $\boldsymbol{\theta}_0 = (\boldsymbol{\alpha}'_0, \boldsymbol{\beta}'_0)'$  be a  $p \times 1$  vector where  $\boldsymbol{\alpha}_0$  is the true value of  $\boldsymbol{\alpha}$  and  $\boldsymbol{\beta}_0$  is the value of  $\boldsymbol{\beta}$  under the null hypothesis. If we let  $\boldsymbol{\beta}_0 = 0$  so that we want to test  $H_0: \boldsymbol{\beta} = \mathbf{0}$ , then  $\boldsymbol{\theta}_0 = (\boldsymbol{\alpha}'_0, \mathbf{0}')'$ . Let  $\tilde{\boldsymbol{\alpha}}$  be a restricted maximum likelihood function of  $\boldsymbol{\alpha}$  under  $H_0: \boldsymbol{\beta} = \mathbf{0}$ , then the score statistic for testing  $H_0: \boldsymbol{\beta} = \mathbf{0}$  is of the form

$$\boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_{0})' \boldsymbol{J}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_{0}) \boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_{0}), \qquad (2.27)$$

where  $\boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_0)$  is a  $q \times 1$  vector of score functions with entries  $\boldsymbol{U}_{\beta_j}(\tilde{\boldsymbol{\theta}}_0), j = 1, \ldots, q$ ; that is,

$$\boldsymbol{U}_{\beta_j}(\tilde{\boldsymbol{\theta}}_0) = \sum_{i=1}^n X_j'(t_i) - \int_0^\tau \rho(s; \tilde{\boldsymbol{\alpha}}) X_j(s) ds, \qquad (2.28)$$

and  $q \times q$  matrix  $\boldsymbol{J}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_0)$  is given by

$$\left\{ \boldsymbol{J}_{\beta\beta}(\tilde{\boldsymbol{\theta}}_0) - \boldsymbol{J}_{\beta\alpha}(\tilde{\boldsymbol{\theta}}_0) \boldsymbol{J}_{\alpha\alpha}(\tilde{\boldsymbol{\theta}}_0)^{-1} \boldsymbol{J}_{\alpha\beta}(\tilde{\boldsymbol{\theta}}_0) \right\}^{-1}.$$
 (2.29)

Replacing  $\boldsymbol{J}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_0)$  with  $\boldsymbol{I}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_0)$  in (2.27) gives the same asymptotic results.

#### 2.3 Simulation Procedures

In this section, we explain the simulation procedures used in the subsequent chapters of this thesis. This simulation procedures can be used (i) to study the null distribution of the test statistic, (ii) to study the distribution of the test statistic under the alternative hypothesis and also obtain the power, (iii) to obtain the *p*-value.

A key result for generating a realization of a counting process is given in the following. Its proof can be found, e.g., in Cook and Lawless (2007, p. 30). For a counting process  $\{N(t), t > 0\}$ , with the associated intensity function  $\lambda(t|\mathcal{H}(t))$ ,

$$\Pr\{N(s,t) = 0 | \mathcal{H}(s^+)\} = \exp\left(-\int_s^t \lambda(u|\mathcal{H}(u))du\right), \qquad (2.30)$$

where in the exponential term  $\mathcal{H}(u)$  = { $\mathcal{H}(s^+)$ , N(s, u) = 0}. Therefore, we can show that  $\Pr\{N(t_{j-1}, t_{j-1} + w) = 0 | \mathcal{H}(t_{j-1}^+)\} = \exp\{-\int_{t_{j-1}}^{t_{j-1}+w} \lambda(u|\mathcal{H}(u))du\}.$ 

The events " $N(t_{j-1}, t_{j-1} + w) = 0 |\mathcal{H}(t_{j-1}^+)|$ " and " $W_j > w |T_{j-1} = t_{j-1}, \mathcal{H}(t_{j-1})$ " are equivalent almost surely. Therefore,

$$\Pr\{W_j > w | T_{j-1} = t_{j-1}, \mathcal{H}(t_{j-1})\} = \exp\left(-\int_{t_{j-1}}^{t_{j-1}+w} \lambda(u|\mathcal{H}(u))du\right).$$
(2.31)

Now, if we let  $E_j = \int_{t_{j-1}}^{t_{j-1}+W_j} \lambda(u|\mathcal{H}(u)) du$  where j = 1, 2, ..., from (2.31) the random variable  $E_j$  has an exponential distribution with mean 1, given  $t_{j-1}, t_0 = 0$  and  $\mathcal{H}(t)$  (Cook and Lawless, 2007, p. 44). Therefore,  $U_j = \exp\left(-\int_{t_{j-1}}^{t_{j-1}+W_j} \lambda(u|\mathcal{H}(u)) du\right) = \exp\{-E_j\}$  has a uniform distribution on (0, 1). Then we can obtain each event  $T_j$  by solving an equation  $E_j = \int_{t_{j-1}}^{t_{j-1}+W_j} \lambda(u|\mathcal{H}(u)) du$  for  $W_j$ . The simulation algorithm given for this procedure is given below:

- 1. Set  $j = 1, t_0 = 0$ .
- 2. Generate  $U_j$  from the standard uniform distribution.
- 3. Obtain  $E_j$  by transforming  $\exp\{-E_j\} = U_j$ , so that  $E_j = -\log(U_j)$ .
- 4. Obtain  $w_j$  by solving an equation  $E_j = \int_{t_{j-1}}^{t_{j-1}+W_j} \lambda(u|\mathcal{H}(u)) du$  for  $W_j$ .

5. If  $T_j \leq \tau$ , then increase j by 1 and go to the step 2, otherwise stop the loop.

In generating a HPP with the rate function  $\rho$ , steps 2-4 give  $W_J = -\log(U_j)/\rho$ , where  $j = 1, 2, \ldots$  In generation of a NHPP with the rate function  $\rho(t)$ , mean function  $\mu(t)$  can be generated by using  $N^*(s) = N(\mu^{-1}(s))$ , where  $\{N^*(s), s > 0\}$ is a HPP with mean function  $\mu^* = 1$  (see Section 1.2). Therefore, in generation of NHPP, steps 2-4 give HPP with rate 1 as  $\mu(T_j) = \mu(t_{j-1}) + E_j$ , where  $j = 1, 2, \ldots$ and  $T_j = t_{j-1} + W_j$  is the *j*th event time. Then  $T_j = \mu^{-1}(\mu(t_{j-1}) + E_j)$  gives the *j*th event time for NHPP, where  $\mu^{-1}$  is the inverse transformation of  $\mu$ .

The above algorithm has been used and recorded widely in the literature (see, e.g., Lewis and Shedler, 1976; Daley and Vare-Jones, 1988; Cook and Lawless, 2007). It should be noted that the integral in step 4 may not have a closed form. In such cases, numerical methods can be applied to obtain  $W_i = w_i$ .

#### 2.3.1 The Use of the Simulations

We are able to discuss the distributions of the statistics and obtain p-values by generating data under null and alternative hypotheses. For example, suppose that  $\{N(t), t > 0\}$  is a counting process with the intensity function  $\lambda(t|\mathcal{H}(t)) = \rho(t; \boldsymbol{\alpha}) \exp\{\boldsymbol{X}(t)'\boldsymbol{\beta}\}$ , and testing following composite hypotheses is of interests.

$$H_0: \boldsymbol{\beta} = \mathbf{0}, \boldsymbol{\alpha} \in \mathbb{R}^k$$
 vs.  $H_a: \boldsymbol{\beta} \neq \mathbf{0}, \boldsymbol{\alpha} \in \mathbb{R}^k$  (2.32)

We generate *B* realizations of recurrent event processes under the null hypothesis to assess the asymptotic distribution of a statistic. For each realization, we obtain  $\tilde{\boldsymbol{\theta}}_0 = (\tilde{\boldsymbol{\alpha}}', \tilde{\boldsymbol{0}}')'$ , the partial score vector  $\boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_0)'$  and the matrix  $\boldsymbol{J}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_0)$ , where  $\tilde{\boldsymbol{\theta}}_0$  is an estimate of  $\boldsymbol{\theta}$  under the null hypothesis in (2.32). Then we can use these to study the distribution of a score test statistic, under the null hypothesis.

Similarly, the power of the test can be obtained by simulation. We let the power function of the test of hypothesis (2.32) as  $P(\boldsymbol{\beta}_{a}) = \Pr\{\text{reject } H_{0} | \boldsymbol{\beta} = \boldsymbol{\beta}_{a}\}$ . We can obtain the power by data generation under the alternative hypothesis in (2.32) and by obtaining  $\hat{\boldsymbol{\theta}} = (\hat{\boldsymbol{\alpha}}', \hat{\boldsymbol{\beta}}')'$ , the partial score vector  $\boldsymbol{U}_{\beta}(\hat{\boldsymbol{\theta}})'$  and the matrix  $\boldsymbol{J}^{\beta\beta}(\hat{\boldsymbol{\theta}})$ , where  $\hat{\boldsymbol{\theta}}$  is an estimate of  $\boldsymbol{\theta}$  under the alternative hypothesis in (2.32).

We can obtain p-value by simulation. We first generate B realizations of recurrent

event processes under the null hypothesis. Then we calculate the partial score statistic  $Z_i$ , i = 1, ..., B. Then *p*-value can be estimated by obtaining

$$\frac{\sum_{i=1}^{B} I(Z_i > Z_{test})}{B},\tag{2.33}$$

where  $Z_{test}$  is the test statistic based on the given data set. Under the null hypothesis in (2.32),  $\boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_{0})' \boldsymbol{J}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_{0}) \boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_{0})$  is asymptotically chi-squared distributed with q degrees freedom. Then p-value can be also obtained by  $Pr\{\chi_{q}^{2} \geq \boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_{0})' \boldsymbol{J}^{\beta\beta}(\tilde{\boldsymbol{\theta}}_{0}) \boldsymbol{U}_{\beta}(\tilde{\boldsymbol{\theta}}_{0})\}$ , where  $\tilde{\boldsymbol{\theta}}_{0}$  is an estimate of  $\boldsymbol{\theta}$  under the null hypothesis in (2.32).

### Chapter 3

# Redundant Systems with Two Components

In this chapter, we consider a system which includes two components; a primary component and a redundant component working in parallel. Our goal is to develop a formal method to assess whether or not there is an adverse effect of repairs of a failed component on the redundant component. We therefore develop partial score tests and discuss their asymptotic properties analytically and through simulations.

### 3.1 Introduction

As discussed previously, systems consisting of parallely connected components can be seen in many industrial settings. Some examples include light bulbs in an automobile, batteries in a laptop computer, computer server nodes and so forth. In repairable systems settings, components of a system are subject to repairs and repair times may not be negligible. For example, in fully redundant systems, the remaining components share the duty of a failed component during its downtimes. This type of downtimes due to unwanted failures in components may have some temporary residual effects resulting in an increased risk of failures in the redundant components of a system during the downtime of the failed components. We refer to such adverse effects as parallel carryover effects. If a parallel carryover effect is significant, the system may not perform well, and the cost of operation can be considerably high. Therefore, there is a room for improving the reliability of a system by detecting parallel carryover effects.
In this section, we develop tests for the presence of parallel carryover effects, and discuss their asymptotic properties. To this end, we consider partial score tests. Partial score tests provide a convenient way of testing for the presence of a parallel carryover effect because they do not require to obtain the maximum likelihood estimates of the parameters under the alternative hypothesis.

# 3.2 Models and Tests for Parallel Carryover Effects

In this section, we first extend the notation introduced in the previous chapter to the "two components" case. Then, we discuss models for a single system and multiple systems under two different settings according to the duration of repairs in the redundant component.

Let  $\{N_1(t), N_2(t), \ldots, N_K(t); t \ge 0\}$  be a K-variate process. If each  $\{N_i(t); t \ge 0\}$ ,  $i = 1, \ldots, K$ , is a counting process and no two or more of the processes jump at the same time, the K-variate process  $\{N_1(t), N_2(t), \ldots, N_K(t); t \ge 0\}$  is called a *multivariate counting process* (Fleming and Harrington, 1991). In this case, we let  $\mathcal{H}(t)$  denote the history of the multivariate counting process at time t, where we assume that  $\mathcal{H}(t)$  includes all information on the event times and the number of events of each counting process  $\{N_i(t), t \ge 0\}$   $(i = 1, \ldots, K)$  in [0, t).

Now, suppose that there is a bivariate counting process  $\{N_A(t), N_B(t); t \ge 0\}$ , where  $\{N_A(t), t \ge 0\}$  is a counting process for Component A and  $\{N_B(t), t \ge 0\}$  is a counting process for Component B in a system of two parallely connected components. We let  $t_{A1}, t_{A2}, \ldots$ , where  $0 < t_{A1} < t_{A2} < \cdots$ , and  $t_{B1}, t_{B2}, \ldots$ , where  $0 < t_{B1} < t_{B2} < \cdots$ , denote the failure times of Components A and B, respectively. The components are subject to repairs and repair times cannot be ignored. Let  $\Delta_A$  and  $\Delta_B$  denote the repair times of Components A and B, respectively. In other words, if Component A fails, for example, we assume that the repair takes  $\Delta_A$  time units. Similarly, if Component B fails, it takes  $\Delta_B$  time units to repair it. We also need to define at-risk indicators. For K = A, B, the function  $Y_K(t)$  is called *at-risk indicator* of the process  $\{N_K(t), t > 0\}$ , which takes the value of 1 when Component K is up and the process  $\{N_K(t), t > 0\}$  is under observation; otherwise, it is equal to 0. For example, if at time t Component A is up and under observation and Component B is down, then  $Y_A(t) = 1$  and  $Y_B(t) = 0$ . A model including parallel carryover effects in Component A is given by

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \,\alpha_A \,\exp\{\beta_A \, X_A(t)\}, \qquad t > 0, \tag{3.1}$$

where  $Y_A(t)$  is the at-risk function of Component A,  $\alpha_A > 0$  is a baseline rate function, and

$$X_A(t) = I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \le \Delta_B\},$$
(3.2)

and  $\beta_A$  is a regression parameter. The intensity function (3.1) increases from  $\alpha_A$  to  $\alpha_A \exp{\{\beta_A\}}$  at each failure time of Component B for  $\Delta_B$  time units. After  $\Delta_B$  time units, it reduces to  $\alpha_A$ . This behaviour is what we refer to as a parallel carryover effect. In the remaining part of this section, we develop formal tests for the presence of such effects. It should be noted that  $\Delta_B$  defines the duration of a carryover effect period in Component A. We therefore call  $\Delta_B$  the carryover effect period in Component A. In this study, we assume that carryover effect periods are constant. We discuss issues related to the choice of the carryover effect periods in the final chapter.

A model for parallel carryover effects can also be similarly defined for Component B. In this case, the intensity function of  $\{N_B(t), t \ge 0\}$  is given by

$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \,\alpha_B \,\exp\{\beta_B \,X_B(t)\}, \qquad t > 0, \tag{3.3}$$

where  $Y_B(t)$  is the at-risk function of Component B,  $\alpha_B > 0$  is a baseline rate function, and

$$X_B(t) = I\{N_A(t^-) > 0\}I\{t - t_{AN_A(t^-)} \le \Delta_A\}.$$
(3.4)

Similarly, parallel carryover effects in Component B can be investigated through model (3.3) with (3.4).

It should be noted that the models (3.1) and (3.3) can be equivalently written in a simple form as follows. For K, J = A, B and  $K \neq J$ ,

$$\lambda_K(t|\mathcal{H}(t)) = Y_K(t) \exp\{\beta_K(1 - Y_J(t))\}, \quad t > 0.$$
(3.5)

From (3.5), it is easy to see that the system is down if and only if both components are down; that is,  $Y_A(t) = 0$  and  $Y_B(t) = 0$  at time t. In the following development in this chapter, we mostly focus on the models of the former types because they provide explicit relation to the duration of repairs.

Let m denote the number of systems in a study, each with two components. In the

next subsections, we develop partial score tests for the presence of parallel carryover effects in four different cases; (i) a single system is under observation and repair times of Component A are negligible (m = 1,  $\Delta_A = 0$ ), (ii) multiple systems are under observation and repair times of Component A are negligible (m > 1,  $\Delta_A = 0$ ), (iii) a single system is under observation and repair times of Component A are not negligible (m = 1,  $\Delta_A > 0$ ), and (iv) multiple systems are under observation and repair times of Component A are not negligible (m > 1,  $\Delta_A > 0$ ). In all cases, we assume the baseline rate functions are constants. We consider the settings in which baseline rate functions depend on time in Chapter 5.

#### **3.2.1** Case 1: $m = 1, \Delta_A = 0$

We first consider a single system (m = 1) with two components; Component A and Component B. In this setting, we assume that repair times of one of the components (say, Component A) is negligible; that is,  $\Delta_A = 0$  so that failures of Component A does not affect the probabilistic characteristics of the failure occurrences in Component B. Furthermore, we assume that failure occurrences are governed by HPPs. Under these assumptions and following the notation given previously, the model for Component A is of the form

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \,\alpha_A \,\exp\{\beta_A \, X_A(t)\}, \qquad t > 0, \tag{3.6}$$

where  $X_A(t)$  is defined in (3.2) and  $\mathcal{H}(t) = \{N_A(u), N_B(u); 0 \le u < t\}$ . Since  $\Delta_A = 0$  and  $\Delta_B > 0$ , the intensity function of Component B is given by

$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t)\,\alpha_B, \qquad t > 0. \tag{3.7}$$

A test for a parallel carryover effect in Component A can be developed by considering the following composite hypothesis.

$$H_0: \beta_A = 0, \ \alpha_A > 0, \qquad \text{vs.} \qquad H_1: \beta_A \neq 0, \ \alpha_A > 0,$$
(3.8)

where  $\alpha_A$  is a nuisance parameter.

Suppose that such a system with its components is under observation over the followup period  $[0, \tau]$ , where  $\tau$  is a fixed end-of-followup time. Notice that, since Component A is continuously under observation over  $[0, \tau]$  and its repair times are

negligible, we can safely drop  $Y_A(t)$  from the model (3.6). Let  $n_A$ , where  $n_A \ge 0$ , denote the number of failures of Component A in  $[0, \tau]$  and  $t_{A1}, t_{A2}, \ldots, t_{An_A}$  be the failure times of Component A. The likelihood function with the outcome " $N_A(\tau) = n_A$ failures of Component A at times  $t_{A1} \le t_{A2} \le \cdots \le t_{An_A}$  in  $[0, \tau]$ " is then given by

$$L(\boldsymbol{\theta}) = \prod_{j=1}^{n_A} \alpha_A \exp\{\beta_A X_A(t_{Aj})\} \exp\{-\int_0^\tau \alpha_A \exp\{\beta_A X_A(s)\}\,ds\},\qquad(3.9)$$

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A)$ . The log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$\ell(\boldsymbol{\theta}) = n_A \log \alpha_A + \sum_{j=1}^{n_A} \beta_A X_A(t_{Aj}) - \int_0^\tau \alpha_A \, e^{\beta_A X_A(s)} \, ds.$$
(3.10)

The components of the score vector  $\boldsymbol{U}(\boldsymbol{\theta}) = (U_{\alpha_A}(\boldsymbol{\theta}), U_{\beta_A}(\boldsymbol{\theta}))'$ , where  $U_{\alpha_A}(\boldsymbol{\theta}) = (\partial/\partial \alpha_A)\ell(\boldsymbol{\theta})$  and  $U_{\beta_A}(\boldsymbol{\theta}) = (\partial/\partial \beta_A)\ell(\boldsymbol{\theta})$  can be written as follows.

$$U_{\alpha_A}(\boldsymbol{\theta}) = \frac{n_A}{\alpha_A} - \int_0^\tau \exp\{\beta_A X_A(s)\} \, ds, \qquad (3.11)$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{j=1}^{n_A} X_A(t_{Aj}) - \alpha_A \int_0^\tau X_A(s) \exp\{\beta_A X_A(s)\} \, ds.$$
(3.12)

The observed information matrix  $I(\theta)$  is given by

$$\boldsymbol{I}(\boldsymbol{\theta}) = \begin{pmatrix} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) & I_{\alpha_A \beta_A}(\boldsymbol{\theta}) \\ I_{\beta_A \alpha_A}(\boldsymbol{\theta}) & I_{\beta_A \beta_A}(\boldsymbol{\theta}) \end{pmatrix}, \qquad (3.13)$$

with the components

$$I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) = \frac{n_A}{\alpha_A^2},\tag{3.14}$$

$$I_{\alpha_A\beta_A}(\boldsymbol{\theta}) = I_{\beta_A\alpha_A}(\boldsymbol{\theta}) = \int_0^\tau X_A(s) \exp\{\beta_A X_A(s)\} \, ds, \qquad (3.15)$$

$$I_{\beta_A\beta_A}(\boldsymbol{\theta}) = \alpha_A \int_0^\tau X_A(s) \exp\{\beta_A X_A(s)\} \, ds.$$
(3.16)

Let  $\tilde{\alpha}_A$  denote the restricted maximum likelihood estimator of  $\alpha_A$  under the null hypothesis  $H_0: \beta_A = 0$ . Letting  $\beta_A = 0$  in (3.11), and then solving  $U_{\alpha_A}(\alpha_A, 0) = 0$  for  $\alpha_A$ , we find that  $\tilde{\alpha}_A = n_A/\tau$ , where  $\tilde{\alpha}_A$  is the restricted maximum likelihood estimator of  $\alpha_A$ . For notational convenience, we define the integral

$$\mathcal{I}(\tau,\beta_A,\Delta_B) = \int_0^\tau X_A(s) \exp\{\beta_A X_A(s)\} \, ds.$$
(3.17)

The integral  $\mathcal{I}(\tau, \beta_A, \Delta_B)$  is a function of the followup period  $[0, \tau]$ , the regression parameter  $\beta_A$ , and the duration of the repairs of Component B; that is,  $\Delta_B$ . With this notation, the partial score statistic  $U_{\beta_A}(\tilde{\alpha}_A, 0)$ , where  $U_{\beta_A}(\alpha_A, \beta_A)$  is given in (3.12), can be written as

$$U_{\beta_A}(\tilde{\alpha}_A, 0) = \sum_{j=1}^{n_A} X_A(t_{Aj}) - \tilde{\alpha}_A \mathcal{I}(\tau, 0, \Delta_B), \qquad (3.18)$$

and the estimated variance of  $U_{\beta_A}(\tilde{\alpha}_A, 0)$  is given by

$$\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0)) = I_{\beta_A\beta_A}(\tilde{\alpha}, 0) - I_{\beta_A\alpha_A}(\tilde{\alpha}, 0) I_{\alpha_A\alpha_A}^{-1}(\tilde{\alpha}, 0) I_{\alpha_A\beta_A}(\tilde{\alpha}, 0), \qquad (3.19)$$

$$= \left(\frac{n_A}{\tau^2}\right) \mathcal{I}(\tau, 0, \Delta_B) \left[\tau - \mathcal{I}(\tau, 0, \Delta_B)\right].$$
(3.20)

The standardized partial score statistic for testing  $H_0$  in (3.8) is then

$$Z = \frac{U_{\beta}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},$$
(3.21)

where  $U_{\beta}(\tilde{\alpha}_A, 0)$  and  $\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0))$  are given in (3.18) and (2.20), respectively. As we show in Section 3.3, the distribution of (3.21) is asymptotically standard normal as  $\tau \to \infty$ . The *p*-values for  $H_0$  in (3.8) can be obtained from this approximation. Alternatively, when  $\tau$  is small, the *p*-values can be computed via simulation. In Section 3.4, we discuss the asymptotic properties and power of the test statistic (3.21) through simulations under various scenarios.

#### **3.2.2** Case 2: m > 1, $\Delta_A = 0$

We now consider the case in which the number of systems m is greater than 1 and repair times of Component A are negligible; that is, m > 1 and  $\Delta_A = 0$ . We consider a similar setup given in Section 3.2.1, so each system has two components connected in parallel and failures are governed by homogeneous Poisson processes (HPPs). We assume that failures of Component B affect the probabilistic characteristics of failure occurrences of Component A, while failures of Component A do not affect the probabilistic characteristics of failure occurrences of Component B.

Since we consider m > 1 in this case, we need to include an index to denote the system under observation. Therefore, we adapt the notation of the previous subsection to this situation. Let's suppose that we have m independent systems under observation, each with two components; Component A and Component B. Furthermore, suppose that there are *m* bivariate counting processes  $\{N_{Ai}(t), N_{Bi}(t); t \ge 0\}, i = 1, ..., m$ , where  $\{N_{Ai}(t); t \ge 0\}$  is a counting process for the failures of Component A in the *i*th system and  $\{N_{Bi}(t); t \geq 0\}$  is a counting process for the failures of Component B in the *i*th systems. For  $i = 1, \ldots, m$ , we let  $t_{Ai1}, t_{Ai2}, \ldots$ , where  $0 < t_{Ai1} < t_{Ai2} < \cdots$ , denote the failure times of Component A in the *i*th system. Similarly, for  $i = 1, \ldots, n$ m, we let  $t_{Bi1}$ ,  $t_{Bi2}$ , ..., where  $0 < t_{Bi1} < t_{Bi2} < \cdots$ , denote the failure times of Component B in the *i*th system. Thus,  $t_{Kij}$  is the *j*th failure time of Component K in the *i*th system, where K = A, B; i = 1, ..., m; and j = 1, 2, ... We use the notation  $Y_{Ai}(t)$  and  $Y_{Bi}(t)$  to denote the at-risk indicators of Component A and Component B in the *i*th system, respectively, so that  $Y_{Ki}$ , K = A, B, takes the value of 0 when Component K in the *i*th system is down; otherwise, it is equal to 1. Finally, we let  $\mathcal{H}_i(t) = \{N_{Ai}(u), N_{Bi}(u); 0 \le u < t\}$  denote the history of the bivariate process  $\{N_{Ai}(t), N_{Bi}(t); t \ge 0\}$ , where i = 1, ..., m.

The intensity function of Component A in the *i*th system, i = 1, ..., m, is given by

$$\lambda_{Ai}(t|\mathcal{H}_i(t)) = Y_{Ai}(t) \,\alpha_A \,\exp\{\beta_A \,X_{Ai}(t)\}, \qquad t > 0, \tag{3.22}$$

where  $X_{Ai}(t) = I\{N_{Bi}(t^{-}) > 0\}I\{t - t_{BiN_{Bi}(t^{-})} \leq \Delta_B\}$  and  $\Delta_B$  is the duration of the repairs of Component B. Since  $\Delta_A = 0$  and  $\Delta_B > 0$ , the intensity function of Component B in the *i*th system, where i = 1, ..., m, is given by

$$\lambda_{Bi}(t|\mathcal{H}_i(t)) = Y_{Bi}(t)\,\alpha_B, \qquad t > 0. \tag{3.23}$$

Once again, we consider the following hypothesis for a test of parallel carryover effects.

$$H_0: \beta_A = 0, \ \alpha_A > 0, \qquad \text{vs.} \qquad H_1: \beta_A \neq 0, \ \alpha_A > 0, \qquad (3.24)$$

where  $\alpha_A$  is a nuisance parameter.

Suppose that m such independent systems are under observation over the interval

 $[0, \tau_i]$ , where  $\tau_i$  is the end-of-followup time of the *i*th system and i = 1, ..., m. Also, let  $N_{Ai}(\tau_i) = n_{Ai}$  and  $t_{Ai1}, ..., t_{Ain_{Ai}}$  be the failure times of Component A in the *i*th system for i = 1, ..., m. The likelihood function for *m* independent systems is given by

$$L(\boldsymbol{\theta}) = \prod_{i=1}^{m} L_i(\boldsymbol{\theta}), \qquad (3.25)$$

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A)$  and

$$L_{i}(\boldsymbol{\theta}) = \prod_{j=1}^{n_{Ai}} \alpha_{A} \exp\{\beta_{A} X_{Ai}(t_{Aij})\} \exp\{-\int_{0}^{\tau_{i}} \alpha_{A} \exp\{\beta_{A} X_{Ai}(s)\} ds\}, \qquad (3.26)$$

which is the likelihood contribution of the *i*th system for the outcome " $N_{Ai}(\tau_i) = n_{Ai}$ failures of Component A in the *i*th system at times  $t_{Ai1} < t_{Ai2} < \cdots < t_{Ain_{Ai}}$ ". Once again, we would like to note that the at-risk indicators  $Y_{Ai}$ ,  $i = 1, \ldots, m$ , are not needed in the likelihood function (3.25). The corresponding log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$\ell(\boldsymbol{\theta}) = \sum_{i=1}^{m} \ell_i(\boldsymbol{\theta}), \qquad (3.27)$$

where

$$\ell_i(\boldsymbol{\theta}) = n_{Ai} \log \alpha_A + \beta_A \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \alpha_A \int_0^{\tau_i} \exp\{\beta_A X_{Ai}(s)\} \, ds.$$
(3.28)

Once again for notational convenience, we define the functions

$$\mathcal{I}_{1}(m,\beta_{A},\Delta_{B}) = \sum_{i=1}^{m} \int_{0}^{\tau_{i}} \exp\{\beta_{A}X_{Ai}(s)\}\,ds,$$
(3.29)

and

$$\mathcal{I}_2(m,\beta_A,\Delta_B) = \sum_{i=1}^m \int_0^{\tau_i} X_{Ai}(s) \, \exp\{\beta_A X_{Ai}(s)\} \, ds.$$
(3.30)

The functions (3.29) and (3.30) depend on the number of systems m, the duration of the downtimes of Component B; that is,  $\Delta_B$  and the parameter  $\beta_A$ . They also depend on the duration of followups  $[0, \tau_i]$ , i = 1, ..., m, but since we are not interested in the asymptotic distribution of the test statistic developed later in this section when  $\tau_i$ 

28

increases, we do not emphasize them in (3.29) and (3.30). With this notation, components of the score vector  $\boldsymbol{U}(\boldsymbol{\theta}) = (U_{\alpha_A}(\boldsymbol{\theta}), U_{\beta_A}(\boldsymbol{\theta}))'$ , where  $U_{\alpha_A}(\boldsymbol{\theta}) = (\partial/\partial \alpha_A)\ell(\boldsymbol{\theta})$ and  $U_{\beta_A}(\boldsymbol{\theta}) = (\partial/\partial \beta_A)\ell(\boldsymbol{\theta})$  are given by

$$U_{\alpha_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \frac{n_{Ai}}{\alpha_A} - \mathcal{I}_1(m, \beta_A, \Delta_B), \qquad (3.31)$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \alpha_A \mathcal{I}_2(m, \beta_A, \Delta_B).$$
(3.32)

The components of the  $2 \times 2$  observed information matrix  $I(\theta)$ , where

$$\boldsymbol{I}(\boldsymbol{\theta}) = \begin{pmatrix} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) & I_{\alpha_A \beta_A}(\boldsymbol{\theta}) \\ I_{\beta_A \alpha_A}(\boldsymbol{\theta}) & I_{\beta_A \beta_A}(\boldsymbol{\theta}) \end{pmatrix}, \qquad (3.33)$$

are given by

$$I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \frac{n_{Ai}}{\alpha_A^2},\tag{3.34}$$

$$I_{\alpha_A\beta_A}(\boldsymbol{\theta}) = I_{\beta_A\alpha_A}(\boldsymbol{\theta}) = \mathcal{I}_2(m, \beta_A, \Delta_B), \qquad (3.35)$$

$$I_{\beta_A\beta_A}(\boldsymbol{\theta}) = \alpha_A \mathcal{I}_2(m, \beta_A, \Delta_B).$$
(3.36)

Let  $\tilde{\alpha}_A$  be the restricted maximum likelihood estimator of  $\alpha_A$  under the null hypothesis  $H_0$ :  $\beta_A = 0$ . By solving  $U_{\alpha_A}(\alpha_A, 0) = 0$  in (3.32) for  $\alpha_A = \tilde{\alpha}_A$ , we can obtain

$$\tilde{\alpha}_{A} = \frac{\sum_{i=1}^{m} n_{Ai}}{\sum_{i=1}^{m} \tau_{i}}.$$
(3.37)

Following the score procedures explained in Chapter 1, we obtain the standardized partial score statistic

$$Z = \frac{U_{\beta}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},$$
(3.38)

for testing the null hypothesis  $H_0: \beta_A = 0$ , where

$$U_{\beta}(\tilde{\alpha}_{A},0) = \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \tilde{\alpha}_{A} \mathcal{I}_{2}(m,0,\Delta_{B}), \qquad (3.39)$$

and

$$\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0)) = \left(\frac{\sum_{i=1}^m n_{Ai}}{(\sum_{i=1}^m \tau_i)^2}\right) \mathcal{I}_2(m, 0, \Delta_B) \left(\sum_{i=1}^m \tau_i - \mathcal{I}_2(m, 0, \Delta_B)\right).$$
(3.40)

We discuss the asymptotic distribution of the test statistic (3.38) analytically as well as through simulations in Sections 3.3 and 3.4, respectively. In Section 3.3, we showed that the standardized partial score statistic Z in (3.38) converges to a standard normal distribution under the null hypothesis stated in (3.24) as  $m \to \infty$  for fixed observation periods. Therefore, this result can be used to calculate p-values for testing the presence of parallel carryover effects in the redundant component (Component A) when m is large. For small m values, p-values can be obtained via simulations.

#### **3.2.3** Case 3: $m = 1, \Delta_A > 0$

In some systems, one of the components operates in a constant or full operating capacity in the up state and does not change its load following the failures of other components, while other components are still redundant. We now focus on such a system with two components; Component A and Component B. The setup of this section is similar to that of Section 3.2.1 except that both Components A and B are subject to non-negligible repair times.

The model for Component A is given by

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \,\alpha_A \,\exp\{\beta_A X_A(t)\}, \qquad t > 0, \tag{3.41}$$

where  $X_A(t)$  is defined in (3.2) and  $\mathcal{H}(t) = \{N_A(u), N_B(u); 0 \le u < t\}$ . It should be noted that this is the same model for Component A given in Section 3.2.1, but in this case, since  $\Delta_A > 0$ , we cannot drop  $Y_A(t)$  from the model. The model for the Component B is once again defined by

$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t)\,\alpha_B, \qquad t > 0. \tag{3.42}$$

Similarly, we want to test the null hypothesis  $H_0$ :  $\beta_A = 0$  against the alternative hypothesis  $H_1: \beta_A \neq 0$ .

Following the setup in Section 3.2.1, the likelihood function with the outcome that " $N_A(\tau) = n_A$  failures of Component A at times  $t_{A1} < t_{A2} < \cdots < t_{An_A}$  in a fixed

interval  $[0, \tau]$ " can be written as follows.

$$L(\boldsymbol{\theta}) = \prod_{j=1}^{n_A} \alpha_A \exp\{\beta_A X_A(t_{Aj})\} \exp\{-\int_0^\tau Y_A(t) \,\alpha_A \,\exp\{\beta_A X_A(s)\} \,ds\}.$$
 (3.43)

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A)$ . The log likelihood function  $\ell(\boldsymbol{\theta})$  is given by

$$l(\boldsymbol{\theta}) = n_A \log \alpha_A + \sum_{j=1}^{n_A} \beta_A X_A(t_{Aj}) - \int_0^\tau Y_A(t) \,\alpha_A \,\exp\{\beta_A X_A(s)\}\,ds \tag{3.44}$$

The components of the score vector  $U(\boldsymbol{\theta})$  are given by

$$U_{\alpha_A}(\boldsymbol{\theta}) = \frac{n_A}{\alpha_A} - \int_0^\tau Y_A(t) \, \exp\{\beta_A X_A(s)\} \, ds, \qquad (3.45)$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{j=1}^{n_A} X_A(t_{Aj}) - \alpha_A \int_0^\tau Y_A(t) X_A(s) \exp\{\beta_A X_A(s)\} \, ds.$$
(3.46)

Also, components of the 2  $\times$  2 observation matrix  $I(\theta)$  are given by

$$I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) = \frac{n_A}{\alpha_A^2},\tag{3.47}$$

$$I_{\alpha_A\beta_A}(\boldsymbol{\theta}) = I_{\beta_A\alpha_A}(\boldsymbol{\theta}) = \int_0^\tau Y_A(s) X_A(s) \exp\{\beta_A X_A(s)\} \, ds, \qquad (3.48)$$

$$I_{\beta_A\beta_A}(\boldsymbol{\theta}) = \alpha_A \int_0^\tau Y_A(s) X_A(s) \exp\{\beta_A X_A(s)\} \, ds.$$
(3.49)

Let  $\tilde{\alpha}_A$  be the restricted maximum likelihood estimator of  $\alpha_A$  under the null hypothesis  $H_0: \beta_A = 0$ . Then, we obtain

$$\tilde{\alpha}_A = \frac{n_A}{\int_0^\tau Y_A(s)ds}.$$
(3.50)

Notice that the restricted maximum likelihood estimator  $\tilde{\alpha}_A$  is the ratio of the observed number of failures in Component A over  $[0, \tau]$  to the total time that Component A stays in the up state in  $[0, \tau]$ . For convenience, we define

$$\mathcal{I}(\tau,\beta_A,\Delta_B) = \int_0^\tau Y_A(s) X_A(s) \exp\{\beta_A X_A(s)\} ds, \qquad (3.51)$$

so that, under the null hypothesis  $H_0: \beta_A = 0, \mathcal{I}(\tau, 0, \Delta_B)$  depends on the end of the follow up time  $\tau$  and the choice of  $\Delta_B$ . Then,

$$U_{\beta}(\tilde{\alpha}_A, 0) = \sum_{j=1}^{n_A} X_A(t_j) - \tilde{\alpha}_A \mathcal{I}(\tau, 0, \Delta_B), \qquad (3.52)$$

and

$$\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0)) = \frac{n_A \mathcal{I}(\tau, 0, \Delta_B)}{\left\{\int_0^{\tau} Y_A(s) ds\right\}^2} \left[\int_0^{\tau} Y_A(s) ds - \mathcal{I}(\tau, 0, \Delta_B)\right].$$
(3.53)

Therefore, we obtain the partial score statistic for testing  $H_0: \beta_A = 0$  as follows

$$Z = \frac{U_{\beta}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}}.$$
(3.54)

Under mild regularity conditions on  $Y_A(t)$ , it can be shown with a very similar method given in Section 3.3 that the distribution of the test statistic (3.53) asymptotically converges to a standard normal distribution as  $\tau$  increases. We discuss this via simulations in Section 3.4.

#### **3.2.4** Case 4: $m > 1, \Delta_A > 0$

We now consider the case where multiple systems are under observation and repair times of failures in Component A are not negligible; that is, m > 1 and  $\Delta_A > 0$ . This case is, therefore, an extension of the case given in Section 3.2.2.

Following the notation introduced in Section 3.2.2 using the model (3.22) for Component A and the model (3.23) for Component B, the likelihood function for m independent systems can be written as follows.

$$L(\boldsymbol{\theta}) = \prod_{i=1}^{m} L_i(\boldsymbol{\theta}), \qquad (3.55)$$

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A)$  and the likelihood contribution of Component A in the *i*th system is given by

$$L_i(\boldsymbol{\theta}) = \prod_{j=1}^{n_{Ai}} \alpha_A \exp\{\beta_A X_{Ai}(t_{Aij})\} \exp\{-\int_0^{\tau_i} Y_{Ai}(s)\alpha_A \exp\{\beta_A X_{Ai}(s)\} \, ds\}.$$
 (3.56)

The log likelihood function is then given by  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta}) = \sum_{i=1}^{m} \log L_i(\boldsymbol{\theta})$  where

 $\log L_i(\boldsymbol{\theta})$  is

$$n_{Ai} \log \alpha_A + \beta_A \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \alpha_A \int_0^{\tau_i} Y_{Ai}(s) \exp\{\beta_A X_{Ai}(s)\} \, ds.$$
(3.57)

Taking the derivatives of  $\ell(\boldsymbol{\theta})$  with respect to  $\alpha_A$  and  $\beta_A$ , we obtain the score functions as follows.

$$U_{\alpha_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \frac{n_{Ai}}{\alpha_A} - \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \exp\{\beta_A X_{Ai}(s)\} \, ds, \tag{3.58}$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \alpha_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) X_{Ai}(s) \exp\{\beta_A X_{Ai}(s)\} \, ds.$$
(3.59)

Now consider the null hypothesis  $H_0: \beta_A = 0$ . Under this null hypothesis, solving  $U_{\alpha_A}(\alpha_A, 0) = 0$  for  $\alpha_A = \tilde{\alpha}_A$  gives

$$\tilde{\alpha}_A = \frac{\sum_{i=1}^m n_{Ai}}{\sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, ds},\tag{3.60}$$

which is the restricted maximum likelihood estimator of  $\alpha_A$ . For convenience, we define

$$\mathcal{I}(m, \beta_A, \Delta_B) = \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) X_{Ai}(s) \exp\{\beta_A X_A\} \, ds,$$
(3.61)

Then, the partial score statistic for testing  $H_0: \beta_A = 0$  is given by

$$Z = \frac{U_{\beta}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},$$
(3.62)

where

$$U_{\beta}(\tilde{\alpha}_{A}, 0) = \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \tilde{\alpha}_{A} \mathcal{I}(m, 0, \Delta_{B}), \qquad (3.63)$$

and

$$\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0)) = \frac{\mathcal{I}(m, 0, \Delta_B) \sum_i^m n_{Ai}}{(\sum_{i=1}^m \int_0^\tau Y_{Ai}(s) \, ds)^2} \left[ \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, ds - \mathcal{I}(m; \Delta_B) \right].$$
(3.64)

By following the method given in Section 3.3, it can be shown that, the asymptotic

distribution of the test statistics (3.62) is a standard normal distribution as  $m \to \infty$ . We discuss this convergence result through simulations in Section 3.4.

## **3.3** Asymptotic Properties of Test Statistics

In this section, we discuss the asymptotic properties of the test statistics (3.21) and (3.38) analytically. We consider the asymptotic distribution of the former statistic when the observation period  $\tau$  increases. We utilize simple results from the martingale theory for this purpose. As for the latter statistic, we consider the asymptotic distribution when the number of systems m increases for a fixed observation period. In this case, we show that our model belongs to the family of point process models considered in Andersen et al. (1993, Chapter VI.1.2) and Peña (1998), and satisfies the conditions stated by them to obtain the large sample properties of the test statistic. The development in this section is primarily based on Cigsar (2010, Section 2.3).

Consider the setup given in Section 3.2.1, where a single system (m = 1) with two components (Components A and B) is under observation over a time interval  $[0, \tau]$ , where  $\tau$  is a prespecified end-of-followup time. Also,  $\Delta_A$  and  $\Delta_B$  denote the repair times of Components A and B, respectively, where  $\Delta_A = 0$  and  $\Delta_B > 0$ . Let  $\{N_A(t), t > 0\}$  be the counting process for failure occurrences in Component A with the associated intensity function

$$\lambda_A(t \mid \mathcal{H}(t)) = \alpha_A \exp\{\beta_A X_A(t)\},\tag{3.65}$$

where  $X_A(t) = I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \le \Delta_B\}$ , and  $\{N_B(t), t > 0\}$  is the counting process for failure occurrences in Component B with the intensity function given in (3.7), and  $\mathcal{H}(t) = \{N_A(u), N_B(u); 0 \le u < t\}$ . In Section 3.2.1, we consider testing the hypothesis  $H_0: \beta_A = 0$  against  $H_1: \beta_A \neq 0$ , and develop the partial score test statistic

$$Z = \frac{U_{\beta}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},$$
(3.66)

where  $\tilde{\alpha}_A = n_A/\tau = (1/\tau) \int_0^\tau dN_A(t)$ , and

$$U_{\beta_A}(\tilde{\alpha}_A, 0) = \sum_{j=1}^{n_A} X_A(t_j) - \tilde{\alpha}_A \int_0^\tau X_A(t) dt$$
 (3.67)

and

$$\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0)) = \left(\frac{n_A}{\tau^2}\right) \int_0^\tau X_A(t) dt \left[\tau - \int_0^\tau X_A(t) dt\right].$$
(3.68)

We want to show that, under the null hypothsis  $H_0: \beta_A = 0$ , the test statistic Z in (3.66) converges in distribution to a standard normal random variable as  $\tau$  increases; that is, under  $H_0, Z \xrightarrow{d} N(0, 1)$  as  $\tau \to \infty$ . Let  $\alpha_{A0}$  be the true value of the parameter  $\alpha_A$  assuming that the null hypothesis is true. In the following development, we use  $\tau$  in the superscripts of score functions to show their dependence on the observation period. From the score function given in (3.18), we can write that

$$\frac{1}{\sqrt{\tau}}U^{\tau}_{\beta_A}(\tilde{\alpha}_A,0) = \frac{1}{\sqrt{\tau}}U^{\tau}_{\beta_A}(\alpha_0) - \sqrt{\tau}(\tilde{\alpha}_A - \alpha_0)\frac{1}{\tau}\int_0^{\tau} X_A(t) dt, \qquad (3.69)$$

where

$$\frac{1}{\sqrt{\tau}} U^{\tau}_{\beta_A}(\alpha_0) = \frac{1}{\sqrt{\tau}} \int_0^{\tau} X_A(t) \, dM_A(t), \qquad (3.70)$$

and  $M_A(t) = \int_0^t [dN_A(s) - \alpha_0 ds]$  is a *bona fide* martingale with respect to the history  $\mathcal{H}(t)$  (Daley and Vere-Jones, 2003, p. 428). We assume that  $X_A(t)$  is measurable with respect to  $\mathcal{H}(t)$  at time  $t^-$  in  $[0, \tau]$  so it is predictable with respect to  $\mathcal{H}(t)$  (Daley and Vere-Jones, 2003, p. 425). Also, notice that, for any t in  $[0, \Delta_B]$ ,  $X_A(t) = 1$  if there is at least one failure of Component B in  $[0, \Delta_B]$ ; otherwise,  $X_A(t) = 0$ . Similarly, for any t in  $[\Delta_B, \tau]$ ,  $X_A(t) = 1$  if there is at least one failure of Component B in  $[t - \Delta_B, t]$ ; otherwise,  $X_A(t) = 0$ . Therefore,

$$E\left\{\int_0^\tau X_A(t)\,dt\right\} = \int_0^{\Delta_B} \left(1 - e^{-\alpha_B t}\right)\,dt + \int_{\Delta_B}^\tau \left(1 - e^{-\alpha_B \Delta_B}\right)\,dt. \tag{3.71}$$

From (3.71), we can show that

$$\lim_{\tau \to \infty} (1/\tau) E\left\{\int_0^\tau X_A(t) dt\right\} = 1 - e^{-\alpha_B \Delta_B}.$$
(3.72)

Let  $\vartheta = 1 - e^{-\alpha_B \Delta_B}$ , where  $0 < \vartheta < 1$ . Therefore, by a weak law of large numbers, as  $\tau \to \infty$ , we obtain

$$\frac{1}{\tau} \int_0^\tau X_A(t) \,\alpha_0 \, dt \xrightarrow{p} \alpha_0 \vartheta. \tag{3.73}$$

Also, notice that, for every  $\varepsilon > 0$  and sufficiently large  $\tau$ , almost surely  $I(|X_A(t)| >$ 

 $\varepsilon\sqrt{\tau}$ ) = 0 for all  $t \in [0, \tau]$ . Therefore, for every  $\varepsilon > 0$ , we have

$$\lim_{\tau \to \infty} (1/\tau) E\left[\int_0^\tau X_A^2(t) I(|X_A(t)| > \varepsilon \sqrt{\tau}) \alpha_0 dt\right] = 0.$$
(3.74)

From the results (3.72) and (3.74) and a central limit theorem for point process martingales (see, Karr, 1991, pp. 421–422, for the theorem and its proof), we obtain the convergence

$$\frac{1}{\sqrt{\tau}} U^{\tau}_{\beta_A}(\alpha_0) \xrightarrow{d} N(0, \alpha_0 \vartheta), \qquad (3.75)$$

as  $\tau \to \infty$ . A convergence result for  $\sqrt{\tau}(\tilde{\alpha}_A - \alpha_0)$  can be shown similarly. First notice that

$$\sqrt{\tau}(\tilde{\alpha}_A - \alpha_0) = \frac{1}{\sqrt{\tau}} \int_0^\tau dM_A(t).$$
(3.76)

Therefore, by taking  $X_A(t) = 1$  in (3.72) and (3.74), we have

$$\sqrt{\tau}(\tilde{\alpha}_A - \alpha_0) \xrightarrow{d} N(0, \alpha_0),$$
 (3.77)

as  $\tau \to \infty$ . Since  $E\{dM_A(t)\} = E\{dN_A(t) - \alpha_0 dt\} = 0$  under the null hypothesis, we have

$$Cov\left\{\frac{1}{\sqrt{\tau}}U^{\tau}_{\beta_{A}}(\alpha_{0}),\sqrt{\tau}(\tilde{\alpha}_{A}-\alpha_{0})\right\} = \frac{1}{\tau}Cov\left\{\int_{0}^{\tau}X_{A}(t)\,dM_{A}(t),\int_{0}^{\tau}dM_{A}(t)\right\}.$$
$$=\frac{1}{\tau}E\left\{\int_{0}^{\tau}X_{A}(t)\,\alpha_{0}\,dt\right\}.$$
(3.78)

Therefore, from (3.72), we obtain

$$Cov\left\{\frac{1}{\sqrt{\tau}}U^{\tau}_{\beta_{A}}(\alpha_{0}),\sqrt{\tau}(\tilde{\alpha}_{A}-\alpha_{0})\right\}\to\alpha_{0}\vartheta,$$
(3.79)

as  $\tau \to \infty$ . From the convergence results in (3.75), (3.77) and (3.79), and by applying Slutsky's Theorem,  $(1/\tau)U^{\tau}_{\beta_A}(\tilde{\alpha}_A, 0)$  in (3.69) converges in distribution to a zero mean normal distribution with the asymptotic variance  $\alpha_0 \vartheta(1 - \vartheta)$  as  $\tau \to \infty$  under the null hypothesis. Therefore, we obtain the following result. Under the null hypothesis  $H_0: \beta_A = 0$ , as  $\tau \to \infty$ ,

$$\frac{\frac{1}{\sqrt{\tau}}U^{\tau}_{\beta_{A}}(\tilde{\alpha}_{A},0)}{\sqrt{\alpha_{0}\vartheta(1-\vartheta)}} \xrightarrow{d} N(0,1).$$
(3.80)

The terms  $\alpha_0$  and  $\vartheta$  in (3.80) can be replaced by their consistent estimators. In this

case, we obtain the variance estimator given in (3.68).

In the remaining part of this section, we discuss the asymptotic properties of the test statistic given in (3.38) as  $m \to \infty$  for a fixed  $\tau$  value. Andersen et al. (1993, pp. 420–421) consider a general intensity model which depends on a vector of parameters  $\boldsymbol{\theta}$  for a multivariate counting process. They state five conditions (Conditions A–E) to derive the large sample properties of the maximum likelihood estimators  $\hat{\boldsymbol{\theta}}$  of  $\boldsymbol{\theta}$ . These conditions have to be checked for each specific model under study. We do not restate these conditions here or give the proofs, but our goal is to show that the model (3.22) satisfies these conditions, and therefore we can apply the results of the theorems. In particular, under these conditions, the score vector  $\boldsymbol{U}(\boldsymbol{\theta}) = \boldsymbol{0}$  has a solution  $\hat{\boldsymbol{\theta}}$  and  $\hat{\boldsymbol{\theta}}$  converges in probability to  $\boldsymbol{\theta}_0$  as  $m \to \infty$  (Andersen et al., 1993, Theorem VI.1.1., p. 422). Furthermore, if  $\hat{\boldsymbol{\theta}}$  is a consistent solution of  $\boldsymbol{U}(\boldsymbol{\theta}) = \boldsymbol{0}$ , then  $(1/\sqrt{m}) \boldsymbol{U}(\hat{\boldsymbol{\theta}}) \stackrel{d}{\to} N(0, \boldsymbol{\Sigma})$  as  $m \to \infty$ , where  $\boldsymbol{\Sigma}$  is defined in Condition D (Andersen et al., pp. 424–426).

Following the setup given in Section 3.2.2, in our case the intensity function of Component A in the *i*th system, i = 1, ..., m, is given by

$$\lambda_{Ai}(t|\mathcal{H}_i(t)) = Y_{Ai}(t) \,\alpha_A \,\exp\{\beta_A X_{Ai}(t)\}, \qquad t > 0, \tag{3.81}$$

where  $X_{Ai}(t) = I\{N_{Bi}(t^-) > 0\}I\{t - t_{BiN_{Bi}(t^-)} \le \Delta_B\}$  and  $\mathcal{H}_i(t) = \{N_{Ai}(u), N_{Bi}(u); 0 \le u < t\}.$ 

Let  $\boldsymbol{\theta} = (\alpha_A, \beta_A)'$ . We now check Conditions A–E of Andersen et al. (1993, pp. 420–421) for the above model under the null hypothesis  $H_0: \beta_A = 0$ . Suppose that, under the null hypothesis, the true value of  $\alpha_A$  is  $\alpha_0$ , where  $\alpha_0 > 0$ , and  $\boldsymbol{\theta}_0 = (\alpha_0, 0)'$ . Condition A is a Cramèr-type regularity condition on  $\lambda_{Ai}(t|\mathcal{H}_i(t))$ ,  $\log \lambda_{Ai}(t|\mathcal{H}_i(t))$  and the log likelihood function given in (3.28). It is easy to see that Condition A holds in our case. Now notice that, for a fixed  $\tau > 0$ , we have

$$\frac{1}{m}\sum_{i=1}^{m}\int_{0}^{\tau}\frac{Y_{Ai}(s)}{\alpha_{0}}\,ds \xrightarrow{p} \sigma_{\alpha_{A}\alpha_{A}}, \qquad \text{as } m \to \infty.$$
(3.82)

Since  $\tau$  is a fixed value and  $\Delta_A = 0$ ,  $Y_{Ai}(t) = 1$  in  $[0, \tau]$ . Therefore,  $\sigma_{\alpha_A \alpha_A} = \tau/\alpha_0$ . Next, we need to show the following convergence result.

$$\frac{1}{m}\sum_{i=1}^{m}\int_{0}^{\tau}Y_{Ai}(s)X_{Ai}(s)\,ds \xrightarrow{p} \sigma_{\alpha_{A}\beta_{A}}, \qquad \text{as } m \to \infty, \tag{3.83}$$

where  $\sigma_{\alpha_A\beta_A} > 0$ . Since  $\tau$  is fixed and  $Y_{Ai}(t) = 1$  in  $[0, \tau]$ , the result in (3.83) follows by a weak law of large numbers, where  $\sigma_{\alpha_A\beta_A}$  is the right hand side of (3.71). It should be noted that if  $Y_{Ai}(t)$  is not always 1 (e.g., when  $\Delta_A > 0$ ), some conditions needs to be applied. Similarly, for  $\tau > 0$ , we have

$$\frac{1}{m} \sum_{i=1}^{m} \int_{0}^{\tau} Y_{Ai}(s) \,\alpha_0 \, X_{Ai}(s) \, ds \xrightarrow{p} \sigma_{\beta_A \beta_A}, \qquad \text{as } m \to \infty, \tag{3.84}$$

where  $\sigma_{\beta_A\beta_A} = \alpha_0 \sigma_{\alpha_A\beta_A}$ . Therefore, under the null hypothesis  $H_0: \beta_A = 0$ , Condition B is satisfied with  $\sigma_{\alpha_A\alpha_A}, \sigma_{\alpha_A\beta_A}$  and  $\sigma_{\beta_A\beta_A}$  given above.

As for Condition C, we need to show that, for all  $\varepsilon > 0$ ,

$$\frac{1}{m}\sum_{i=1}^{m}\int_{0}^{\tau}Y_{Ai}(s)\frac{1}{\alpha_{0}}I\left\{\frac{1}{\alpha_{0}\sqrt{m}}>\varepsilon\right\}\,ds\xrightarrow{p}0,\qquad\text{as }m\to\infty,\qquad(3.85)$$

and

$$\frac{1}{m}\sum_{i=1}^{m}\int_{0}^{\tau}Y_{Ai}(s)X_{Ai}^{2}(s)I\left\{\frac{X_{Ai}(s)}{\sqrt{m}}>\varepsilon\right\}\,ds\xrightarrow{p}0,\qquad\text{as }m\to\infty.$$
(3.86)

Notice that, for a fixed  $\tau$ ,  $Y_{Ai}(s) = 1$ , where  $s \in [0, \tau]$ , and  $\alpha_0 > 0$ , the left hand side of (3.85) is  $(\tau/\alpha_0)I\left\{\frac{1}{\sqrt{m}} > \alpha_0 \varepsilon\right\}$ , which converges to 0 as  $m \to \infty$ . Therefore, the convergence in (3.85) holds. The latter convergence result can be shown with similar arguments which lead to the convergence result in (3.74). In this case,  $X_A(s)$ in (3.74) needs to be replaced by  $X_{Ai}(s)$ , and limit should be taken on the average of the expectation of the terms when  $m \to \infty$ , instead of  $\tau \to \infty$ . The result follows from the weak law of large numbers. Therefore, Condition C holds for the model in (3.81).

Let  $\Sigma$  be a 2 × 2 matrix defined as follows.

$$\boldsymbol{\Sigma} = \begin{pmatrix} \sigma_{\alpha_A \alpha_A} & \sigma_{\alpha_A \beta_A} \\ \sigma_{\beta_A \alpha_A} & \sigma_{\beta_A \beta_A} \end{pmatrix}, \qquad (3.87)$$

where  $\sigma_{\alpha_A\alpha_A}$ ,  $\sigma_{\alpha_A\beta_A}$  and  $\sigma_{\beta_A\beta_A}$  are given above, and  $\sigma_{\beta_A\alpha_A} = \sigma_{\alpha_A\beta_A}$ . Condition D requires that the matrix  $\Sigma$  should be positive definite. Notice that  $\sigma_{\alpha_A\alpha_A} = \tau/\alpha_0$  is positive because  $\alpha_0 > 0$ ,  $\tau > 0$  and  $\tau$  is fixed. We also need to show that

 $\sigma_{\alpha_A\beta_A}(\tau - \sigma_{\alpha_A\alpha_B}) > 0$ . Recall that

$$0 < \sigma_{\alpha_A \beta_A} = \int_0^{\Delta_B} \left( 1 - e^{-\alpha_B t} \right) dt + \int_{\Delta_B}^{\tau} \left( 1 - e^{-\alpha_B \Delta_B} \right) dt$$
$$< \Delta_B (1 - e^{-\alpha_B \Delta_B}) + (\tau - \Delta_B) (1 - e^{-\alpha_B \Delta_B}) = \tau (1 - e^{-\alpha_B \Delta_B}) < \tau.$$
(3.88)

Therefore, the matrix  $\Sigma$  is positive definite, and Condition D holds. Condition E is required for the boundedness of the third derivaties of  $\lambda_{Ai}(t|\mathcal{H}_i(t))$  and  $\log \lambda_{Ai}(t|\mathcal{H}_i(t))$ given in (3.81) with respect to  $\theta$  under the null hypothesis, as well as for the regularity of the remainder term of a Taylor series expansion. It is easy to see that the requirements on the boundedness is satified for the model (3.81). Also, requirements on the remainder term hold for the model (3.81). It is worth noting that these conditions can be shown for the model (2.23) by following a similar method. In this case, the above approach should include mild regularity conditions on the at-risk indicator  $Y_{Bi}(t), i = 1, \ldots, m$ , such as the integrals  $\int_0^{\tau} Y_{Bi}(s) ds$  should not be equal to 0.

Since ConditionA–E given in Andersen et al. (1993, pp. 420–421) are satisfied for the model (3.81). Therefore, from Andersen et al. (1993, p. 424–426), we conclude that, under the null hypothesis  $H_0: \beta_A = 0, \alpha_A > 0$ ,

$$\frac{\frac{1}{\sqrt{m}}U_{\beta_A}(\tilde{\alpha}_A, 0)}{\sigma(\alpha_0)} \xrightarrow{d} N(0, 1), \qquad \text{as } m \to \infty,$$
(3.89)

where  $\tilde{\alpha}_A = \sum_{i=1}^m n_{Ai} / \sum_{i=1}^m \tau_i$  is the restricted maximum likelihood estimator of  $\alpha_A$ and  $\sigma^2(\alpha_0) = \alpha_{\beta_A\beta_A} - (\sigma^2_{\alpha_A\beta_A}/\sigma_{\alpha_A\alpha_A})$ . We obtain the final convergence result by replacing  $\sigma^2(\alpha_0)$  with a consistent estimator of it. Therefore, we obtain that, by Slutsky's theorem, under the null hypothesis  $H_0: \beta_A = 0, \alpha_A > 0$ ,

$$Z = \frac{U_{\beta}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}} \xrightarrow{d} N(0, 1), \quad \text{as } m \to \infty, \quad (3.90)$$

where Z is given in (3.38).

## 3.4 Simulation Studies

In this section, we present the results of simulation studies carried out to assess when asymptotic normal approximation for testing parallel carryover effects of redundant system is satisfactory. We consider two settings; (i) where m = 1 and  $\tau \to \infty$ , and (ii)  $m \to \infty$  and  $\tau$  is fixed.

We consider testing for parallel carryover effects in a single system (m=1) where repair times of Component A are negligible; that is, m = 1,  $\Delta_A = 0$ ,  $\Delta_B > 0$ . The composite hypothesis of no parallel carryover effects in Component A in this case is  $H_0$ :  $\beta_A = 0$ ,  $\alpha_A > 0$  and the standardized partial score statistic associated with the null hypothesis, as given in Section 3.2.1, is

$$Z = \frac{U_{\beta_A}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},\tag{3.91}$$

where the score statistic  $U_{\beta_A}(\tilde{\alpha}_A, 0)$  is given by (3.18) and its estimated variance  $\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}$  is given by (3.19). Our goal is to investigate the null distribution of Z and to assess the adequacy of the standard normal approximation when  $\tau$  increases. We generated 10,000 realizations of HPPs based on (3.6) and (3.7). In the simulation, we took  $\alpha_A = \alpha_B = 0.1$  and  $\beta_A = 0$ . We consider  $\Delta_B = 1, 3, 7$  and 14.

Normal quantile-quantile (Q-Q) plots of 10,000 values of Z are presented in Figure 3.1 when  $\Delta_B = 1$  for  $\tau = 100, 200, 500$  and 1000. The standard normal approximations are not quite accurate in those four cases. However, it is noted that as  $\tau$ increases, there is an apparent improvement in the standard normal approximation. Figure 3.2 shows the results when  $\Delta_B = 3$ . In this scenario, the standard normal approximations are suitable for  $\tau = 500$  and 1000. In Figure 3.3, we can see that the standard normal approximations are quite accurate for  $\tau = 500$  and  $\tau = 1000$ when  $\Delta_B = 7$ . Figure 3.4 shows also that, when  $\Delta_B = 14$ , the standard normal approximations are adequate for all  $\tau = 100, 200, 500$  and 1000.

Table 3.1 presents estimates of  $Q_p$  and  $Pr(Z > Q_p)$  where  $\hat{Q}_p$  is the empirical *p*th quantile of Z computed from 10,000 samples,  $Q_p$  is the *p*th quantile of the standard normal distribution, and  $\hat{Pr}(Z > Q_p) = 1 - p$  where p = 0.950, 0.975 and 0.990. In all the scenarios with  $\Delta_B = 1, 3, 7$  and 14, the standard normal approximations are quite accurate.

The power of the statistic (3.91) against the alternative hypothesis  $H_A$ :  $\beta_A \neq 0$ ,  $\alpha_A > 0$  is investigated. We use Monte Carlo simulation methods to obtain the power of the test. We use 0.95 quantile of the standard normal distribution and the empirical 0.95 quantile of the test statistic in the 10,000 simulations runs that are under the null hypothesis with different  $\tau$  and  $\Delta_B$  values as discussed above. We generated 1,000 processes under the model (3.6) and (3.7). We took  $\alpha_A = \alpha_B = 0.1$ . We set  $\beta_A$  at 0.693 and 0.916, where  $\exp\{0.693\} = 2$  and  $\exp\{0.916\} = 2.5$ . Here,  $\exp\{\beta_A\}$  means that while Component B is under repair, the rate of occurrences of failures in Component A is  $\exp\{\beta_A\}$  times higher comparing with the rate while Component B is in the up state. The power results are provided in Table 3.2, where entries are the proportions of the values of Z in 1,000 samples which are larger than the quantile values. This shows that the power of the test is high overall except that when  $\exp\{\beta_A\} = 2$  and  $\tau$  is small.



Figure 3.1: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 3.2: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 3$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 3.3: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 7$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 3.4: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 14$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 

$\Delta_B$	au	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	100	1.854	2.356	2.940	0.0657	0.0445	0.0267
	200	1.840	2.265	2.712	0.0676	0.0416	0.0228
	500	1.766	2.118	2.675	0.0608	0.0349	0.0168
	1000	1.753	2.169	2.586	0.0605	0.0355	0.0181
3	100	1.775	2.204	2.663	0.0636	0.0387	0.0198
	200	1.699	2.036	2.440	0.0563	0.0296	0.0125
	500	1.709	2.097	2.457	0.0557	0.0314	0.0144
	1000	1.694	2.005	2.449	0.0547	0.0278	0.0119
7	100	1.707	2.029	2.434	0.0576	0.0318	0.0127
	200	1.690	2.001	2.401	0.0553	0.0281	0.0116
	500	1.636	1.939	2.312	0.0496	0.0238	0.0099
	1000	1.637	1.975	2.334	0.0494	0.0259	0.0102
14	100	1.643	1.909	2.224	0.0499	0.0232	0.0075
	200	1.590	1.876	2.239	0.0452	0.0209	0.0080
	500	1.605	1.869	2.254	0.0456	0.0211	0.0078
	1000	1.622	1.944	2.289	0.0482	0.0243	0.0087

Table 3.1:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m = 1.  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta_B$	au	$\hat{Pr}\{Z > 1.645   H_a\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}   H_a\}$	$\hat{Pr}\{Z > 1.645   H_a\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}   H_a\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 6$	$e^{\beta_A} = 6$
1	100	0.254	0.214	0.861	0.836
	200	0.357	0.312	0.974	0.964
	500	0.565	0.528	1.000	1.000
	1000	0.805	0.782	1.000	1.000
3	100	0.327	0.292	0.974	0.971
	200	0.523	0.493	0.999	0.998
	500	0.818	0.800	1.000	1.000
	1000	0.975	0.971	1.000	1.000
7	100	0.382	0.345	0.995	0.994
	200	0.548	0.536	1.000	1.000
	500	0.885	0.886	1.000	1.000
	1000	0.991	0.991	1.000	1.000
14	100	0.308	0.309	0.995	0.995
	200	0.566	0.588	1.000	1.000
	500	0.877	0.881	1.000	1.000
	1000	0.992	0.992	1.000	1.000

Table 3.2: Power of Z :  $m = 1, \Delta_A = 0$ 

We now consider testing for the presence of parallel carryover effects in multiple systems where repair times of Component A are negligible, which is Case 2 of Section 3.2.2, where m > 1,  $\Delta_A = 0$  and  $\Delta_B > 0$ . The hypothesis of no parallel carryover effects in Component A is conducted by using the statistic Z in (3.38). We generated 10,000 realizations of m HPPs under the null hypothesis with fixed values of the parameters, where  $\alpha_A = \alpha_B = 0.1$ , and  $\beta_A = 0$ . We considered  $\Delta_B = 1, 3, 7$  and 14 and fixed  $\tau$  at 100 this time. Normal quantile-quantile (Q-Q) plots of 10,000 values of Z are given in Figures 3.5, 3.6, 3.7 and 3.8 with various combinations of m and  $\Delta_B$ . When  $\Delta_B = 1$  in Figure 3.5, the standard normal approximation is appropriate except when m = 10. The standard normal approximations are quite suitable when  $\Delta_B = 3, 7$  and 14.

Table 3.3 shows estimates of  $Q_p$  and  $Pr(Z > Q_p)$  when p = 0.950, 0.975 and 0.990. Table 3.3 also indicates that the standard normal approximations are overall accurate for all m and  $\Delta_B$  values except that when m = 10 and 20 and  $\Delta_B = 1$  and 3 in the tails.

The power of the statistic (3.38) against the alternative hypothesis  $H_A : \beta_A \neq 0$ ,  $\alpha_A > 0$  is investigated by using Monte Carlo simulation methods. We use the 0.95 quantile of the standard normal distribution and the empirical 0.95 quantile of the test statistic in the 10,000 simulations runs that are under the null hypothesis with different m and  $\Delta_B$ . We generated 1,000 processes under the alternative model where we took  $\alpha_A = \alpha_B = 0.1$  when  $\exp\{\beta_A\} = 2$  or 2.5. The power results are presented in Table 3.4 where entries are the proportions of the values of Z in 1,000 samples which are larger than the quantile values. This indicates that the power of the test is quite high overall. It can also easily seen that the power of the test increases as m increases.



Figure 3.5: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100, \Delta_B = 1$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 3.6: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100$ ,  $\Delta_B = 3$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 3.7: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100$ ,  $\Delta_B = 7$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 3.8: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100$ ,  $\Delta_B = 14$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100

$\Delta_B$	m	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	10	1.728	2.130	2.516	0.059	0.0340	0.0158
	20	1.721	2.079	2.508	0.056	0.0329	0.0159
	50	1.665	1.982	2.369	0.0516	0.0267	0.0110
	100	1.678	2.010	2.417	0.0544	0.0272	0.0123
3	10	1.705	2.036	2.388	0.0551	0.0301	0.0115
	20	1.735	2.083	2.476	0.0583	0.0319	0.0140
	50	1.646	1.971	2.394	0.0502	0.0258	0.0119
	100	1.624	1.947	2.290	0.0477	0.0245	0.0092
7	10	1.688	2.013	2.368	0.0548	0.0295	0.0116
	20	1.657	2.007	2.438	0.0513	0.0272	0.0125
	50	1.653	1.983	2.334	0.0508	0.026	0.0105
	100	1.691	1.982	2.413	0.0541	0.0266	0.0126
14	10	1.636	1.950	2.287	0.0490	0.0243	0.0095
	20	1.631	1.958	2.312	0.0490	0.0249	0.0249
	50	1.638	1.964	2.302	0.0493	0.0253	0.0097
	100	1.620	1.918	2.359	0.0475	0.0221	0.0109

Table 3.3:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m > 1 and  $\tau = 100$ .  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta$	m	$\hat{Pr}\{Z > 1.645   H_a\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}   H_a\}$	$\hat{Pr}\{Z > 1.645   H_a\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}   H_a\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 2.5$	$e^{\beta_A} = 2.5$
1	10	0.790	0.770	0.950	0.950
	20	0.970	0.960	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000
3	10	0.970	0.970	1.000	1.000
	20	1.000	1.000	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000
7	10	0.990	0.990	1.000	1.000
	20	1.000	1.000	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000
14	10	0.980	0.990	1.000	1.000
	20	1.000	1.000	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000

Table 3.4: Power of Z :  $m > 1, \Delta_A = 0$ 

We also considered the score statistic (3.54) where m = 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ . We generated 10,000 realizations of HPPs based on (3.41) and (3.42). We took  $\alpha_A = \alpha_B = 0.1$  and  $\beta_A = 0$ . We considered  $\Delta_B = 1, 3, 7$  and 14. For  $\Delta_B = 1$  in Figure 3.9, the normal approximation is not quite adequate, however with large  $\tau$  values, the approximation gets better. For  $\Delta_B = 3$  (Figure 3.10), the normal approximation is adequate when  $\tau = 1000$ . For  $\Delta_B = 7$  and 14 (Figures 3.11 and 3.12, respectively), the approximations are overall good. Table 3.4 shows estimated  $Q_p$  and  $\hat{Pr}(Z > Q_p)$ values with p = 0.950, 0.975 and 0.990, and this indicates also that the standard normal approximation is accurate for large  $\tau$ .

The power of the statistic (3.54) against the alternative hypothesis  $H_A$ :  $\beta_A \neq 0$ ,  $\alpha_A > 0$  is investigated by using Monte Carlo simulation methods as well. We used 0.95 quantile of the standard normal distribution and the empirical 0.95 quantile of the test statistic obtained from 10,000 simulations runs with different  $\tau$  and  $\Delta_B$  combinations. We generated 1,000 processes under the model (3.41) and (3.42) taking  $\alpha_A = \alpha_B = 0.1$  when  $\beta_A = 0.693$  or 1.098, where  $\exp\{0.693\} = 2$  and  $\exp\{1.098\} = 3$ . Table 3.6 provides the power results. Entries in Table 3.6 are the proportions of the values of Z in 1,000 samples which are larger than the quantile values. Table 3.6 shows that the power of the test is high overall, and power increases as  $\tau$  increases.



Figure 3.9: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 1$ ,  $\Delta_A > 0$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 3.10: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 3$ ,  $\Delta_A > 0$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 3.11: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 7$ ,  $\Delta_A > 0$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 3.12: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 14$ ,  $\Delta_A > 0$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 

$\Delta_B$	au	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	100	1.985	2.571	3.101	0.079	0.052	0.034
	200	1.868	2.317	2.874	0.071	0.044	0.025
	500	1.787	2.194	2.690	0.061	0.037	0.021
	1000	1.715	2.064	2.600	0.059	0.032	0.016
3	100	1.830	2.231	2.694	0.065	0.038	0.020
	200	1.769	2.168	2.620	0.062	0.036	0.018
	500	1.715	2.093	2.493	0.056	0.032	0.014
	1000	1.730	2.110	2.493	0.058	0.033	0.015
7	100	1.677	1.994	2.344	0.055	0.026	0.011
	200	1.679	2.033	2.345	0.053	0.028	0.011
	500	1.683	2.009	2.366	0.055	0.028	0.012
	1000	1.638	1.976	2.399	0.049	0.026	0.012
14	100	1.633	1.918	2.188	0.049	0.021	0.006
	200	1.620	1.917	2.244	0.047	0.022	0.008
	500	1.603	1.891	2.228	0.045	0.021	0.007
	1000	1.640	1.977	2.342	0.050	0.026	0.011

Table 3.5:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m = 1 and  $\Delta_A > 0$ .  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta$	au	$\hat{Pr}\{Z > 1.645   H_a\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}   H_a\}$	$\hat{Pr}\{Z > 1.645   H_a\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}   H_a\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 3$	$e^{\beta_A} = 3$
1	100	0.950	0.910	1.000	1.000
	200	0.950	0.940	1.000	1.000
	500	0.940	0.940	1.000	1.000
	1000	0.940	0.950	1.000	1.000
3	100	0.960	0.950	1.000	1.000
	200	0.940	0.940	1.000	1.000
	500	0.940	0.940	1.000	1.000
	1000	0.960	0.960	1.000	1.000
7	100	0.950	0.950	1.000	1.000
	200	0.950	0.940	1.000	1.000
	500	0.960	0.960	0.99	1.000
	1000	0.950	0.950	1.000	1.000
14	100	0.950	0.950	1.000	1.000
	200	0.930	0.930	1.000	1.000
	500	0.950	0.950	1.000	1.000
	1000	0.950	0.960	0.99	1.000

Table 3.6: Power of Z :  $m = 1, \Delta_A > 0$ 

Finally, we considered the score statistic (3.62) when m > 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ . We fixed  $\tau$  at 100, and generated 10,000 realizations of m HPPs taking  $\alpha_A = \alpha_B = 0.1$ ,  $\beta_A = 0$  for various m values with  $\Delta_B = 1, 3, 7$  and 14. We fixed  $\Delta_A$  at 1. Normal probability plots (Figures 3.13, 3.14, 3.15 and 3.16) show that normal approximations are quite suitable for overall scenarios with these various combinations with  $\Delta_B$  and m except for some cases in which  $\Delta_B = 1$ . Table 3.7 presents the value of estimated  $Q_p$  and  $\hat{Pr}(Z > Q_p)$  where p = 0.950, 0.975 and 0.990. This shows that for  $\Delta_B = 3, 7$  and 14, the standard normal approximations are suitable.

The power of the statistic (3.62) against the alternative hypothesis  $H_A : \beta_A \neq 0$ ,  $\alpha_A > 0$  is investigated by using Monte Carlo simulation study. To obtain the power, 0.95 quantile of the standard normal distribution and the empirical 0.95 quantile of the test statistic in the 10,000 simulations runs with different m and  $\Delta_B$  used. The power results are provided in Table 3.8. This shows that the power of the test is overall quite high.



Figure 3.13: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100, \Delta_A = 1, \Delta_B = 1$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 3.14: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100, \Delta_A = 1, \Delta_B = 3$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 3.15: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100, \Delta_A = 1, \Delta_B = 7$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 3.16: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100$ ,  $\Delta_A = 1$ ,  $\Delta_B = 14$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100

$\Delta_B$	m	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	10	1.781	2.111	2.540	0.061	0.034	0.015
	20	1.719	2.074	2.490	0.056	0.032	0.015
	50	1.675	1.996	2.426	0.052	0.028	0.012
	100	1.729	2.083	2.496	0.058	0.032	0.015
3	10	1.685	1.990	2.384	0.054	0.027	0.011
	20	1.668	1.990	2.396	0.052	0.027	0.012
	50	1.673	1.989	2.344	0.053	0.027	0.010
	100	1.647	1.996	2.419	0.050	0.026	0.013
7	10	1.650	1.953	2.350	0.050	0.024	0.011
	20	1.669	1.983	2.406	0.053	0.026	0.012
	50	1.653	1.950	2.350	0.051	0.024	0.010
	100	1.650	1.985	2.385	0.051	0.026	0.012
14	10	1.633	1.950	2.340	0.049	0.024	0.011
	20	1.681	2.005	2.369	0.054	0.028	0.011
	50	1.639	1.963	2.327	0.050	0.025	0.010
	100	1.685	2.032	2.369	0.055	0.029	0.011

Table 3.7:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m > 1,  $\Delta_A = 1$  and  $\tau = 100$ .  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta_B$	m	$\hat{Pr}\{Z > 1.645   H_a\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}   H_a\}$	$\hat{Pr}\{Z > 1.645   H_a\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}   H_a\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 3$	$e^{\beta_A} = 3$
1	10	0.830	0.850	0.990	0.990
	20	0.950	0.960	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000
3	10	0.940	0.940	1.000	1.000
	20	1.000	1.000	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000
7	10	0.980	0.980	1.000	1.000
	20	1.000	1.000	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000
14	10	0.970	0.970	1.000	1.000
	20	1.000	1.000	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000

Table 3.8: Power of Z :  $m > 1, \Delta_A = 1$ 

# Chapter 4

# Redundant Systems with Three Components

In this chapter, as an extension of Chapter 3, we consider repairable systems with three components working in parallel; a redundant component and two primary components. In Section 4.2, we develop a test of parallel carryover effects in redundant systems with three components. We present simulation results in Section 4.3. In Section 4.4, we illustrate methods by analyzing a failure data set from diesel operated power plants.

## 4.1 Introduction

Redundant systems may include more than two components working in parallel. In this section, we discuss testing for the presence of parallel carryover effects in redundant systems with three components. We develop partial score tests of parallel carryover effects in such systems, and discuss asymptotic properties of them through simulations.

# 4.2 Models and Tests for Parallel Carryover Effects

In this section, we consider three components in a system, instead of two. We discuss models for a single system and multiple systems under two different settings based on the duration of repairs in the redundant component. Suppose that there is a multivariate counting processes  $\{N_A(t), N_B(t), N_C(t); t \ge 0\}$ , where  $\{N_A(t); t \ge 0\}$  is a counting process for Component A,  $\{N_B(t); t \ge 0\}$  is a counting process for Component B and  $\{N_C(t); t \ge 0\}$  is a counting process for Component C. The three components are working in parallel in a system. We let  $t_{A1}, t_{A2}, \ldots$ , where  $0 < t_{A1} < t_{A2} < \ldots, t_{B1}, t_{B2}, \ldots$ , where  $0 < t_{B1} < t_{B2} < \ldots$ , and  $t_{C1}, t_{C2}, \ldots$ , where  $0 < t_{C1} < t_{C2} < \ldots$ , denote the failure times of Components A, B and C, respectively. All components are subject to repairs and repair times in the primary components cannot be ignored. Let  $\Delta_A$ ,  $\Delta_B$  and  $\Delta_C$  denote the repair times of Components A, B and C, respectively. For K = A, B and C,  $Y_K(t)$  is the *at-risk indicator* of process  $\{N_K(t); t > 0\}$ .

A model including parallel carryover effects for Component A in a redundant system having three components is given by

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t)\,\alpha_A\,\exp\{\beta_A X_{AB}(t) + \beta_A X_{AC}(t)\}, \quad t > 0, \tag{4.1}$$

where  $Y_A(t)$  is the at-risk indicator function of Component A,  $\alpha_A > 0$  is a baseline rate function, and

$$X_{AB}(t) = I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \le \Delta_B\},$$
(4.2)

and

$$X_{AC}(t) = I\{N_C(t^-) > 0\}I\{t - t_{CN_C(t^-)} \le \Delta_C\},$$
(4.3)

and  $\beta_A$  is a regression parameter. Notice that, for  $K = B, C, X_{AK}(t)$  takes the value of 1 while Component K is under a repair; otherwise, it equals 0. Also, we assume that one parameter  $\beta_A$  represents the effect of repairs of Component B or C on Component A. This assumption is applicable in many settings, but models can be extended by including separate parameters for the effects of the repairs of Components B and C. For simplicity, we do not pursue this case in this thesis. However, if there is a need for more detailed modelling, methods in this section can be applied after obvious modifications.

While Component A is in the up state at time t (i.e.,  $Y_A(t) = 1$ ), there are four possibilities: (i) Both Components B and C are in the up state, (ii) Component B is in the down state and Component C is in the up state, (iii) Component B is in the up state and Component C is in the down state, and (iv) both Components B and C are in the down state. In case (i), failures of Component A are governed by a HPP with
rate function  $\alpha_A$ . In case (*ii*), the intensity function of Component A jumps from  $\alpha_A$  to  $Y_A(t) \alpha_A \exp{\{\beta_A\}}$  after each failure in Component B, and stays there for  $\Delta_B$  time units as long as there is no failures in Component A (i.e.,  $Y_A(t) = 1$ ) and Component C (i.e.,  $Y_C(t) = 1$ ). Similarly, in case (*iii*), the intensity function of Component A jumps from  $\alpha_A$  to  $Y_A(t) \alpha_A \exp{\{\beta_A\}}$  after each failure in Component C, and stays there for  $\Delta_C$  time units as long as there is no failures in Component A (i.e.,  $Y_A(t) = 1$ ) and Component B (i.e.,  $Y_B(t) = 1$ ). In the last case, when both Components B and C are in the down state, the intensity function (4.1) becomes  $\alpha_A \exp{\{2\beta_A\}}$ , and stays there as long as there is no failures in Component A (i.e.,  $Y_A(t) = 1$ ) and both Components B and C are as long as there is no failures in Component A (i.e.,  $Y_A(t) = 1$ ) and both Components B and C are under repair ( $Y_B(t) = 0$  and  $Y_C(t) = 0$ ).

A model for parallel carryover effects can be also similarly defined for Component B. In this case, the intensity function of  $\{N_B(t); t \ge 0\}$  is given by

$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \,\alpha_B \,\exp\{\beta_B X_{BA}(t) + \beta_B X_{BC}(t)\}, \quad t > 0, \tag{4.4}$$

where  $\alpha_B > 0$  is a baseline rate function, and

$$X_{BA}(t) = I\{N_A(t^-) > 0\}I\{t - t_{AN_A(t^-)} \le \Delta_A\},$$
(4.5)

and

$$X_{BC}(t) = I\{N_C(t^-) > 0\}I\{t - t_{CN_C(t^-)} \le \Delta_A\}.$$
(4.6)

Similarly, the intensity function of  $\{N_C(t); t \ge 0\}$  is given by

$$\lambda_C(t|\mathcal{H}(t)) = Y_C(t) \,\alpha_C \,\exp\{\beta_C X_{CA}(t) + \beta_C X_{CA}(t)\}, \quad t > 0, \tag{4.7}$$

where  $\alpha_C > 0$  is a baseline rate function, and

$$X_{CA}(t) = I\{N_A(t^-) > 0\}I\{t - t_{AN_A(t^-)} \le \Delta_A\},$$
(4.8)

and

$$X_{CB}(t) = I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \le \Delta_B\}.$$
(4.9)

Let *m* denote the number of systems, each with three components. In the remaining part of this section, we develop partial score tests for the presence of parallel carryover effects in four different cases; (*i*) a single system is under observation and repair times of Components A and B are negligible ( $m = 1, \Delta_A = 0, \Delta_B = 0$ ,  $\Delta_C > 0$ ), (*ii*) a single system is under observation and repair times of Component A are negligible (m = 1,  $\Delta_A = 0$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ ), (*iii*) a single system is under observation and repair times of Component A are not negligible (m = 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ ), and (*iv*) multiple systems are under observation and repair times of Component A are not negligible (m > 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ ). In all cases, we assume the baseline rate functions are constants.

#### **4.2.1** Case 1: $m = 1, \Delta_A = 0, \Delta_B = 0, \Delta_C > 0$

We first consider a simple case in a system with 3 components connected in parallel; Components A, B and C. In this setting, we assume that repair times of Components A and B are negligible; that is  $\Delta_A = 0$  and  $\Delta_B = 0$  so that failures of Component A and failures of Component B do not affect the probabilistic characteristics of failure occurrences in Component C. We assume also that failure occurrences are governed by HPPs. In this case, the intensity functions for Components A, B and C are given by

Component A: 
$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \alpha_A \exp\{\beta_A X_A(t)\}, \quad t > 0,$$
 (4.10)

where  $Y_A(t)$  is at-risk indicator function  $\alpha_A > 0$  is a baseline rate function and  $X_A(t) = I\{N_C(t^-) > 0\}I\{t - T_{N_C(t^-)} \leq \Delta_C\}$ , and

Component B: 
$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \alpha_B \exp\{\beta_B X_B(t)\}, \quad t > 0,$$
 (4.11)

where  $Y_B(t)$  is at-risk indicator function  $\alpha_B > 0$  is a baseline rate function and  $X_B(t) = I\{N_C(t^-) > 0\}I\{t - T_{N_C(t^-)} \leq \Delta_C\}$ , and since  $\Delta_A = 0$  and  $\Delta_B = 0$ 

Component C: 
$$\lambda_C(t|\mathcal{H}(t)) = Y_C(t) \alpha_C, \quad t > 0,$$
 (4.12)

where  $\alpha_C > 0$  is a baseline rate function and  $\mathcal{H}(t) = \{N_A(u), N_B(u), N_A(u), Y_C(s); 0 \le u < t, 0 \le s \le t\}.$ 

A test for the presence of parallel carryover effects in Component A can be developed by considering the following composite hypothesis.

$$H_0: \beta_A = 0, \alpha_A > 0, \quad \text{vs.} \quad H_a: \beta_A \neq 0, \alpha_A > 0,$$
 (4.13)

where  $\beta_A$  is the parameter of interest and  $\alpha_A$  is a nuisance parameter. We suppose that the system is under observation over the followup period  $[0, \tau]$ , where  $\tau$  is the end-of-followup time. Note that, since  $\Delta_A = 0$  and  $\Delta_B = 0$ , we can safely drop at-risk indicator functions  $Y_A(t)$  and  $Y_B(t)$  from the model (4.10) and (4.11), respectively. Let  $n_A$ , where  $n_A \ge 0$ , denote the number of failures of Component A over  $[0, \tau]$ , and  $t_{A1}, t_{A2}, \ldots, t_{An_A}$  be the failure times of Component A in  $[0, \tau]$ . Then the likelihood function of the outcome " $N_A(\tau) = n_A$  failures of Component A at times  $t_{A1} \le t_{A2} \le$  $\cdots \le t_{An_A}$  in  $[0, \tau]$ " is given by

$$L(\boldsymbol{\theta}) = \prod_{j=1}^{n_A} \alpha_A \exp\{\beta_A X_A(t_{Aj})\} \exp\{-\int_0^\tau \alpha_A \exp\{\beta_A X_A(s)\}\,ds\},\tag{4.14}$$

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A)$ . The likelihood function (4.14) is of the same form with the likelihood function (3.9) given in Section 3.2.1, and thus, the partial score test statistic based on (4.14) is the same with the one obtained in Section 3.2.1. We only state the form of the test statistic here, but not discuss its derivation. Let

$$\mathcal{I}(\tau,\beta_A,\Delta_B) = \int_0^\tau X_A(s) \exp\{\beta_A X_A(s)\} \, ds.$$
(4.15)

Then, the partial score test for testing the null hypothesis  $H_0: \beta_A = 0$  is given by

$$Z = \frac{U_{\beta}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},$$
(4.16)

where

$$U_{\beta_A}(\tilde{\alpha}_A, 0) = \sum_{j=1}^{n_A} X_A(t_j) - \tilde{\alpha}_A \mathcal{I}(\tau, 0, \Delta_B), \qquad (4.17)$$

and

$$\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0)) = \mathcal{I}(\tau, 0, \Delta_B) \left[\tau - \mathcal{I}(\tau, 0, \Delta_B)\right].$$
(4.18)

The asymptotic properties of the test statistic (4.16) are discussed in Section 3.3 analytically and in Section 3.4 through simulations.

### **4.2.2** Case 2: $m = 1, \Delta_A = 0, \Delta_B > 0, \Delta_C > 0$

In this section, we consider a case in which there is a single system under observation and the repair times are negligible for Component A, but not negligible for Components B and C. This means that m = 1 and  $\Delta_A = 0$ ,  $\Delta_B > 0$  and  $\Delta_C > 0$ .

The intensity functions of Components A, B and C are respectively given as follows.

Component A: 
$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \alpha_A \exp\{\beta_A X_{AB}(t) + \beta_A X_{AC}(t)\}, \quad t > 0,$$

$$(4.19)$$

where  $Y_A(t)$  is the at-risk indicator of Component A,  $X_{AB}(t)$  and  $X_{AC}(t)$  are respectively defined in (4.2) and (4.3),  $\beta_A$  is a regression parameter,  $\Delta_A = 0$ , and  $\alpha_A > 0$  is a baseline rate function.

Component B: 
$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \alpha_B \exp\{\beta_B X_{BC}(t)\}, \quad t > 0,$$
 (4.20)

where  $Y_B(t)$  is the at-risk indicator function of Component B,  $\Delta_B > 0$ ,  $X_{BC}(t)$  is defined in (4.6), and  $\alpha_B > 0$  is a baseline rate function.

Component C: 
$$\lambda_C(t|\mathcal{H}(t)) = Y_C(t) \alpha_C \exp\{\beta_C X_{CB}(t)\}, \quad t > 0,$$
 (4.21)

where  $Y_C(t)$  is the at-risk indicator function of Component C,  $X_{CB}(t)$  is defined in (4.9),  $\Delta_C > 0$ , and  $\alpha_C > 0$  is a baseline rate function and  $\mathcal{H}(t) = \{N_A(u), N_B(u), N_C(u), Y_B(s), Y_C(s); 0 \le u < t, 0 \le s \le t\}.$ 

Then the likelihood function of the outcome that " $N_A(\tau) = n_A$  failures of Component A occur at times  $t_{A1} \leq t_{A2} \leq \cdots \leq t_{An_A}$  in  $[0, \tau]$ " is given by

$$L(\boldsymbol{\theta}) = \prod_{j=1}^{n_A} \alpha_A \, e^{\beta_A X_{AB}(t_{Aj}) + \beta_A X_{AC}(t_{Aj})} \, \exp\{-\int_0^\tau \alpha_A \, e^{\beta_A X_{AB}(s) + \beta_A X_{AC}(s)} \, ds\}.$$
(4.22)

Then the log likelihood function is given by

$$\ell(\boldsymbol{\theta}) = n_A \log \alpha_A + \sum_{j=1}^{n_A} \{\beta_A X_{AB}(t_{Aj}) + \beta_A X_{AC}(t_{Aj})\} - \alpha_A \int_0^\tau e^{\beta_A X_{AB}(s) + \beta_A X_{AC}(s)} ds.$$
(4.23)

The score vector is then defined by  $\boldsymbol{U}(\boldsymbol{\theta}) = (U_{\alpha_A}(\boldsymbol{\theta}), U_{\beta_A}(\boldsymbol{\theta}))'$  with components

$$U_{\alpha_A}(\theta) = \frac{n_A}{\alpha_A} - \int_0^\tau e^{\beta_A(X_{AB}(s) + X_{AC}(s))} \, ds, \qquad (4.24)$$

and

$$U_{\beta_{A}}(\boldsymbol{\theta}) = \sum_{j=1}^{n_{A}} X_{AB}(t_{Aj}) + \sum_{j=1}^{n_{A}} X_{AC}(t_{Aj}) - \alpha_{A} \int_{0}^{\tau} (X_{AB}(s) + X_{AC}(s)) e^{\beta_{A}(X_{AB}(s) + X_{AC}(s))} ds.$$
(4.25)

Also, the observed information matrix  $I(\theta)$  is given by

$$\boldsymbol{I}(\boldsymbol{\theta}) = \begin{pmatrix} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) & I_{\alpha_A \beta_A}(\boldsymbol{\theta}) \\ I_{\beta_A \alpha_A}(\boldsymbol{\theta}) & I_{\beta_A \beta_A}(\boldsymbol{\theta}) \end{pmatrix}, \qquad (4.26)$$

where the components are given by

$$I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) = \frac{n_A}{\alpha_A^2},\tag{4.27}$$

$$I_{\alpha_A\beta_A}(\boldsymbol{\theta}) = I_{\beta_A\alpha_A}(\boldsymbol{\theta}) = \int_0^\tau (X_{AB}(s) + X_{AC}(s)) e^{\beta_A X_{AB}(s) + \beta_A X_{AC}(s)} ds, \qquad (4.28)$$

$$I_{\beta_A\beta_A}(\boldsymbol{\theta}) = \alpha_A \int_0^\tau (X_{AB}(s) + X_{AC}(s))^2 e^{\beta_A X_{AB}(s) + \beta_A X_{AC}(s)} \, ds.$$
(4.29)

Let  $\tilde{\alpha}_A$  be the restricted maximum likelihood estimator of  $\alpha_A$  under  $H_0: \beta_A = 0$ . By solving  $U_{\alpha_A}(\alpha_A, 0) = 0$  in (4.24) with  $\beta_A = 0$  for  $\alpha_A = \tilde{\alpha}_A$ , we obtain

$$\tilde{\alpha}_A = \frac{n_A}{\tau} \tag{4.30}$$

Therefore, the standardized partial score statistic for testing the hypotheses (4.13) is given by

$$Z = \frac{U_{\beta_A}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},\tag{4.31}$$

where

$$U_{\beta}(\tilde{\alpha}_{A},0) = \sum_{j=1}^{n_{A}} X_{AB}(t_{Aj}) + \sum_{j=1}^{n_{A}} X_{AC}(t_{Aj}) - \tilde{\alpha}_{A} \int_{0}^{\tau} (X_{AB}(s) + X_{AC}(s)) ds, \quad (4.32)$$

$$\widehat{Var}(U_{\beta}(\tilde{\alpha}_{A},0)) = I_{\beta_{A}\beta_{A}}(\tilde{\alpha}_{A},0) - I_{\beta_{A}\alpha_{A}}(\tilde{\alpha}_{A},0)I_{\alpha_{A}\alpha_{A}}^{-1}(\tilde{\alpha}_{A},0)I_{\alpha_{A}\beta_{A}}(\tilde{\alpha}_{A},0)$$
(4.33)

$$= \frac{n_A}{\tau^2} \left[ \tau \int_0^\tau (X_{AB}(s) + X_{AC}(s))^2 ds - \left[ \int_0^\tau (X_{AB}(s) + X_{AC}(s)) \, ds \right]^2 \right]$$
(4.34)

We discuss the asymptotic properties of this test statistic in Section 4.3 through simulations. However, a proof based on the method in Section 3.3 can be useful to show its asymptotic distribution analytically.

#### **4.2.3** Case 3: $m = 1, \Delta_A > 0, \Delta_B > 0, \Delta_C > 0$

This section presents the derivation of a test statistics developed for testing the presence of parallel carryover effects in redundant systems. In this case, there is a single system under observation and the repair times are not negligible for Components A, B and C; that is, m = 1 and  $\Delta_A > 0$ ,  $\Delta_B > 0$  and  $\Delta_C > 0$ . Component A is the redundant component in the system.

In this case, the intensity functions of Components A, B and C are given below:

Component A: 
$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \alpha_A \exp\{\beta_A X_{AB}(t) + \beta_A X_{AC}(t)\}, \quad t > 0,$$

$$(4.35)$$

where  $Y_A(t)$  is the at-risk indicator function of Component A,  $X_{AB}(t)$  and  $X_{AC}(t)$ are respectively defined in (4.2) and (4.3),  $\alpha_A > 0$  is a baseline rate function,  $\beta_A$  is a regression parameter.

Component B: 
$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \alpha_B \exp\{\beta_B X_{BC}(t)\}, \quad t > 0,$$
 (4.36)

where  $Y_B(t)$  is the at-risk indicator function of Component B,  $X_{BC}(t)$  is defined in (3.6),  $\alpha_B > 0$  is a baseline rate function, and  $\beta_B$  is a regression parameter.

Component C: 
$$\lambda_C(t|\mathcal{H}(t)) = Y_C(t) \alpha_C \exp\{\beta_C X_{CB}(t)\}, \quad t > 0.$$
 (4.37)

where  $Y_C(t)$  is the at-risk indicator function of Component C,  $X_{CB}(t)$  is defined in (4.9),  $\alpha_C > 0$  is a baseline rate function, and  $\beta_C$  is a regression parameter. The history of the processes is given by  $\mathcal{H}(t) = \{N_A(u), N_B(u), N_C(u), Y_A(s), Y_B(s), Y_C(s); 0 \leq 0\}$ 

and

 $u < t, 0 \le s \le t \}.$ 

As explained previously in Section 4.2, in this case the intensity function (4.35) jumps from  $\alpha_A$  to  $\alpha_A \exp\{\beta_A\}$  when either one of Component B or Component C fails. Then, it increases from  $\alpha_A \exp\{\beta_A\}$  to  $\alpha_A \exp\{2\beta_A\}$  when both of Components B and C fail.

In this case, the likelihood function of the outcome that " $N_A(\tau) = n_A$  failures of Component A occur at times  $t_{A1} \leq t_{A2} \leq \cdots \leq t_{An_A}$  in  $[0, \tau]$ , where  $\tau$  is fixed" is given by

$$L(\boldsymbol{\theta}) = \prod_{j=1}^{n_A} \alpha_A \, e^{\beta_A X_{AB}(t_{Aj}) + \beta_A X_{AC}(t_{Aj})} \, \exp\{-\int_0^\tau Y_A(s) \, \alpha_A \, e^{\beta_A X_{AB}(s) + \beta_A X_{AC}(s)} \, ds\},$$
(4.38)

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A)$ . Then, the log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$n_A \log \alpha_A + \sum_{j=1}^{n_A} \{\beta_A X_{AB}(t_{Aj}) + \beta_A X_{AC}(t_{Aj})\} - \alpha_A \int_0^\tau Y_A(s) \, e^{\beta_A X_{AB}(s) + \beta_A X_{AC}(s)} \, ds.$$
(4.39)

The components of the score vector  $\boldsymbol{U}(\boldsymbol{\theta})$  are given by

$$U_{\alpha_A}(\boldsymbol{\theta}) = \frac{n_A}{\alpha_A} - \int_0^\tau Y_A(s) \, e^{\beta_A(X_{AB}(s) + X_{AC}(s))} \, ds, \tag{4.40}$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{j=1}^{n_A} X_{AB}(t_{Aj}) + \sum_{j=1}^{n_A} X_{AC}(t_{Aj}) - \alpha_A \int_0^\tau Y_A(s) \left( X_{AB}(s) + X_{AC}(s) \right) e^{\beta_A (X_{AB}(s) + X_{AC}(s))} \, ds.$$
(4.41)

By solving  $U_{\alpha_A}(\alpha_A, 0) = 0$  in (4.40) with  $\beta_A = 0$ , we can easily show that the restricted maximum likelihood estimator  $\tilde{\alpha}_A$  of  $\alpha_A$  is given as  $\tilde{\alpha}_A = n_A / (\int_0^{\tau} Y_A(s) \, ds)$ .

For testing the presence of parallel carryover effects in Component A, we consider the hypothesis (4.13). The partial score function under the null hypothesis  $H_0$ :  $\beta_A = 0$ can be written as follows.

$$U_{\beta}(\tilde{\alpha}_{A},0) = \sum_{j=1}^{n_{A}} \left( X_{AB}(t_{Aj}) + X_{AC}(t_{Aj}) \right) - \tilde{\alpha}_{A} \int_{0}^{\tau} Y_{A}(s) \left( X_{AB}(s) + X_{AC}(s) \right) ds.$$
(4.42)

The components of the observed information matrix  $I(\alpha_A, \beta_A)$  evaluated at  $(\alpha_A, \beta_A) = (\tilde{\alpha}_A, 0)$  are given by

$$I_{\alpha_A \alpha_A}(\tilde{\alpha}_A, 0) = \frac{n_A}{\tilde{\alpha}_A^2},\tag{4.43}$$

$$I_{\alpha_A\beta_A}(\tilde{\alpha}_A, 0) = I_{\beta_A\alpha_A}(\tilde{\alpha}_A, 0) = \int_0^\tau Y_A(s) \left( X_{AB}(s) + X_{AC}(s) \right) ds, \tag{4.44}$$

$$I_{\beta_A\beta_A}(\tilde{\alpha}_A, 0) = \tilde{\alpha}_A \int_0^\tau Y_A(s) \left( X_{AB}(s) + X_{AC}(s) \right)^2 ds.$$
(4.45)

From the score procedures explained in Chapter 1, we obtain the following standardized partial score statistic for testing the presence of parallel carryover effects in Component A.

$$Z = \frac{U_{\beta_A}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},\tag{4.46}$$

where  $U_{\beta}(\tilde{\alpha}_A, 0)$  is given in (3.42) and  $\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))$  is given by

$$\frac{n_A}{(\int_0^\tau Y_A(s)\,ds)^2} \left[ \left( \int_0^\tau Y_A(s)\,ds \right) \mathcal{I}_1(\tau,\beta_A,\Delta_B,\Delta_C) - \mathcal{I}_2(\tau,\beta_A,\Delta_B,\Delta_C) \right], \quad (4.47)$$

with

$$\mathcal{I}_1(\tau, \beta_A, \Delta_B, \Delta_C) = \int_0^\tau Y_A(s) \left( X_{AB}(s) + X_{AC}(s) \right)^2 ds, \qquad (4.48)$$

and

$$\mathcal{I}_2(\tau,\beta_A,\Delta_B,\Delta_C) = \left[\int_0^\tau Y_A(s) \left(X_{AB}(s) + X_{AC}(s)\right) ds\right]^2.$$
(4.49)

Asymptotic properties of the test statistic (4.46) are discussed in Section 4.3 through simulations. Under some mild regularity conditions on the at-risk indicator  $Y_A(t)$ (see, Karr, 1991, p. 421), the asymptotic distribution of the test statistic (4.46) can be analytically shown with a similar method given in Section 3.3.

### **4.2.4** Case 4: $m > 1, \Delta_A > 0, \Delta_B > 0, \Delta_C > 0$

In this section, we consider a similar setting of Section 4.2.3, but here we assume that m independent systems are under observation, each with three components: Component A, B and C. The repair times of the components are not negligible; that is,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ . Furthermore, we assume that failures of Components B and C affect the probabilistic characteristics of failure occurrences of Component A, while failures of Component A do not affect the probabilistic characteristics of failure occurrences of Components B and C. Therefore, Component A operates as a redundant component in the system.

Now, suppose that there are *m* independent multivariate counting processes  $\{N_{Ai}(t), N_{Bi}(t), N_{Ci}(t); t \ge 0\}$ , i = 1, ..., m, where  $\{N_{Ki}(t); t \ge 0\}$ , for K = A, B or *C*, is a counting process of failure occurrences in Component A in the *i*th system, i = 1, ..., m. For K = A, B or *C*, we let  $t_{Ki1}, t_{Ki2}, ...,$  where  $0 < t_{Ki1} < t_{Ki2} < ...$ , denote the failure times of the *K*th component in the *i*th system, i = 1, ..., m. Let  $\Delta_{Ai}$ ,  $\Delta_{Bi}$  and  $\Delta_{Ci}$  denote the repair times of Components A, B and C in the *i*th system, respectively. For K = A, B or *C*, the at-risk indicator  $Y_{Ki}(t)$  of the process  $\{N_{Ki}(t); t > 0\}$  is under observation; otherwise, it equals to 0. The history of the multivariate processes  $\{N_{Ai}(t), N_{Bi}(t), N_{Ci}(t); t \ge 0\}$ , i = 1, ..., m, is denoted by  $\mathcal{H}_i(t) = \{N_{Ai}(u), N_{Bi}(u), N_{Ci}(u); 0 \le u < t\}$ .

In this case, the intensity functions of Components A, B and C in the ith system are given as follows.

Component A: 
$$\lambda_{Ai}(t|\mathcal{H}_i(t)) = Y_{Ai}(t) \alpha_A \exp\{\beta_A X_{ABi}(t) + \beta_A X_{ACi}(t)\}, \quad t > 0,$$

$$(4.50)$$

where  $Y_{Ai}(t)$  is the at-risk indicator function of Component A in *i*th system,  $\alpha_A > 0$  is a baseline rate function,  $\beta_A$  is a regression parameter, and

$$X_{ABi}(t) = I\{N_{Bi}(t^{-}) > 0\}I\{t - t_{BiN_{Bi}(t^{-})} \le \Delta_{Bi}\},$$
  

$$X_{ACi}(t) = I\{N_{Ci}(t^{-}) > 0\}I\{t - t_{CiN_{Ci}(t^{-})} \le \Delta_{Ci}\}.$$
(4.51)

Component B: 
$$\lambda_{Bi}(t|\mathcal{H}_i(t)) = Y_{Bi}(t) \alpha_B \exp\{\beta_B X_{BCi}(t)\}, \quad t > 0,$$
 (4.52)

where  $Y_{Bi}(t)$  is the at-risk indicator function of Component B in the *i*th system,  $\alpha_B > 0$  is a baseline rate function, and

$$X_{BCi}(t) = I\{N_{Ci}(t^{-}) > 0\}I\{t - t_{CiN_{Ci}(t^{-})} \le \Delta_{Ci}\}.$$
(4.53)

Component C:  $\lambda_{Ci}(t|\mathcal{H}_i(t)) = Y_{Ci}(t) \alpha_C \exp\{\beta_C X_{CBi}(t)\}, \quad t > 0, \quad (4.54)$ 

where  $Y_{Ci}(t)$  is the at-risk indicator function of Component C in the *i*th system,

 $\alpha_C > 0$  is a baseline rate function, and

$$X_{CBi}(t) = I\{N_{Bi}(t^{-}) > 0\}I\{t - T_{BiN_{Bi}(t^{-})} \le \Delta_{Bi}\}.$$
(4.55)

By considering the following composite hypothesis, a test of the presence of parallel carryover effects in Component A can be developed.

$$H_0: \beta_A = 0, \alpha_A > 0, \quad \text{vs.} \quad H_a: \beta_A \neq 0, \alpha_A > 0,$$
 (4.56)

where  $\alpha_A$  is a nuisance parameter and  $\beta_A$  is the parameter of interest.

We suppose that m systems are independent and under observation period over  $[0, \tau_i]$ , where  $\tau_i$  is the fixed end-of-followup time of the *i*th system, i = 1, 2, ..., m. We let  $t_{Ai1}, t_{Ai1}, ..., t_{Ain_{Ai}}$  be the failure times of Component A in the *i*th system, and let  $N_{Ai}(\tau_i) = n_{Ai}$  be the observed number of failures of Component A in the *i*th system over the time interval  $[0, \tau_i]$ . The likelihood function of the outcome that " $N_{Ai}(\tau_i) = n_{Ai}$  failures of Component A observed at times  $t_{Ai1}, t_{Ai1}, ..., t_{Ain_{Ai}}$  in  $[0, \tau_i]$ " is given by

$$L(\boldsymbol{\theta}) = \prod_{i=1}^{m} L_i(\boldsymbol{\theta}), \qquad (4.57)$$

where

$$L_{i}(\boldsymbol{\theta}) = \prod_{j=1}^{n_{Ai}} \alpha_{A} e^{\beta_{A} X_{ABi}(t_{Aij}) + \beta_{A} X_{ACi}(t_{Aij})} \exp\{-\int_{0}^{\tau_{i}} Y_{Ai}(s) \alpha_{A} e^{\beta_{A} X_{ABi}(s) + \beta_{A} X_{ACi}(s)} ds\}.$$
(4.58)

Then, the log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$\sum_{i=1}^{m} n_{A_i} \log \alpha_A + \beta_A \sum_{i=1}^{m} \sum_{j=1}^{n_{A_i}} (X_{ABi}(t_{Aij}) + X_{ACi}(t_{Aij})) - \alpha_A \sum_{i=1}^{m} \int_0^{\tau_i} Y_{Ai}(s) e^{\beta_A X_{ABi}(s) + \beta_A X_{ACi}(s)} ds,$$
(4.59)

The components of the score vector  $U(\boldsymbol{\theta})$  are followed by

$$U_{\alpha_A}(\boldsymbol{\theta}) = \frac{\sum_{i=1}^m n_{A_i}}{\alpha_A} - \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, e^{\beta_A X_{ABi}(s) + \beta_A X_{ACi}(s)} \, ds, \qquad (4.60)$$

and

$$U_{\beta_{A}}(\boldsymbol{\theta}) = \sum_{i=1}^{m} \sum_{j=1}^{n_{A_{i}}} (X_{ABi}(t_{Aij}) + X_{ACi}(t_{Aij})) - \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \left( X_{ABi}(s) + \beta_{A} X_{ACi} \right) e^{\beta_{A} X_{ABi}(s) + \beta_{A} X_{ACi}(s)} ds.$$
(4.61)

Also, the components of the observed information matrix  $I(\theta)$  are given by

$$I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) = \frac{\sum_{i=1}^m n_{Ai}}{\alpha_A^2},\tag{4.62}$$

$$I_{\alpha_A\beta_A}(\boldsymbol{\theta}) = I_{\beta_A\alpha_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \left( X_{ABi}(s) + X_{ACi}(s) \right) e^{\beta_A X_{ABi}(s) + \beta_A X_{ACi}(s)} \, ds,$$
(4.63)

$$I_{\beta_A\beta_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \left( X_{ABi}(s) + X_{ACi}(s) \right)^2 e^{\beta_A X_{ABi}(s) + \beta_A X_{ACi}(s)} \, ds. \tag{4.64}$$

Under the null hypothesis  $H_0: \beta_A = 0$ , we obtain the restricted maximum likelihood estimator  $\tilde{\alpha}_A$  of  $\alpha_A$  as  $\tilde{\alpha}_A = (\sum_{i=1}^m n_{A_i}) / \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, ds$ . Then, from the score function  $U_{\beta_A}(\boldsymbol{\theta})$  given in (4.61), we can write the partial score function evaluated at  $(\alpha_A, \beta_A) = (\tilde{\alpha}_A, 0)$  as follows.

$$U_{\beta_A}(\tilde{\alpha}_A, 0) = \sum_{i=1}^m \sum_{j=1}^{n_{A_i}} (X_{ABi}(t_{Aij}) + X_{ACi}(t_{Aij})) - \tilde{\alpha}_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \left( X_{ABi}(s) + \beta_A X_{ACi} \right) ds$$
(4.65)

We define the integrals  $\mathcal{I}_1(m, \beta_A, \Delta_B, \Delta_C)$  and  $\mathcal{I}_2(m, \beta_A, \Delta_B, \Delta_C)$ , for notational convenience, as

$$\mathcal{I}_1(m,\beta_A,\Delta_B,\Delta_C) = \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \left( X_{ABi}(s) + X_{ACi}(s) \right)^2 ds,$$
(4.66)

and

$$\mathcal{I}_{2}(m,\beta_{A},\Delta_{B},\Delta_{C}) = \left(\sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \left(X_{ABi}(s) + X_{ACi}(s)\right) ds\right)^{2}.$$
 (4.67)

Then, the estimated variance of  $U_{\beta_A}(\tilde{\alpha}_A, 0)$ , which is denoted by  $\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0))$ , can

be written as

$$\frac{\sum_{i=1}^{m} n_{A_i}}{(\sum_{i=1}^{m} \int_0^{\tau_i} Y_{A_i}(s) \, ds)^2} \left[ \left( \sum_{i=1}^{m} \int_0^{\tau_i} Y_{A_i}(s) \, ds \right) \mathcal{I}_1(m, \beta_A, \Delta_B, \Delta_C) - \mathcal{I}_2(m, \beta_A, \Delta_B, \Delta_C) \right]$$

$$(4.68)$$

With the partial score function (4.65) and its estimated variance (4.68), we obtain the standardized partial score statistic for testing  $H_0: \beta_A = 0$  as

$$Z = \frac{U_{\beta}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}}.$$
(4.69)

Under some mild convergence conditions on the at-risk indicators  $Y_{Ai}$ , i = 1, ..., m, (see, Andersen et al., 1993, pp. 426–427 for an example), we can obtain the asymptotic distribution of the test statistic (4.69) as  $m \to \infty$ . We discuss some of the large sample properties of the test statistic (4.69) in Section 4.3 through simulations.

### 4.3 Simulation Studies

In this section, we present the results of simulation studies conducted to assess when asymptotic normal approximation for the test statistics developed in Section 4.2.2, Section 4.2.3 and Section 4.2.4 are satisfactory. We consider two settings (i) m = 1and  $\tau \to \infty$  and  $(ii) m \to \infty$  and  $\tau$  fixed.

We first consider the setting given in Section 4.2.2; that is, Case 2: m = 1,  $\Delta_A = 0$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ . The hypothesis of no parallel carryover effects in Component A is  $H_0$ :  $\beta_A = 0$ ,  $\alpha_A > 0$  and the test statistic for testing the null hypothesis is

$$Z = \frac{U_{\beta_A}(\tilde{\alpha}_A, 0)}{\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}},\tag{4.70}$$

where the score statistic  $U_{\beta_A}(\tilde{\alpha}_A, 0)$  is given by (4.32) and its estimated variance  $\widehat{Var}(U_{\beta_A}(\tilde{\alpha}_A, 0))^{\frac{1}{2}}$  is given by (4.34). We look into the null distribution of Z and assess the standard normal approximation when  $\tau$  increases.

We generated 10,000 realizations of HPPs based on (4.19), (4.20), and (4.21) where  $\alpha_A = \alpha_B = \alpha_C = 0.1$ , and  $\beta_A = 0$ ,  $\beta_B = \beta_C = 0.693$  so that  $\exp\{0.693\} =$ 2. We consider  $\Delta_B$ ,  $\Delta_C = 1, 3, 7$  and 14 time units. Normal quantile-quantile (Q-Q) plots of 10,000 values of Z are presented with  $\Delta_B$ ,  $\Delta_C = 1$  for four different end-of-followup times  $\tau$  in Figure 4.1. The normal approximation is not adequate for small  $\tau$  values. The distribution of Z in (4.70) converges to a standard normal distribution as  $\tau$  increases, but the convergence rate is slow. Figure 4.2 shows the results when  $\Delta_B = \Delta_C = 3$ . In this case, the distribution converges faster to the standard normal approximation. When  $\Delta_B = \Delta_C = 7$  in Figure 4.3, the standard normal approximation is quite accurate even for small  $\tau$  values, except for the tails when  $\tau = 100$ . Figure 4.4 shows similar results to those of Figure 4.3.

We let  $Q_p$  be the *p*th quantile of the standard normal distribution and  $Q_p$  be the empirical *p*th quantile of *Z* computed from 10,000 samples. We obtain the estimates of  $Pr(Z > Q_p) = 1 - p$ , where p = 0.950, 0.975 and 0.990 in Table 4.1. The results are in line with those obtained from Figures 4.1–4.4. Therefore, except the case in which repair times of primary components are too small, the standard normal approximations are adequate when approximately 20 or more failures are observed in the redundant component over the followup period.

The power of the statistic (4.70) was investigated by Monte Carlo simulation methods. We used the 0.95 quantile of the standard normal distribution as well as the empirical 0.95 quantile of the test statistic obtained from 10,000 simulations runs under the null hypothesis. We generated 1,000 processes under the models

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \,\alpha_A \,\exp\{\beta_A X_{AB}(t) + \beta_A X_{AC}(t)\},$$
  

$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \,\alpha_B \,\exp\{\beta_B X_{BC}(t)\},$$
  

$$\lambda_C(t|\mathcal{H}(t)) = Y_C(t) \,\alpha_C \,\exp\{\beta_C X_{CB}(t)\},$$
  
(4.71)

where we took  $\alpha_A = \alpha_B = \alpha_C = 0.1$ . We took (i)  $\beta_A = \beta_B = \beta_C = 0.693$  and (ii)  $\beta_A = \beta_B = \beta_C = 1.386$ , so that  $e^{0.693} = 2$  and  $e^{1.386} = 4$ . The power results are provided in Table 4.2, where entries are the proportions of the values of Z in 1,000 samples which are larger than the quantile values. Table 4.2 shows poor results when  $\Delta_B$  and  $\Delta_C$  are small. This is not surprising because the expected number of followup failures for  $\Delta_B$  and  $\Delta_C$  times are small, especially when  $\exp\{\beta_A\} = 2$ . However, power of the test is increasing as  $\tau$  and/or  $\Delta_B$  and  $\Delta_C$  increase. Also, power increases as  $e^{\beta_A}$  increases.



Figure 4.1: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 0$ ,  $\Delta_B = 1$ ,  $\Delta_C = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 4.2: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 0$ ,  $\Delta_B = 3$ ,  $\Delta_C = 3$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 4.3: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 0$ ,  $\Delta_B = 7$ ,  $\Delta_C = 7$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 4.4: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 0$ ,  $\Delta_B = 14$ ,  $\Delta_C = 14$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 

$\Delta_B$ and $\Delta_C$	au	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	100	1.814	2.248	2.854	0.0630	0.0406	0.0229
	200	1.794	2.193	2.67	0.0640	0.0381	0.0197
	500	1.722	2.070	2.517	0.0560	0.0315	0.0143
	1000	1.686	2.046	2.433	0.0539	0.0295	0.0136
3	100	1.731	2.102	2.55	0.0585	0.0348	0.0146
	200	1.718	2.095	2.488	0.0587	0.0318	0.0150
	500	1.693	2.048	2.385	0.0550	0.0301	0.0116
	1000	1.643	1.986	2.398	0.0499	0.0263	0.0119
7	100	1.662	1.997	2.371	0.0518	0.0274	0.0114
	200	1.659	1.993	2.361	0.0515	0.0273	0.0106
	500	1.615	1.936	2.336	0.0472	0.0237	0.0103
	1000	1.652	1.968	2.386	0.0508	0.0252	0.0110
14	100	1.607	1.859	2.178	0.0454	0.0195	0.0070
	200	1.638	1.906	2.261	0.0490	0.0224	0.0085
	500	1.627	1.947	2.287	0.0484	0.0244	0.0092
	1000	1.657	1.944	2.325	0.0515	0.0235	0.0100

Table 4.1:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when  $m = 1, \Delta_A = 0, \Delta_B > 0, \Delta_C > 0$ .  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta_B$ and $\Delta_C$	au	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 4$	$e^{\beta_A} = 4$
3	100	0.185	0.174	0.286	0.263
	200	0.227	0.216	0.396	0.383
	500	0.324	0.313	0.575	0.575
	1000	0.433	0.433	0.726	0.733
7	100	0.510	0.506	0.664	0.653
	200	0.678	0.675	0.857	0.852
	500	0.921	0.924	0.993	0.993
	1000	0.995	0.995	1.000	1.000
14	100	0.586	0.592	0.656	0.655
	200	0.778	0.779	0.870	0.873
	500	0.982	0.982	0.997	0.997
	1000	1.000	1.000	1.000	1.000

Table 4.2: Power of Z :  $m = 1, \Delta_A = 0, \Delta_B > 0, \Delta_C > 0$ 

We also considered the test statistic (4.46) developed in Section 4.2.3, where m = 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ . We generated 10,000 realizations of HPPs based on (4.35), (4.36), and (4.37) with  $\alpha_A = \alpha_B = \alpha_C = 0.1$ , and  $\beta_A = 0$ ,  $\beta_B = \beta_C = 0.693$ . We considered  $\Delta_B$ ,  $\Delta_C = 1, 3, 7$  and 14. For  $\Delta_B$ ,  $\Delta_C = 3$ , the results are presented in Figure 4.6, which shows that the normal approximations are quite accurate when  $\tau = 500$  and 1000. For  $\Delta_B$ ,  $\Delta_C = 5$  in Figure 4.7, the normal approximations are adequate when  $\tau = 500$  and 1000. For  $\Delta_B$ ,  $\Delta_C = 14$  (See Figure 4.8), the approximations are good when  $\tau = 200,500$  and 1000. Table 3.3 shows estimated  $Q_p$ and  $\hat{Pr}(Z > Q_p)$  values where p = 0.950, 0.975 and 0.990. The results in this table indicate also that the standard normal approximation is adequate for large  $\tau$ .

The power of the statistic (4.46) against the alternative hypothesis  $H_A$ :  $\beta_A \neq 0$ ,  $\alpha_A > 0$  is investigated by using Monte Carlo simulation methods. The 0.95 quantile of the standard normal distribution and the empirical 0.95 quantile of the test statistic obtained from 10,000 simulations runs were used to obtain the power. We generated 1,000 processes under the models

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \,\alpha_A \,\exp\{\beta_A X_{AB}(t) + \beta_A X_{AC}(t)\},$$
  

$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \,\alpha_B \,\exp\{\beta_B X_{BC}(t)\},$$
  

$$\lambda_C(t|\mathcal{H}(t)) = Y_C(t) \,\alpha_C \,\exp\{\beta_C X_{CB}(t)\},$$
  
(4.72)

where we took  $\alpha_A = \alpha_B = \alpha_C = 0.1$ . We took (i)  $\beta_A = \beta_B = \beta_C = 0.916$  and (ii)  $\beta_A = \beta_B = \beta_C = 1.098$ , so that  $e^{0.916} = 2.5$  and  $e^{1.098} = 3$ . The power results are provided in Table 4.4. Each entry in Table 4.4 is the proportions of the values of Z in 1,000 samples which are larger than the quantile values and this shows that the power of the test is high overall, and power increases as  $\tau$  increases.



Figure 4.5: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 1$ ,  $\Delta_B = 1$ ,  $\Delta_C = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 4.6: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 1$ ,  $\Delta_B = 3$ ,  $\Delta_C = 3$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 4.7: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 1$ ,  $\Delta_B = 7$ ,  $\Delta_C = 7$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 4.8: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 1$ ,  $\Delta_B = 14$ ,  $\Delta_C = 14$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 

$\Delta_B$ and $\Delta_C$	au	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	100	1.879	2.280	2.839	0.0695	0.0426	0.0227
	200	1.830	2.246	2.725	0.0652	0.0391	0.0215
	500	1.755	2.128	2.609	0.0611	0.0327	0.0171
	1000	1.716	2.122	2.516	0.0565	0.0343	0.0163
3	100	1.767	2.133	2.553	0.0625	0.0355	0.0171
	200	1.714	2.080	2.511	0.0578	0.032	0.0162
	500	1.677	2.037	2.422	0.0537	0.0287	0.0122
	1000	1.694	1.995	2.387	0.0557	0.0272	0.0116
7	100	1.667	2.004	2.405	0.0537	0.028	0.013
	200	1.658	1.974	2.359	0.0511	0.0264	0.0111
	500	1.646	1.990	2.341	0.0501	0.0267	0.0104
	1000	1.605	1.924	2.294	0.0451	0.0231	0.0092
14	100	1.587	1.906	2.259	0.0449	0.0222	0.0084
	200	1.614	1.906	2.294	0.0474	0.0219	0.0092
	500	1.637	1.951	2.258	0.0492	0.0242	0.0081
	1000	1.649	1.990	2.353	0.0506	0.0271	0.0115

Table 4.3:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m = 1 and  $\Delta_A = 1$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ .  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta_B$ and $\Delta_C$	au	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 2.5$	$e^{\beta_A} = 2.5$	$e^{\beta_A} = 3$	$e^{\beta_A} = 3$
3	100	0.116	0.099	0.108	0.092
	200	0.104	0.098	0.112	0.104
	500	0.104	0.100	0.120	0.116
	1000	0.100	0.094	0.113	0.108
7	100	0.243	0.239	0.278	0.277
	200	0.203	0.202	0.315	0.312
	500	0.249	0.249	0.389	0.389
	1000	0.315	0.326	0.453	0.464
14	100	0.400	0.409	0.509	0.512
	200	0.478	0.484	0.575	0.581
	500	0.579	0.579	0.788	0.788
	1000	0.735	0.735	0.929	0.929

Table 4.4: Power of Z :  $m = 1, \Delta_A = 1, \Delta_B > 0, \Delta_C > 0$ 

We now considered the score statistic (4.69) given in Section 4.2.4, where m > 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ . We fixed  $\tau$  at 100, and generated 10,000 realizations of m HPPs based on (4.50), (4.52), and (4.54) with  $\alpha_A = \alpha_B = \alpha_C = 0.1$ , and  $\beta_A = 0$ ,  $\beta_B = \beta_C = 0.693$ . We considered  $\Delta_B$ ,  $\Delta_C = 1, 3, 7$  and 14, and m = 10, 20, 50 and 100. Normal Q-Q plots (Figures 4.9, 4.10, 4.11 and 4.12) show that normal approximations are quite suitable with these various combinations with  $\Delta_B$ ,  $\Delta_C$  and m. Table 4.5 presents the value of estimated  $Q_p$  and  $\hat{Pr}(Z > Q_p)$  where p = 0.950, 0.975 and 0.990. This shows that when  $\Delta_B$ ,  $\Delta_C = 1, 3, 7$  and 14, the normal approximation is suitable.

The power of the statistic (4.69) under the alternative hypothesis  $H_A$ :  $\beta_A \neq 0$ ,  $\alpha_A > 0$  is investigated by using Monte Carlo simulation study. To obtain the power, 0.95 quantile of the standard normal distribution and the empirical 0.95 quantile of the test statistic obtained from 10,000 simulations runs with different m,  $\Delta_B$  and  $\Delta_C$ used. The power results are provided in Table 4.6, which shows that the power of the test is high for long  $\tau$  and/or m values. Also, power increases as  $e^{\beta_A}$  increases.



Figure 4.9: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 1$ ,  $\Delta_B = 1$ ,  $\Delta_C = 1$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 4.10: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 1$ ,  $\Delta_B = 3$ ,  $\Delta_C = 3$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 4.11: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 1$ ,  $\Delta_B = 7$ ,  $\Delta_C = 7$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 4.12: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_A = 1$ ,  $\Delta_B = 14$ ,  $\Delta_C = 14$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100

$\Delta_B$ and $\Delta_C$	m	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	10	1.717	2.071	2.540	0.0579	0.0307	0.0151
	20	1.714	2.041	2.458	0.0573	0.0306	0.0132
	50	1.662	1.987	2.392	0.0514	0.0264	0.0122
	100	1.674	1.999	2.381	0.0526	0.0271	0.0112
3	10	1.702	2.032	2.406	0.0570	0.0288	0.0121
	20	1.703	1.979	2.366	0.0551	0.0268	0.0112
	50	1.689	2.011	2.398	0.0548	0.0274	0.0114
	100	1.674	1.984	2.350	0.0531	0.0265	0.0109
7	10	1.636	1.952	2.304	0.0493	0.0246	0.0096
	20	1.654	1.983	2.376	0.0508	0.0263	0.0112
	50	1.637	1.928	2.308	0.0485	0.0237	0.0097
	100	1.664	1.984	2.329	0.0517	0.0263	0.0101
14	10	1.570	1.845	2.220	0.0408	0.0195	0.0076
	20	1.638	1.950	2.318	0.0495	0.0245	0.0097
	50	1.649	1.938	2.349	0.0509	0.0240	0.0103
	100	1.596	1.880	2.261	0.0456	0.0209	0.0083

Table 4.5:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m > 1.  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta_B$ and $\Delta_C$	m	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 2.5$	$e^{\beta_A} = 2.5$	$e^{\beta_A} = 3$	$e^{\beta_A} = 3$
3	10	0.120	0.113	0.149	0.137
	20	0.110	0.103	0.162	0.149
	50	0.154	0.143	0.181	0.176
_	100	0.207	0.204	0.277	0.270
7	10	0.324	0.324	0.444	0.445
	20	0.483	0.480	0.614	0.611
	50	0.725	0.726	0.824	0.826
	100	0.901	0.896	0.960	0.958
14	10	0.646	0.662	0.772	0.781
	20	0.823	0.823	0.907	0.908
	50	0.984	0.984	0.996	0.996
_	100	1.000	1.000	1.000	1.000

Table 4.6: Power of Z : m > 1,  $\Delta_A = 1$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ 

### 4.4 Application

We give an illustration of the methods considered in Section 4.2.4 where we discuss testing for parallel carryover effects when m > 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$  and  $\Delta_C > 0$  case. In Section 1.1.1, we introduced an overview of a data set of failures of diesel operated power plants.

The data set received from the power company has information on locations of communities, the number of generators in each community, maximum capacity and connected capacity of the generators in each community, engine hours effective of each generator on December 31, 2009, engine hours effective of each generator on December 31, 2011. Other information includes model, purchase year and operating speed of each generator. We also received an availability data set, which includes failure data only for a few communities for about two years. This availability data set includes times when generators are out of service as well as times when generators return back to service.

Since there is very limited information on the failure times, we generated a failure data set based on the available information received from the company, which includes limited failure data, capacities of generators and purchase year of generators. In the information sheet received from the company, there are 22 communities where each community has 3, 4 or 5 generators working together, but most of the communities have 3 generators to provide the electricity to the community. Therefore, we first assume that all communities have 3 generators working in parallel. We assume also that the redundant component (Component A) is identical for all communities. However, components in a system maybe different from system to system.

We analyze the data for 16 communities. These are the communities with 3 power generators. The purchase dates of each generator are provided in the original data. Therefore, we used the purchase dates as the start of the observation times of each generator. The observation period of each generator ends on September 1, 2015. In each community, the redundant component is the oldest generator. Thus, the start time of the observation (i.e. t = 0) is the purchase date of the redundant component in a system.

Repair times are identical within communities; that is,  $\Delta_{Ai} = \Delta_{Bi} = \Delta_{Ci} = d_i$ , where  $d_i > 0$  is a constant and i = 1, 2, ..., 16. However, the repair times between communities can be different, because repair times also include travelling time from the headquarters of the company and the distances from the headquarters are diverse. Therefore, we let repair times be 15, 30 or 45 days according to their distances from the headquarters. The average, median and variance of the number of failures of the redundant generators in the communities during the observation periods is 30.6, 29 and 105.8, respectively, and ranging from 13 to 56. A part of the data generated is given in Appendix A.1. It is worthy noting that we did not consider time trends in the data generation.



Figure 4.13: Dot plots of failures of generators in communities



Figure 4.14: Cumulative failures of community 7



Figure 4.15: Cumulative failures of community 11



Figure 4.16: Cumulative failures of community 12



Figure 4.17: Cumulative failures of community 14

Figure 4.13 shows the dot plots of the generated failure times of power generators for 8 communities. There are some clusters of failure events togeher in the redundant generators (called Machine A in the plots) soon after failures of other generators. Figures 4.14, 4.15, 4.16 and 4.17 present the plots of cumulative failures of generators (Machine A, Machine B and Machine C) with respect to the calendar time in the communities 7, 11, 12 and 14, respectively. There are some clusters of failures in the redundant generator (Machine A) noted in the plots.

Clustering of failures in Figures 4.13–4.17 suggests the presence of parallel carryover effects in redundant components. Therefore, we consider the following model for the redundant generators: For, i = 1, ..., 16,

$$\lambda_{Ai}(t|\mathcal{H}_i(t)) = Y_{Ai}(t) \,\alpha_A \,\exp\{\beta_A X_{ABi}(t) + \beta_A X_{ACi}(t)\}, \quad t > 0, \tag{4.73}$$

where  $Y_{Ai}(t)$  is the at-risk indicator function of Component A (the redundant generator) in the *i*th system,  $\alpha_A > 0$  is a baseline rate function,  $\beta_A$  is a regression parameter, and

$$X_{ABi}(t) = I\{N_{Bi}(t^{-}) > 0\}I\{t - t_{BiN_{Bi}(t^{-})} \le \Delta_{Bi}\},\$$
  
$$X_{ACi}(t) = I\{N_{Ci}(t^{-}) > 0\}I\{t - t_{CiN_{Ci}(t^{-})} \le \Delta_{Ci}\}.$$

We test the presence of parallel carryover effects in the redundant components by testing  $H_0$ :  $\beta_A = 0$ ,  $\alpha_A > 0$ . Table 4.7 presents the maximum likelihood estimates  $\hat{\alpha}_A$ and  $\hat{\beta}_A$  of  $\alpha_A$  and  $\beta_A$ , respectively, and the restricted maximum likelihood estimate  $\tilde{\alpha}_A$  of  $\alpha_A$ , when  $\beta_A = 0$ , as well as their standard errors within parentheses. The value of the test statistic Z is given in Table 4.8 which is 1.915; the standard normal approximation gives a two-sided *p*-value of 0.0554 and 1000 simulation run gives a two-sided *p*-value of 0.071. These results suggest that there is some evidence for the presence of parallel carryover effects in the redundant components.

$\hat{lpha}_A$	$\hat{eta}_A$	$ ilde{lpha}_A$
$0.004716 \ (0.000217)$	0.645239(0.138141)	$0.005008 \ (0.000228)$

Table 4.7: Estimates of  $\alpha_A$ ,  $\beta_A$  and  $\alpha_A$  when  $\beta_A = 0$ . The numbers in parentheses are the standard errors

$U_{eta_A}( ilde{oldsymbol{ heta}})$	$\widehat{Var}(U_{\beta_A}(\tilde{\boldsymbol{\theta}}))$	Z	$\ell(\hat{oldsymbol{ heta}})$	$\ell( ilde{oldsymbol{ heta}})$
13.15811	47.20691	1.915096	3019.458	3028.644

Table 4.8: The test statistic Z,  $\ell(\hat{\theta})$  and  $\ell(\tilde{\theta})$ 

It should be noted that, even though we did not generate the data from a trend model, the convex shapes of plots of cumulative failures of generators against calendar time given in Figures 4.14–4.17 suggest the presence of monotonic trends. For example, see the plot of Machine A in Figure 3.16. We discuss modeling trends with parallel carryover effects in the next chapter. However, to investigate this issue with the current data, we now consider the model (4.73) with a trend term as follows.

$$\lambda_{Ai}(t|\mathcal{H}_i(t)) = Y_{Ai}(t) \,\alpha_A \,\exp\{\beta_A X_{ABi}(t) + \beta_A X_{ACi}(t) + \gamma \,t\},\tag{4.74}$$

where i = 1, 2, ..., 16. We test the absence of trend by testing  $H_0$ :  $\gamma = 0$  with the pooled data. Table 4.9 shows the maximum likelihood estimates of  $\alpha_A$ ,  $\beta_A$  and  $\gamma$  in model (4.74), and their standard errors. A Wald-type statistic  $W = \hat{\gamma}^2/se^2(\hat{\gamma})$  gives a value of 3.536449 in Table 4.9; The *p*-value is 0.060033. This result suggests that there is a mild increasing trend in the data.

$\hat{lpha}_A$	$\hat{eta}_A$	$\hat{\gamma}$
$0.015488 \ (0.001331)$	$0.693139\ (0.101230)$	$0.000035 \ (0.0000101)$
$W = \hat{\gamma}^2 / se^2(\hat{\gamma})$	<i>p</i> -value	
3.536449	0.060033	

Table 4.9: Estimates of  $\alpha_A$ ,  $\beta_A$ ,  $\gamma$  and Wald type statistic W and p-value. The numbers in parentheses are the standard errors

## Chapter 5

# Redundant Systems with Trends and Covariates

In this chapter, we consider the tests for parallel carryover effects in redundant systems with trends. This chapter is organized as follows. In Section 5.1, we briefly introduce the concept of trend in recurrent event processes. In Section 5.2, we provide tests for the presence of parallel carryover effects when monotonic trends due to stochastic aging are present. We present the results of simulation studies in Section 4.2. We develop a score test for the presence of parallel carryover effects in models with external covariates and trends in Section 5.3. In Section 5.4, we illustrate the methods by analyzing a generated data set.

### 5.1 Introduction

Stochastic aging is an important concept in the analysis of repairable systems. If there is no effect of the age of a system on probabilistic characteristics of event occurrences, we say that there is no stochastic aging. On the contrary, if probabilistic characteristics of event occurrences of a system depend on the age of a system, we say that there is stochastic aging (Lai and Xie, 2006). Many repairable systems are subject to stochastic aging. In this chapter, we discuss the assessment of parallel carryover effects in repairable systems subject to stochastic aging.

In recurrent event processes, stochastic aging is usually studied within the context of time trends (Cox and Lewis, 1966; Cook and Lawless, 2007). As discussed by Lawless et al. (2012), the definition of a trend in recurrent event processes is elusive. Trends can appear in processes because of various reasons including stochastic aging due to wear-out phenomenon, previous number of events in a process or some external factors. There are monotonically increasing or decreasing trends as well as nonmonotonic trends such as seasonal or bathtub type. Trends due to stochastic aging are usually in a monotonically increasing nature because repairable systems are more prone to fail as they age (Thompson, 1988). In this case, a trend can be defined as a systematic change in the rate function of a recurrent event process (Lawless et al., 2012). Non-homogenous Poisson processes (NHPPs) are useful to model increasing time trends. We discussed NHPPs in Section 2.1.

In the remainder of this chapter, we consider models for parallel carryover effects and monotonic trends together, and develop tests for the presence of parallel carryover effects in various settings. We study the asymptotic properties of the tests developed. We also extend our models to include external covariates, and develop tests for parallel carryover effects when external covariates are present.

### 5.2 Models and Tests for Parallel Carryover Effects with Trends

In this section, we discuss models and tests for parallel carryover effects with monotonic trends in a single system and multiple systems. We first introduce the notation.

Suppose that there is a bivariate counting processes  $\{N_A(t), N_B(t); t \ge 0\}$ , where  $\{N_A(t); t \ge 0\}$  is a counting process for Component A and  $\{N_B(t); t \ge 0\}$  is a counting process for Component B in a system with two components working in parallel. We let  $t_{A1}, t_{A2}, \ldots$ , where  $0 < t_{A1} < t_{A2} < \ldots$ , and  $t_{B1}, t_{B2}, \ldots$ , where  $0 < t_{B1} < t_{B2} < \ldots$  denote the failure times of Component A and Component B, respectively. The components A and B are subject to repairs, and repair times cannot be ignored. Let  $\Delta_A$  and  $\Delta_B$  denote the repair times of Component A and Component B, respectively. For  $K = A, B, Y_K(t)$  is the at-risk indicator of process  $\{N_k(t); t > 0\}$ , which takes value of 1 when Component K is up and the process  $\{N_k(t); t > 0\}$  is under observation; otherwise, it equals to 0.

A model including parallel carryover effects for Component A with monotonic trend due to stochastic aging is given by

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t)\,\rho_A(t)\,\exp\{\beta_A X_A(t)\}, \quad t > 0, \tag{5.1}$$

where  $Y_A(t)$  is the at-risk indicator of Component A,  $\beta_A$  is a parameter and  $\rho_A(t) > 0$  is a time dependent baseline rate function. The rate function  $\rho_A(t)$  is given by  $\rho_A(t) = \alpha_A \exp(\gamma_A t)$ , where  $\gamma_A \in \mathbb{R}$  and t > 0. Then the intensity function (5.1) can be written as

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \,\alpha_A \,\exp\{\beta_A X_A(t) + \gamma_A t\}, \quad t > 0, \tag{5.2}$$

and

$$X_A(t) = I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \le \Delta_B\},$$
(5.3)

where function  $X_A(t)$  takes the value of 1, if Component B is in the down state at time t; otherwise it is 0. The intensity function (5.2) jumps from  $\alpha_A \exp\{\gamma_A t\}$  to  $\alpha_A \exp\{\beta_A + \gamma_A t\}$  when Component B fails, and stays for  $\Delta_B$  time units; otherwise, it remains  $\alpha_A \exp\{\gamma_A t\}$  if Component B is in the up states.

Similarly, the intensity function of the counting process  $\{N_B(t); t > 0\}$  is given by

$$\lambda_A(t|\mathcal{H}(t)) = Y_B(t) \,\alpha_B \,\exp\{\beta_B X_B(t) + \gamma_B t\}, \quad t > 0, \tag{5.4}$$

and

$$X_B(t) = I\{N_A(t^-) > 0\}I\{t - t_{AN_A(t^-)} \le \Delta_A\},$$
(5.5)

We let m denote the number of systems. In the remaining part of this section, we develop partial score tests for the presence of parallel carryover effects in five different cases; (i) a single system with monotonic trend is under observation and repair times of Components A and B are negligible (m = 1,  $\Delta_A = 0$ ,  $\Delta_B > 0$ ), (ii) multiple systems with monotonic trend are under observation and repair times of Component A are negligible (m > 1,  $\Delta_A = 0$ ,  $\Delta_B > 0$ ), (iii) a single system with monotonic trend is under observation and repair times of Component A are negligible (m > 1,  $\Delta_A = 0$ ,  $\Delta_B > 0$ ), (iii) a single system with monotonic trend is under observation and repair times of Component A are not negligible (m = 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ), (iv) multiple systems with monotonic trend are under observation and repair times of Component A are not negligible (m > 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ), (v) a single 3-component A are not negligible (m = 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ), (v) a single 3-component A are not negligible (m = 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ). The asymptotic distribution of the test statistic developed in this chapter are discussed through simulations in Section 5.3.

### **5.2.1** Case 1: $m = 1, \Delta_A = 0, \Delta_B > 0$

In this section, we consider a redundant system with two components; Components A and B. We first discuss a model with a single system and repair times of Component A are negligible. Which means that m = 1 and  $\Delta_A = 0$ . Since we consider monotonic trend in the model, we use NHPPs for this purpose. Under these assumptions, the intensity function of Component A is given by

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \,\alpha_A \, \exp(\beta_A \, X_A(t) + \gamma_A t), \quad t > 0, \tag{5.6}$$

where  $Y_A(t)$  is at-risk indicator of Component A,  $\alpha_A > 0$  is a baseline rate function and  $\beta_A$  and  $\gamma_A$  are parameters and  $X_A(t) = I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \leq \Delta_B\}$ . Since  $\Delta_A = 0$  and  $\Delta_B > 0$ , the intensity function of Component B is given by

$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \,\alpha_B \,\exp(\gamma_A t) \quad t > 0, \tag{5.7}$$

where  $Y_B(t)$  is at-risk indicator of Component B,  $\alpha_B > 0$  is a baseline rate function,  $\gamma_B$  is a parameter and  $\mathcal{H}(t) = \{N_A(u), N_B(u), Y_B(s); 0 \le u < t, 0 \le s \le t\}$ . A test for the presence of a parallel carryover effect in Component A can be developed by considering the following hypothesis.

$$H_0: \beta_A = 0, \alpha_A > 0, \gamma_A \in \mathbb{R}, \quad \text{vs.} \quad H_a: \beta_A \neq 0, \alpha_A > 0, \gamma_A \in \mathbb{R}, \tag{5.8}$$

where  $\alpha_A, \gamma_A$  are nuisance parameters and  $\beta_A$  is the parameter of interest.

We suppose that a system is under observation over the followup period  $[0, \tau]$ , where  $\tau$  is a fixed end-of-followup time. Note that, we can safely drop the at-risk indicator  $Y_A(t)$  from the model (5.6), because Component A is continuously under observation in  $[0, \tau]$  and its repair times are negligible. Let  $n_A$  denote the number of failures of Component A in  $[0, \tau]$ . The likelihood function of the outcome " $N_A(\tau) = n_A$ failures of Component A at times  $t_{A1} \leq t_{A2} \leq \cdots \leq t_{An_A}$  in  $[0, \tau]$ " is given as follows.

$$L(\boldsymbol{\theta}) = \prod_{j=1}^{n_A} \alpha_A \, e^{\beta_A \, X_A(t_{Aj}) + \gamma_A t_{Aj}} \, \exp\{-\int_0^\tau \alpha_A \, e^{\beta_A \, X_A(s) + \gamma_A s} \, ds\}, \tag{5.9}$$

where  $\boldsymbol{\theta} = (\alpha_A, \gamma_A, \beta_A)$ . The log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given as follows.

$$\ell(\boldsymbol{\theta}) = n_A \log \alpha_A + \sum_{j=1}^{n_A} \beta_A X_A(t_{Aj}) + \sum_{j=1}^{n_A} \gamma_A t_{Aj} - \int_0^\tau \alpha_A \, e^{\beta_A X_A(s) + \gamma_A s} \, ds.$$
(5.10)

Let  $\tilde{\alpha}_A$ ,  $\tilde{\gamma}_A$  denote the restricted maximum likelihood estimator of  $\alpha_A$ ,  $\gamma_A$ , respectively under the null hypothesis in (5.8). They can be obtained by maximizing  $l(\boldsymbol{\theta}_0)$  where  $\boldsymbol{\theta}_0 = (\alpha_A, \gamma_A, 0)$ . This can be done with an optimizing software such as the nlm package in R (R core team, 2013). The score vector is then defined by  $\boldsymbol{U}(\boldsymbol{\theta}) = (U_{\alpha_A}(\boldsymbol{\theta}), U_{\gamma_A}(\boldsymbol{\theta}), U_{\beta_A}(\boldsymbol{\theta}))'$  with components

$$U_{\alpha_A}(\boldsymbol{\theta}) = \frac{n_A}{\alpha_A} - \int_0^\tau e^{\beta_A X_A(s) + \gamma_A s} \, ds, \qquad (5.11)$$

$$U_{\gamma_A}(\boldsymbol{\theta}) = \sum_{j=1}^{n_A} t_{Aj} - \alpha_A \int_0^\tau s \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \qquad (5.12)$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{j=1}^{n_A} X_A(t_{Aj}) - \alpha_A \int_0^\tau X_A(s) \, e^{\beta_A X_A(s) + \gamma_A s} \, ds.$$
(5.13)

Then the observed information matrix  $I(\theta)$  is given by

$$\boldsymbol{I}(\boldsymbol{\theta}) = \begin{pmatrix} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) & I_{\alpha_A \gamma_A}(\boldsymbol{\theta}) & I_{\alpha_A \beta_A}(\boldsymbol{\theta}) \\ I_{\gamma_A \alpha_A}(\boldsymbol{\theta}) & I_{\gamma_A \gamma_A}(\boldsymbol{\theta}) & I_{\gamma_A \beta_A}(\boldsymbol{\theta}) \\ I_{\beta_A \alpha_A}(\boldsymbol{\theta}) & I_{\beta_A \gamma_A}(\boldsymbol{\theta}) & I_{\beta_A \beta_A}(\boldsymbol{\theta}) \end{pmatrix},$$
(5.14)

where its components are given as follows.

$$\begin{split} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) &= \frac{n_A}{\alpha_A^2}, \\ I_{\alpha_A \gamma_A}(\boldsymbol{\theta}) &= I_{\gamma_A \alpha_A}(\boldsymbol{\theta}) = \int_0^\tau s \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \\ I_{\alpha_A \beta_A}(\boldsymbol{\theta}) &= I_{\beta_A \alpha_A}(\boldsymbol{\theta}) = \int_0^\tau X_A(s) \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \\ I_{\gamma_A \gamma_A}(\boldsymbol{\theta}) &= \alpha_A \int_0^\tau s^2 \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \\ I_{\gamma_A \beta_A}(\boldsymbol{\theta}) &= I_{\beta_A \gamma_A}(\boldsymbol{\theta}) = \alpha_A \int_0^\tau s \, X_A(s) \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \\ I_{\beta_A \beta_A}(\boldsymbol{\theta}) &= \alpha_A \int_0^\tau \{X_A(s)\}^2 \, e^{\beta_A X_A(s) + \gamma_A s} \, ds. \end{split}$$

Let  $\tilde{\boldsymbol{\theta}}_0 = (\tilde{\alpha}_A, \tilde{\gamma}_A, 0)$ . Then, the standardized partial score statistic for testing the presence of parallel carryover effects is given by

$$Z = \frac{U_{\beta_A}(\hat{\boldsymbol{\theta}}_0)}{\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_0)^{\frac{1}{2}})},\tag{5.15}$$

where  $U_{\beta_A}(\tilde{\boldsymbol{\theta}}_0)$  is given by

$$U_{\beta_A}(\tilde{\alpha}_A, \tilde{\gamma}_A, 0) = \sum_{j=1}^{n_A} X_A(t_{Aj}) - \tilde{\alpha}_A \int_0^\tau X_A(s) e^{\tilde{\gamma}_A s} ds, \qquad (5.16)$$

and variance estimate  $\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_{0}))$  is

$$I_{\beta_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) - \left(I_{\beta_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \quad I_{\beta_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0})\right) \begin{pmatrix} I_{\alpha_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\alpha_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\gamma_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}^{-1} \begin{pmatrix} I_{\alpha_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}.$$
(5.17)

### **5.2.2** Case 2: m > 1, $\Delta_A = 0$ , $\Delta_B > 0$

We now consider multiple systems with monotonic trend. We assume that repair times of Component A are negligible while the repair times of Component B are not negligible; That is, m = 1 and,  $\Delta_A = 0$  and  $\Delta_B > 0$ . We assume that there are m independent systems under observation each with two components; Components
A and B. We let  $\{N_{Ai}(t), N_{Bi}(t); t \geq 0\}$ , i = 1, 2, ..., m, where, for K = A, B,  $\{N_{Ki}(t); t \geq 0\}$  is a counting process for Component K in the *i*th system. We let  $t_{Ki1}, t_{Ki2}, ...,$  where  $0 < t_{Ki1} < t_{Ki2} < ...$ , denote the failure times of Component K (K = A, B) in the *i*th system. The components are subject to repairs, and repair times cannot be ignored. Let  $\Delta_{Ai}$  and  $\Delta_{Bi}$  denote the repair times of Components A and B in the *i*th system, respectively. For  $K = A, B, Y_{Ki}(t)$  is the at-risk indicator of the process  $\{N_{Ki}(t); t > 0\}$ . History of the *i*th counting process  $\{N_{Ai}(t), N_{Bi}(t); t \geq 0\}$ , i = 1, 2, ..., m, is denoted by  $\mathcal{H}_i(t) = \{N_{Ai}(u), N_{Bi}(u); 0 \leq u < t\}$ .

A model including parallel carryover effects for Component A in the ith system with monotonic trends is given by

$$\lambda_{Ai}(t|\mathcal{H}_i(t)) = Y_{Ai}(t)\,\rho_A(t)\,\exp\{\beta_A X_{Ai}(t)\}, \quad t > 0, \tag{5.18}$$

where  $\beta_A$  is a regression parameter and  $\rho_A(t) > 0$  is a time dependent baseline rate function. The rate function  $\rho_A(t)$  is given by  $\rho_A(t) = \alpha_A \exp(\gamma_A t)$ . The intensity function (5.18) can be rewritten as

$$\lambda_{Ai}(t|\mathcal{H}_i(t)) = Y_{Ai}(t) \,\alpha_A \,\exp\{\beta_A X_{Ai}(t) + \gamma_A t\}, \quad t > 0, \tag{5.19}$$

where  $\gamma_A$  is a real valued parameter and

$$X_{Ai}(t) = I\{N_{Bi}(t^{-}) > 0\}I\{t - t_{BiN_{Bi}(t^{-})} \le \Delta_B\}.$$
(5.20)

Since  $\Delta_{Ai} = 0$  and  $\Delta_{Bi} > 0$ , the probabilistic characteristics of failure occurrences of Component B are not affected by failures of Component A. Therefore, the intensity function of Component B in *i*th system is then given by

$$\lambda_{Bi}(t|\mathcal{H}_i(t)) = Y_{Bi}(t)\,\alpha_B\,\exp\{\gamma_B t\}, \quad t > 0,\tag{5.21}$$

where  $\gamma_B$  is a real valued parameter, and

$$X_{Bi}(t) = I\{N_{Ai}(t^{-}) > 0\}I\{t - t_{AiN_{Ai}(t^{-})} \le \Delta_A\}.$$
(5.22)

A test of parallel carryover effects in Component A in this case can be developed by considering the hypothesis given in (5.8).

We suppose that m independent systems are under observation over  $[0, \tau_i]$ , where  $\tau_i$ 

$$L(\boldsymbol{\theta}) = \prod_{i=1}^{m} L_i(\boldsymbol{\theta}), \qquad (5.23)$$

where

$$L_{i}(\boldsymbol{\theta}) = \prod_{j=1}^{n_{Ai}} \alpha_{A} e^{\beta_{A} X_{Ai}(t_{Aij}) + \gamma_{A} t_{Aij}} \exp\{-\int_{0}^{\tau_{i}} \alpha_{A} e^{\beta_{A} X_{Ai}(s) + \gamma_{A} s} ds\},$$
(5.24)

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A, \gamma)$ . The log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$\log \alpha_A \sum_{i=1}^m n_{Ai} + \beta_A \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) + \gamma_A \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} t_{Aij} - \alpha_A \sum_{i=1}^m \int_0^{\tau_i} e^{\beta_A X_{Ai}(s) + \gamma_A s} \, ds.$$
(5.25)

The components of score vector  $\boldsymbol{U}(\boldsymbol{\theta}) = (U_{\alpha_A}(\boldsymbol{\theta}), U_{\gamma_A}(\boldsymbol{\theta}), U_{\beta_A}(\boldsymbol{\theta}))'$  are followed by

$$U_{\alpha_A}(\boldsymbol{\theta}) = \frac{\sum_{i=1}^m n_{Ai}}{\alpha_A} - \sum_{i=1}^m \int_0^{\tau_i} e^{\beta_A X_{Ai}(s) + \gamma_A s} \, ds, \qquad (5.26)$$

$$U_{\gamma_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} t_{Aij} - \alpha_A \sum_{i=1}^m \int_0^{\tau_i} s \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \qquad (5.27)$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \alpha_A \sum_{i=1}^m \int_0^{\tau_i} X_{Ai}(s) e^{\beta_A X_{Ai}(s) + \gamma_A s} \, ds.$$
(5.28)

Then the elements of the observed information matrix  $I(\theta)$  are given by

$$\begin{split} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) &= \frac{\sum_{i=1}^m n_{Ai}}{\alpha_A^2}, \\ I_{\alpha_A \gamma_A}(\boldsymbol{\theta}) &= I_{\gamma_A \alpha_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \int_0^{\tau_i} s \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \\ I_{\alpha_A \beta_A}(\boldsymbol{\theta}) &= I_{\beta_A \alpha_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \int_0^{\tau_i} X_{Ai}(s) \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \\ I_{\gamma_A \gamma_A}(\boldsymbol{\theta}) &= \alpha_A \sum_{i=1}^m \int_0^{\tau_i} s^2 \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \\ I_{\gamma_A \beta_A}(\boldsymbol{\theta}) &= I_{\beta_A \gamma_A}(\boldsymbol{\theta}) = \alpha_A \sum_{i=1}^m \int_0^{\tau_i} s \, X_{Ai}(s) \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds \\ I_{\beta_A \beta_A}(\boldsymbol{\theta}) &= \alpha_A \sum_{i=1}^m \int_0^{\tau_i} X_{Ai}(s) \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds. \end{split}$$

We let  $\tilde{\boldsymbol{\theta}}_{\mathbf{0}} = (\tilde{\alpha}_A, \tilde{\gamma}_A, 0)$ . When  $\beta_A = 0$ , the restricted maximum likelihood estimators  $\tilde{\alpha}_A$  and  $\tilde{\gamma}_A$  can be obtained by optimizing software packages. The standardized partial score statistic for testing the presence of parallel carryover effects is then given by

$$Z = \frac{U_{\beta_A}(\boldsymbol{\theta}_0)}{\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_0))^{\frac{1}{2}}},\tag{5.29}$$

where the partial score function is given by

$$U_{\beta_A}(\tilde{\theta}_0) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \tilde{\alpha}_A \sum_{i=1}^m \int_0^{\tau_i} X_{Ai}(s) \, e^{\tilde{\gamma}_A s} \, ds, \tag{5.30}$$

and variance estimate  $\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_{0}))$  is given by

$$I_{\beta_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) - \left(I_{\beta_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \quad I_{\beta_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0})\right) \begin{pmatrix} I_{\alpha_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\alpha_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\gamma_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}^{-1} \begin{pmatrix} I_{\alpha_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}.$$
(5.31)

## **5.2.3** Case 3: $m = 1, \Delta_A > 0, \Delta_B > 0$

In this section, we consider a single system with two components, in which repair times of Component A are not ignorable, but failures of Component A do not affect the probabilistic characteristics of the failure occurrences of Component B. In this case, the intensity functions of Components A, and B are given by

Component A: 
$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \alpha_A \exp(\beta_A X_A(t) + \gamma_A t), \quad t > 0,$$
 (5.32)

where  $Y_A(t)$  is the at-risk indicator of Component A,  $\alpha_A > 0$  is a baseline rate function,  $\beta_A$  and  $\gamma_A$  are parameters and

Component B: 
$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \alpha_B \exp(\gamma_A t), \quad t > 0,$$
 (5.33)

where  $Y_B(t)$  is the at-risk indicator,  $\alpha_B > 0$  is a baseline rate function and  $\gamma_A$  is a parameter and  $\mathcal{H}(t) = \{N_A(u), N_B(u), Y_B(s); 0 \le u < t, 0 \le s \le t\}$ . Then the likelihood function of the outcome " $N_A(\tau) = n_A$  failures of Component A at times  $t_{A1} \le t_{A2} \le \cdots \le t_{An_A}$  in  $[0, \tau]$ " is given as follows.

$$L(\boldsymbol{\theta}) = \prod_{j=1}^{n_A} \alpha_A \, e^{\beta_A \, X_A(t_{Aj}) + \gamma_A t_{Aj}} \, \exp\{-\int_0^\tau Y_A(s) \, \alpha_A \, e^{\beta_A \, X_A(s) + \gamma_A s} \, ds\}, \tag{5.34}$$

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A, \gamma)$ . The log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given as follows.

$$\ell(\boldsymbol{\theta}) = n_A \log \alpha_A + \sum_{j=1}^{n_A} \beta_A X_A(t_{Aj}) + \sum_{j=1}^{n_A} \gamma_A t_{Aj} - \int_0^\tau Y_A(s) \,\alpha_A \, e^{\beta_A X_A(s) + \gamma_A s} \, ds.$$
(5.35)

Let  $\tilde{\alpha}_A$  and  $\tilde{\gamma}_A$  denote the restricted maximum likelihood estimator of  $\alpha_A$  and  $\gamma_A$  when  $\beta_A = 0$ .  $\tilde{\alpha}_A$  and  $\tilde{\gamma}_A$  can be obtained by maximizing  $l(\boldsymbol{\theta}_0)$  where  $\boldsymbol{\theta}_0 = (\alpha_A, \gamma_A, 0)$ . The score vector is then defined by  $\boldsymbol{U}(\boldsymbol{\theta}) = (U_{\alpha_A}(\boldsymbol{\theta}), U_{\gamma_A}(\boldsymbol{\theta}), U_{\beta_A}(\boldsymbol{\theta}))'$  with components

$$U_{\alpha_A}(\boldsymbol{\theta}) = \frac{n_A}{\alpha_A} - \int_0^{\tau} Y_A(s) \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \qquad (5.36)$$

$$U_{\gamma_A}(\boldsymbol{\theta}) = \sum_{j=1}^{n_A} t_{Aj} - \alpha_A \int_0^\tau Y_A(s) \, s \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \qquad (5.37)$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{j=1}^{n_A} X_A(t_{Aj}) - \alpha_A \int_0^\tau Y_A(s) X_A(s) e^{\beta_A X_A(s) + \gamma_A s} ds.$$
(5.38)

The observed information matrix  $I(\theta)$  is given by

$$\boldsymbol{I}(\boldsymbol{\theta}) = \begin{pmatrix} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) & I_{\alpha_A \gamma_A}(\boldsymbol{\theta}) & I_{\alpha_A \beta_A}(\boldsymbol{\theta}) \\ I_{\gamma_A \alpha_A}(\boldsymbol{\theta}) & I_{\gamma_A \gamma_A}(\boldsymbol{\theta}) & I_{\gamma_A \beta_A}(\boldsymbol{\theta}) \\ I_{\beta_A \alpha_A}(\boldsymbol{\theta}) & I_{\beta_A \gamma_A}(\boldsymbol{\theta}) & I_{\beta_A \beta_A}(\boldsymbol{\theta}) \end{pmatrix},$$
(5.39)

where

$$\begin{split} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) &= \frac{n_A}{\alpha_A^2}, \\ I_{\alpha_A \gamma_A}(\boldsymbol{\theta}) &= I_{\gamma_A \alpha_A}(\boldsymbol{\theta}) = \int_0^\tau Y_A(s) \, s \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \\ I_{\alpha_A \beta_A}(\boldsymbol{\theta}) &= I_{\beta_A \alpha_A}(\boldsymbol{\theta}) = \int_0^\tau Y_A(s) \, X_A(s) \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \\ I_{\gamma_A \gamma_A}(\boldsymbol{\theta}) &= \alpha_A \int_0^\tau Y_A(s) \, s^2 \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \\ I_{\gamma_A \beta_A}(\boldsymbol{\theta}) &= I_{\beta_A \gamma_A}(\boldsymbol{\theta}) = \alpha_A \int_0^\tau Y_A(s) \, s \, X_A(s) \, e^{\beta_A X_A(s) + \gamma_A s} \, ds, \\ I_{\beta_A \beta_A}(\boldsymbol{\theta}) &= \alpha_A \int_0^\tau Y_A(s) \, \{X_A(s)\}^2 \, e^{\beta_A X_A(s) + \gamma_A s} \, ds. \end{split}$$

Let  $\tilde{\boldsymbol{\theta}}_0 = (\tilde{\alpha}_A, \tilde{\gamma}_A, 0)$ . The standardized partial score statistic for testing the presence of parallel carryover effects is given by

$$Z = \frac{U_{\beta_A}(\tilde{\boldsymbol{\theta}}_0)}{\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_0)^{\frac{1}{2}})},\tag{5.40}$$

where  $U_{\beta_A}(\tilde{\boldsymbol{\theta}}_0)$  is given by

$$U_{\beta_A}(\tilde{\alpha}_A, \tilde{\gamma}_A, 0) = \sum_{j=1}^{n_A} X_A(t_{Aj}) - \tilde{\alpha}_A \int_0^\tau Y_A(s) \, X_A(s) \, e^{\tilde{\gamma}_A s} \, ds, \tag{5.41}$$

and variance estimate  $\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_{0}))$  is

$$I_{\beta_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) - \left(I_{\beta_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \quad I_{\beta_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0})\right) \begin{pmatrix} I_{\alpha_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\alpha_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\gamma_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}^{-1} \begin{pmatrix} I_{\alpha_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}.$$
(5.42)

#### **5.2.4** Case 4: m > 1, $\Delta_A > 0$ , $\Delta_B > 0$

We now consider multiple systems with two components; Components A and B. Repair times of Component A are not negligible. We assume that the failures of redundant component (Component A) in the *i*th system, where i = 1, 2, ..., m. Failures of Component A do not affect the probabilistic characteristics of failure occurrences of Component B while failures of Component B affect the probabilistic characteristics of failure occurrences of Component A.

In this case, the intensity functions of Components A and B in the ith system can be defined by

Component A: 
$$\lambda_{Ai}(t|\mathcal{H}_i(t)) = Y_{Ai}(t) \alpha_A \exp\{\beta_A X_{Ai}(t)\gamma_A t\}, \quad t > 0, \quad (5.43)$$

and

Component B: 
$$\lambda_{Bi}(t|\mathcal{H}_i(t)) = Y_{Bi}(t) \alpha_B \exp\{\gamma_A t\}, \quad t > 0,$$
 (5.44)

where  $X_{Ai}(t) = I\{N_{Bi}(t^-) > 0\}I\{t - t_{BiN_{Bi}(t^-)} \le \Delta_B\}$ ,  $Y_{Ai}(t)$  is the at-risk indicator of Component A in the *i*th system,  $\beta_A$  and  $\gamma_A$  are parameters, and  $\mathcal{H}_i(t) = \{N_{Ai}(u), N_{Bi}(u), Y_{Ai}(s), Y_{Bi}(s); 0 \le u < t, 0 \le s \le t\}$ .

The likelihood function of the outcome " $N_{Ai}(\tau_i) = n_{Ai}$  failures of Component A at times  $t_{Ai1} \leq t_{Ai2} \leq \cdots \leq t_{Ain_{Ai}}$  in  $[0, \tau_i]$ " can be written as

$$L(\boldsymbol{\theta}) = \prod_{i=1}^{m} L_i(\boldsymbol{\theta}), \qquad (5.45)$$

where

$$L_{i}(\boldsymbol{\theta}) = \prod_{j=1}^{n_{Ai}} \alpha_{A} e^{\beta_{A} X_{Ai}(t_{Aij}) + \gamma_{A} t_{Aij}} \exp\{-\int_{0}^{\tau_{i}} Y_{Ai}(s) \alpha_{A} e^{\beta_{A} X_{Ai}(s) + \gamma_{A} s} ds\}, \quad (5.46)$$

where  $\boldsymbol{\theta} = (\alpha_A, \beta_A, \gamma)$ . The log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$\log \alpha_{A} \sum_{i=1}^{m} n_{Ai} + \beta_{A} \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) + \gamma_{A} \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} t_{Aij} - \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) e^{\beta_{A} X_{Ai}(s) + \gamma_{A} s} ds.$$
(5.47)

Then, the components of score vector  $\boldsymbol{U}(\boldsymbol{\theta}) = (U_{\alpha_A}(\boldsymbol{\theta}), U_{\gamma_A}(\boldsymbol{\theta}), U_{\beta_A}(\boldsymbol{\theta}))'$  are followed by

$$U_{\alpha_A}(\boldsymbol{\theta}) = \frac{\sum_{i=1}^m n_{Ai}}{\alpha_A} - \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \qquad (5.48)$$

$$U_{\gamma_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} t_{Aij} - \alpha_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) s \, e^{\beta_A X_{Ai}(s) + \gamma_A s} \, ds, \tag{5.49}$$

and

$$U_{\beta_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \alpha_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) X_{Ai}(s) e^{\beta_A X_{Ai}(s) + \gamma_A s} ds.$$
(5.50)

The elements of observed information matrix  $\boldsymbol{I}(\boldsymbol{\theta})$  are given by

$$\begin{split} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) &= \frac{\sum_{i=1}^m n_{Ai}}{\alpha_A^2}, \\ I_{\alpha_A \gamma_A}(\boldsymbol{\theta}) &= I_{\gamma_A \alpha_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, s \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \\ I_{\alpha_A \beta_A}(\boldsymbol{\theta}) &= I_{\beta_A \alpha_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \int_0^{\tau_i} X_{Ai}(s) \, Y_{Ai}(s) \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \\ I_{\gamma_A \gamma_A}(\boldsymbol{\theta}) &= \alpha_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, s^2 \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \\ I_{\gamma_A \beta_A}(\boldsymbol{\theta}) &= I_{\beta_A \gamma_A}(\boldsymbol{\theta}) = \alpha_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, s \, X_{Ai}(s) \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds, \\ I_{\beta_A \beta_A}(\boldsymbol{\theta}) &= \alpha_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, X_{Ai}(s) \, e^{\beta_A \, X_{Ai}(s) + \gamma_A s} \, ds. \end{split}$$

Let  $\tilde{\boldsymbol{\theta}}_0 = (\tilde{\alpha}_A, \tilde{\gamma}_A, 0)$ .  $\tilde{\alpha}_A$  are  $\tilde{\gamma}_A$  restricted maximum likelihood estimators of  $\alpha_A$  and  $\gamma_A$ , respectively, when  $\beta_A = 0$ . Good optimizing software packages such as nlm in R, give those restricted maximum likelihood estimators without analytical derivations.

Then the standardized partial score statistic for testing parallel carryover effects is given by

$$Z = \frac{U_{\beta_A}(\tilde{\boldsymbol{\theta}}_0)}{\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_0))^{\frac{1}{2}}},\tag{5.51}$$

where the partial score function is given by

$$U_{\beta_A}(\tilde{\theta}_0) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} X_{Ai}(t_{Aij}) - \tilde{\alpha}_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) X_{Ai}(s) e^{\tilde{\gamma}_A s} ds, \qquad (5.52)$$

and variance estimate  $\widehat{Var}(U_{\beta}(\tilde{\theta}_0))$  is

$$I_{\beta_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) - \left(I_{\beta_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \quad I_{\beta_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0})\right) \begin{pmatrix} I_{\alpha_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\alpha_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\gamma_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}^{-1} \begin{pmatrix} I_{\alpha_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}.$$
(5.53)

## **5.2.5** Case 5: $m = 1, \Delta_A > 0, \Delta_B > 0, \Delta_C > 0$

In this section, we consider a single system with 3 components; Components A, B and C. We consider the case where repair times of Components A, B and C are not negligible. That is, m = 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$  and  $\Delta_C > 0$ . However, one of the components in the system, that is Component A, is a redundant component. Therefore, failures of Component A do not affect the probabilistic characteristics of failure occurrences of Components B and C. Failures of Component B affect the probabilistic characteristics of component C affect the probabilistic characteristics of Component C affect the probabilistic characteristics of Component C affect the probabilistic characteristics of Component B.

We now give extensions of notations given in previous sections. Suppose that there is a multivariate counting processes  $\{N_A(t), N_B(t), N_C(t); t \ge 0\}$ , where  $\{N_A(t); t \ge 0\}$ is a counting process for Component A,  $\{N_B(t); t \ge 0\}$  is a counting process for Component B and  $\{N_C(t); t \ge 0\}$  is a counting process for Component C in a system with three components working in parallel. We let  $t_{A1}, t_{A2}, \ldots$ , where  $0 < t_{A1} < t_{A2} < \ldots$ ,  $t_{B1}, t_{B2}, \ldots$ , where  $0 < t_{B1} < t_{B2} < \ldots$ , and  $t_{C1}, t_{C2}, \ldots$ , where  $0 < t_{C1} < t_{C2} < \ldots$ , denote the failure times of Components A, B and C, respectively. The components are subject to repairs and repair times cannot be ignored. Let  $\Delta_A$ ,  $\Delta_B$  and  $\Delta_C$  denote the repair times of Components A, B and C, respectively. For K = A, B and  $C, Y_K(t)$  is the at-risk indicator of process  $\{N_K(t); t > 0\}$ .

A model including parallel carryover effects for Component A is given by

$$\lambda_A(t|\mathcal{H}(t)) = Y_A(t)\,\alpha_A\,\exp\{\beta_A X_{AB}(t) + \beta_A X_{AC}(t) + \gamma_A t\}, \quad t > 0, \tag{5.54}$$

where  $\alpha_A > 0$  is a baseline rate function,  $\beta_A$  and  $\gamma_A$  are parameters,  $X_{AB}(t) =$ 

 $I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \le \Delta_B\}, \text{ and } X_{AC}(t) = I\{N_C(t^-) > 0\}I\{t - t_{CN_C(t^-)} \le \Delta_C\}.$ 

When Component A is in the up state at time t, (i) the intensity function (5.54) becomes  $\alpha_A \exp\{2\beta_A + \gamma_A t\}$  when both Component B and Component C fails, (ii) $\alpha_A \exp\{\beta_A + \gamma_A t\}$  when one of Components B or C fails,  $(iii) \alpha_A \exp\{\gamma_A t\}$  if none of Components B and C fails. This is a similar situation that we explained in Section 3.2.

A model for parallel carryover effects in this case can be also defined for Component B. In this case, the intensity function of Component B is given by

$$\lambda_B(t|\mathcal{H}(t)) = Y_B(t) \,\alpha_B \,\exp\{\beta_B \,X_{BC}(t) + \gamma_B t\}, \quad t > 0, \tag{5.55}$$

where  $\alpha_B > 0$  is a baseline rate function,  $\gamma_B$  is a parameter, and  $X_{BC}(t) = I\{N_C(t^-) > 0\}I\{t - t_{CN_C(t^-)} \leq \Delta_C\}.$ 

Similarly, the intensity function of Component C is given by

$$\lambda_C(t|\mathcal{H}(t)) = Y_C(t) \,\alpha_C \,\exp(\beta_C \,X_{CB}(t) + \gamma_C t), \quad t > 0, \tag{5.56}$$

where  $\alpha_C > 0$  is a baseline rate function,  $\gamma_C$  is a parameter,  $X_{CB}(t) = I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \leq \Delta_B\}.$ 

A test for the presence of a parallel carryover effect in Component A can be developed by considering the following composite hypothesis given in (5.8).

We suppose that a single system is under observation over the followup period  $[0, \tau]$ , where  $\tau$  is fixed end-of-followup time. Let  $n_A$ , where  $n_A \ge 0$ , denote the number of failures of Component A over  $[0, \tau]$  and  $t_{A1}, t_{A1}, \ldots, t_{An_A}$  be the failure times of Component A.

Let  $\boldsymbol{\theta} = (\alpha_A, \gamma_A, \beta_A)$ . The likelihood function of the outcome " $N_A(\tau) = n_A$  failures of Component A at times  $t_{A1}, t_{A2}, \ldots, t_{An_A}$  in  $[0, \tau]$ " is given by

$$L(\boldsymbol{\theta}) = \prod_{j=1}^{n_A} \alpha_A \, e^{\beta_{AB} \, (X_{AB}(t_{Aj}) + X_{AC}(t_{Aj})) + \gamma_A t_{Aj}} \, \exp\{-\int_0^\tau Y_A(s) \, \alpha_A \, e^{\beta_A \, (X_{AB}(s) + X_{AC}(s)) + \gamma_A s} \, ds\},$$
(5.57)

The log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$\ell(\boldsymbol{\theta}) = n_A \log \alpha_A + \beta_A \sum_{j=1}^{n_A} (X_{AB}(t_{Aj}) + X_{AC}(t_{Aj})) + \sum_{j=1}^{n_A} \gamma_A t_{Aj}$$
(5.58)  
$$- \alpha_A \int_0^\tau Y_A(s) \, e^{\beta_A (X_{AB}(s) + X_{AC}(s)) + \gamma_A s} \, ds.$$

Let  $\tilde{\boldsymbol{\theta}}_0 = (\tilde{\alpha}_A, \tilde{\gamma}_A, 0)$  where  $\tilde{\alpha}_A$  and  $\tilde{\gamma}_A$  are restricted maximum likelihood estimators of  $\alpha_A$  and  $\gamma_A$ , respectively, when  $\beta_A = 0$ .

Then, we obtain the following standardized partial score test statistic for testing parallel carryover effects

$$Z = \frac{U_{\beta_A}(\boldsymbol{\theta}_0)}{\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_0))^{\frac{1}{2}}},\tag{5.60}$$

where the partial score function  $U_{\beta_A}(\tilde{\boldsymbol{\theta}}_0)$  is given by

$$U_{\beta_A}(\tilde{\theta}_0) = \sum_{j=1}^{n_A} (X_{AB}(t_{Aj}) + X_{AC}(t_{Aj})) - \tilde{\alpha}_A \int_0^\tau Y_A(s) \left( X_{AB}(s) + X_{AC}(s) \right) e^{\tilde{\gamma}_A s} \, ds.$$
(5.61)

Variance estimate  $\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_0))$  is followed by

$$I_{\beta_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) - \left(I_{\beta_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \quad I_{\beta_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0})\right) \begin{pmatrix} I_{\alpha_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\alpha_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\gamma_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}^{-1} \begin{pmatrix} I_{\alpha_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\beta_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix},$$
(5.62)

where

$$\begin{split} I_{\alpha_A\alpha_A}(\tilde{\boldsymbol{\theta}}_0) &= \frac{n_A}{\tilde{\alpha}_A^2}, \\ I_{\alpha_A\gamma_A}(\tilde{\boldsymbol{\theta}}_0) &= I_{\gamma_A\alpha_A}(\tilde{\boldsymbol{\theta}}_0) = \int_0^\tau Y_A(s) \, s \, e^{\tilde{\gamma}_A s} \, ds, \\ I_{\alpha_A\beta_A}(\tilde{\boldsymbol{\theta}}_0) &= I_{\beta_A\alpha_A}(\tilde{\boldsymbol{\theta}}_0) = \int_0^\tau Y_A(s) \, (X_{AB}(s) + X_{AC}(s)) \, e^{\tilde{\gamma}_A s} \, ds, \\ I_{\gamma_A\gamma_A}(\tilde{\boldsymbol{\theta}}_0) &= \tilde{\alpha}_A \int_0^\tau Y_A(s) \, s^2 \, e^{\tilde{\gamma}_A s} \, ds, \\ I_{\gamma_A\beta_A}(\tilde{\boldsymbol{\theta}}_0) &= I_{\beta_A\gamma_A}(\tilde{\boldsymbol{\theta}}_0) = \tilde{\alpha}_A \int_0^\tau Y_A(s) \, s \, (X_{AB}(s) + X_{AC}(s)) \, e^{\tilde{\gamma}_A s} \, ds \\ I_{\beta_A\beta_A}(\tilde{\boldsymbol{\theta}}_0) &= \tilde{\alpha}_A \int_0^\tau Y_A(s) \, (X_{AB}(s) + X_{AC}(s))^2 \, e^{\tilde{\gamma}_A s} \, ds. \end{split}$$

### 5.3 Simulation Studies

In this section, we present the results of simulation studies conducted to assess when asymptotic normal approximation for test statistics developed in Section 5.2.3, Section 5.2.4 and Section 5.2.5 are satisfactory where

(i) 
$$m = 1, \Delta_A > 0, \Delta_B > 0$$
  
(ii)  $m > 1, \Delta_A > 0, \Delta_B > 0$ ,  
(iii)  $m = 1, \Delta_A > 0, \Delta_B > 0, \Delta_C > 0$ .

We consider two settings where m = 1 and  $\tau \to \infty$ , and  $m \to \infty$  and  $\tau$  is fixed in each of the 3 cases given above.

We first consider testing for presence of parallel carryover effects in a single system with monotonic trend. In this case, m = 1,  $\Delta_A > 0$  and  $\Delta_B > 0$  so we consider the models (5.32) and (5.33). The hypothesis of no parallel carryover effects in Component A is  $H_0$ :  $\beta_A = 0$ ,  $\alpha_A > 0$ ,  $\gamma_A \in \mathbb{R}$  and this is conducted by using the statistic Z in (5.40). We generated 10,000 realizations of NHPPs under the null hypothesis with fixed values of the parameters for all scenarios where  $\alpha_A = \alpha_B = 0.1$ ,  $\beta_A = 0$ ,  $\beta_B = 0.693$ , and  $\gamma_A = \gamma_B = 0.001$ . We considered  $\Delta_A = 1$  and  $\Delta_B = 1,3,7$ , and 14. We fixed  $\Delta_A$  at 1. Normal quantile-quantile (Q-Q) plots of 10,000 values of Z are given in Figures 5.1 when  $\Delta_A = 1$  for  $\tau = 100,200,500$  and 1000. The standard normal approximations are not accurate in those cases when  $\tau = 100,200$  and 500. However, it is noted that as  $\tau$  increases, the standard normal approximation improves. Figure 5.2 shows the results when  $\Delta = 3$ . In this scenario, the standard normal approximations are accurate when  $\tau = 500$  and 1000. Figures 5.3 and 5.4 present the results when  $\Delta = 7$  and  $\Delta = 14$ , respectively. In those scenarios, the standard normal approximations are accurate, as  $\tau$  increases.

Table 5.1 presents estimates of  $Q_p$  and  $Pr(Z > Q_p)$  where p = 0.950, 0.975 and 0.990. This also indicates that the standard normal approximation is not adequate for small and moderate  $\tau$  values and small  $\Delta_B$  values, but the approximation becomes accurate as  $\tau$  increases.

The power of the statistic (5.40) against the alternative hypothesis  $H_A : \beta_A \neq 0$ is investigated by Monte Carlo simulation methods. We use the 0.95 quantile of the standard normal distribution and the empirical 0.95 quantile of the test statistic obtained from 10,000 simulations runs conducted under the null hypothesis with different  $\tau$ ,  $\Delta_B$  and  $\Delta_C$  values as discussed above. We generated 1,000 processes under the alternative model where we took  $\alpha_A = \alpha_B = 0.1$  and  $\beta_A = \beta_B = 0.693$ . The power results are presented in Table 4.2 where entries are the proportions of the values of Z in 1,000 samples which are larger than the quantile values. Table 5.2 shows that the power of the test is high overall, and power increases as  $\tau$  increases.



Figure 5.1: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 1$ ,  $\Delta_A = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 5.2: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 3$ ,  $\Delta_A = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 5.3: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 7$ ,  $\Delta_A = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 5.4: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = 14$ ,  $\Delta_A = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 

$\Delta_B$	au	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	100	1.901	2.517	3.191	0.0781	0.0527	0.0324
	200	1.874	2.314	2.845	0.0699	0.0447	0.0244
	500	1.780	2.172	2.614	0.0629	0.0365	0.019
	1000	1.693	2.011	2.459	0.0546	0.0278	0.014
3	100	1.870	2.267	2.729	0.0710	0.0436	0.0226
	200	1.736	2.091	2.502	0.0590	0.0342	0.0158
	500	1.685	2.030	2.375	0.0537	0.0275	0.0117
	1000	1.696	2.046	2.448	0.0557	0.0299	0.0133
7	100	1.759	2.058	2.476	0.0622	0.0321	0.0152
	200	1.692	2.029	2.434	0.0538	0.0299	0.0127
	500	1.639	1.947	2.32	0.0488	0.0244	0.0100
	1000	1.656	1.974	2.321	0.0513	0.0257	0.0099
14	100	1.687	1.988	2.318	0.0551	0.0268	0.0097
	200	1.669	1.944	2.278	0.0528	0.0239	0.0089
	500	1.619	1.890	2.281	0.0483	0.0205	0.0085
	1000	1.593	1.942	2.348	0.0460	0.0241	0.0103

Table 5.1:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m=1.  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta_B$	au	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 2.5$	$e^{\beta_A} = 2.5$
1	100	0.284	0.219	0.390	0.309
	200	0.398	0.330	0.574	0.512
	500	0.724	0.691	0.927	0.910
	1000	0.989	0.988	1.000	1.000
3	100	0.319	0.258	0.506	0.432
	200	0.509	0.483	0.723	0.694
	500	0.876	0.871	0.988	0.986
	1000	0.998	0.998	1.000	1.000
7	100	0.369	0.338	0.516	0.476
	200	0.580	0.568	0.787	0.771
	500	0.918	0.918	0.996	0.996
	1000	0.999	0.999	1.000	1.000
14	100	0.310	0.294	0.461	0.443
	200	0.528	0.515	0.765	0.750
	500	0.874	0.880	0.991	0.993
	1000	0.994	0.994	1.000	1.000

Table 5.2: Power of Z :  $m = 1, \Delta_A > 0, \Delta_B > 0$ 

We now consider testing for the presence of parallel carryover effects in multiple systems with trend. This is the case in which m > 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ , and the models are given in (5.43) and (5.44). The hypothesis of no parallel carryover effects in Component A is conducted by employing the statistic Z in (5.51). We generated 10,000 realizations of m NHPPs under the null hypothesis with the parameter values  $\alpha_A = \alpha_B = 0.1$ ,  $\beta_A = 0$ ,  $\beta_B = 0.693$ , and  $\gamma_A = \gamma_B = 0.001$ . We considered  $\Delta_B = 1,3,7$  and 14 and we fixed  $\Delta_A$  at 1 and  $\tau$  at 100. Normal quantile-quantile (Q-Q) plots of 10,000 values of Z are given in Figures 5.5, 5.6, 5.7, and 5.8 with various combinations of m and  $\Delta_B$ . The standard normal approximation is very accurate in each setting with all m values. Table 5.3 presents estimates of  $Q_p$  and  $Pr(Z > Q_p)$  when p = 0.950, 0.975 and 0.990. This also indicates that the standard normal approximation is adequate as  $\tau$  increases.

The power of the statistic (5.51) against the alternative hypothesis  $H_A : \beta_A \neq 0$ is investigated by Monte Carlo simulation methods. We use the 0.95 quantile of the standard normal distribution and the empirical 0.95 quantile of the test statistic obtained from 10,000 simulations runs calculated under the null hypothesis with different m and  $\Delta_B$  values. We generated 1,000 processes under the alternative model where we took  $\alpha_A = \alpha_B = 0.1$  and  $\gamma_A = \gamma_B = 0.001$  when  $e^{\beta_A} = e^{\beta_B} = 1.5$  or 2. The power results are presented in Table 5.4 where entries are the proportions of the values of Z in 1,000 samples which are larger than the quantile values. And Table 5.4 indicates that the power of the test is high as m increases.



Figure 5.5: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100$ ,  $\Delta_B = 1$ ,  $\Delta_A = 1$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 5.6: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100, \Delta_B = 3, \Delta_A = 1$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 5.7: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100$ ,  $\Delta_B = 7$ ,  $\Delta_A = 1$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100



Figure 5.8: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\tau = 100$ ,  $\Delta_B = 14$ ,  $\Delta_A = 1$ , and (1) m = 10, (2) m = 20, (3) m = 50, (4) m = 100

$\Delta_B$	m	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
1	10	1.694	2.110	2.552	0.0543	0.0316	0.0161
	20	1.714	2.061	2.471	0.0562	0.0301	0.0136
	50	1.729	2.093	2.471	0.0581	0.0328	0.0134
	100	1.670	2.004	2.426	0.0531	0.0276	0.0121
3	10	1.707	2.026	2.457	0.0552	0.0286	0.0132
	20	1.699	2.056	2.443	0.0549	0.0305	0.0129
	50	1.684	1.962	2.368	0.0539	0.0256	0.0105
	100	1.655	1.977	2.381	0.0510	0.0263	0.0116
7	10	1.622	1.935	2.326	0.0477	0.0241	0.0101
	20	1.667	1.975	2.303	0.0522	0.0263	0.0095
	50	1.668	1.977	2.327	0.0532	0.0258	0.0101
	100	1.690	1.995	2.321	0.0546	0.0272	0.0099
14	10	1.664	1.943	2.316	0.0519	0.024	0.0098
	20	1.648	1.960	2.319	0.0504	0.0251	0.0099
	50	1.650	1.974	2.342	0.0513	0.0254	0.0108
	100	1.675	1.989	2.321	0.0535	0.0275	0.0099

Table 5.3:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m > 1,  $\Delta_A = 1$ , and  $\tau = 100$ .  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

$\Delta_B$	m	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 1.5$	$e^{\beta_A} = 1.5$	$e^{\beta_A} = 2$	$e^{\beta_A} = 2$
1	10	0.463	0.448	0.834	0.822
	20	0.686	0.660	0.978	0.978
	50	0.947	0.934	1.000	1.000
	100	0.998	0.998	1.000	1.000
3	10	0.643	0.624	0.975	0.970
	20	0.868	0.854	0.999	0.999
	50	0.996	0.996	1.000	1.000
	100	1.000	1.000	1.000	1.000
7	10	0.682	0.693	0.987	0.987
	20	0.918	0.913	1.000	1.000
	50	1.000	1.000	1.000	1.000
	100	1.000	1.000	1.000	1.000
14	10	0.678	0.673	0.987	0.985
	20	0.927	0.927	1.000	1.000
	50	0.999	0.999	1.000	1.000
	100	1.000	1.000	0.999	0.999

Table 5.4: Power of Z :  $m > 1, \Delta_A = 1, \Delta_B > 0$ 

We consider the score statistic (5.60) given in Section 5.2.5 where m = 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$ ,  $\Delta_C > 0$ . We use the models (5.54), (5.55), and (5.56) for Components A, B and C, respectively. We generated 10,000 realizations of NHPPs with  $\alpha_A = \alpha_B = \alpha_C = 0.1$ ,  $\beta_A = 0$ ,  $\beta_B = \beta_C = 0.693$ ,  $\gamma_A = \gamma_B = \gamma_C = 0.001$ . We considered  $\Delta_B$ ,  $\Delta_C = 1, 3, 7$  and 14 when  $\tau = 100, 200, 500$  and 1000.

For  $\Delta_B$ ,  $\Delta_C = 1$ , from the Q-Q plots in Figure 5.9, the normal approximations are quite accurate when  $\tau = 500$  and 1000. For  $\Delta_B$ ,  $\Delta_C = 1$  from the Q-Q plots in Figure 5.9, the normal approximation is adequate when  $\tau = 1000$ . For  $\Delta_B$ ,  $\Delta_C = 3$ from the Q-Q plots in Figure 5.10, the normal approximations are adequate when  $\tau =$ 500 and 1000. For  $\Delta_B$ ,  $\Delta_C = 7$  from the Q-Q plots in Figure 5.11, the approximations are accurate when  $\tau = 200,500$  and 1000. For  $\Delta_B$ ,  $\Delta_C = 14$  from the Q-Q plots in Figure 5.12, the approximations are quite accurate at  $\tau = 500$  and 1000.

Table 5.5 shows estimated  $Q_p$  and  $\hat{Pr}(Z > Q_p)$  values where p = 0.950, 0.975 and 0.990. Table 5.5 indicates also that the standard normal approximations are adequate for large  $\tau$  when  $\Delta_B \ \Delta_C = 1, 3$  and 7. However, the normal approximation is less accurate when  $\Delta_B$  and  $\Delta_C$  are 14 time units.

We next consider the power of the tests with size 0.05. So, we used 0.95 quantile of the standard normal distribution and 0.95 empirical quantile of the test statistic obtained from 10,000 simulations runs with various combinations of  $\tau$ ,  $\Delta_B$  and  $\Delta_C$ . We generated 1,000 processes where we took  $\alpha_A = \alpha_B = \alpha_C = 0.1$ ,  $\beta_A = \beta_B = \beta_C =$ 0.693, and  $\gamma_A = \gamma_B = \gamma_C = 0.001$ . The results of the power of the test are presented in Table 5.6 where entries are the proportions of the values of Z in 1,000 samples which are larger than the quantile values. It shows that the power of the test is high overall, and power increases as  $\tau$  increases.



Figure 5.9: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = \Delta_C = 1$ ,  $\Delta_A = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 5.10: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = \Delta_C = 3$ ,  $\Delta_A = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 5.11: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = \Delta_C = 7$ ,  $\Delta_A = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 



Figure 5.12: Normal Q-Q plots of 10,000 simulated values of the test statistic Z when  $\Delta_B = \Delta_C = 14$ ,  $\Delta_A = 1$ , and (1)  $\tau = 100$ , (2)  $\tau = 200$ , (3)  $\tau = 500$ , (4)  $\tau = 1000$ 

$\Delta_B$	au	$\hat{Q}_{0.950}$	$\hat{Q}_{0.975}$	$\hat{Q}_{0.990}$	$\hat{Pr}(Z > 1.645)$	$\hat{Pr}(Z > 1.960)$	$\hat{Pr}(Z > 2.326)$
$\Delta_C$							
1	100	1.910	2.430	3.084	0.0719	0.0468	0.0283
	200	1.823	2.197	2.603	0.0671	0.0392	0.0182
	500	1.711	2.040	2.419	0.0584	0.0296	0.0121
	1000	1.646	1.963	2.297	0.0501	0.0255	0.0093
3	100	1.910	2.270	2.919	0.0768	0.0459	0.0233
	200	1.755	2.118	2.524	0.0604	0.0355	0.0146
	500	1.737	2.067	2.518	0.0601	0.0317	0.0151
	1000	1.655	1.935	2.323	0.0512	0.0237	0.0100
7	100	1.842	2.236	3.002	0.0695	0.0413	0.0217
	200	1.663	2.017	2.408	0.0521	0.0283	0.0115
	500	1.734	2.057	2.513	0.0605	0.0323	0.0145
	1000	1.659	1.968	2.339	0.051	0.0256	0.0103
14	100	1.653	2.008	2.509	0.0507	0.0274	0.0138
	200	1.588	1.897	2.288	0.0442	0.0223	0.0094
	500	1.597	1.928	2.270	0.0446	0.0234	0.0087
	1000	1.600	1.897	2.265	0.0451	0.0212	0.0088

Table 5.5:  $\hat{Q}_p$  is the empirical *p*th quantile of *Z* computed from 10,000 samples when m=1.  $\hat{Pr}(Z > Q_p)$  is the proportion of the values of *Z* in 10,000 samples which are larger than the *p*th quantile of a standard normal distribution

A 1.A		$\hat{D}$ ( $T > 1.04$ )	$\hat{\mathcal{D}}$ $(\mathbf{Z} > \hat{\mathbf{Q}} > 1)$	$\hat{D}$ ( $T > 1.04T$ )	$\hat{\mathcal{D}}$ $(\mathbf{Z} > \hat{\mathbf{Q}} > 1)$
$\Delta_B$ and $\Delta_C$	au	$Pr\{Z > 1.645\}$	$Pr\{Z > Q_{0.950}\}$	$Pr\{Z > 1.645\}$	$Pr\{Z > Q_{0.950}\}$
		$e^{\beta_A} = 2.5$	$e^{\beta_A} = 2.5$	$e^{\beta_A} = 3$	$e^{\beta_A} = 3$
3	100	0.051	0.030	0.087	0.061
	200	0.059	0.050	0.083	0.069
	500	0.078	0.069	0.091	0.079
	1000	0.076	0.074	0.174	0.173
7	100	0.127	0.096	0.156	0.120
	200	0.160	0.156	0.234	0.230
	500	0.220	0.199	0.425	0.402
	1000	0.381	0.376	0.691	0.687
14	100	0.168	0.326	0.243	0.593
	200	0.313	0.565	0.443	0.460
	500	0.525	0.549	0.725	0.734
	1000	0.725	0.741	0.912	0.917

Table 5.6: Power of Z :  $m = 1, \Delta_A = 1$  with Stochastic Aging

# 5.4 Redundant Systems with Trends and Covariates

In this section, we consider a redundant system with trend due to stochastic aging and external covariates. We consider multiple systems with 3 components; Components A, B and C. We consider the case where repair times of Components A, B and C are not negligible. That is, m > 1,  $\Delta_A > 0$ ,  $\Delta_B > 0$  and  $\Delta_C > 0$ . However, one of the components in the system, that is Component A, is a redundant component. For K = A, B and C,  $\mathbf{z}_{Ki}(t) = (z_{1,Ki}(t), z_{2,Ki}(t), \dots, z_{p,Ki}(t))'$  be a  $p \times 1$  matrix to include possibly time-varying external covariates and fixed covariates. The intensity functions of Components A, B and C in the *i*th system,  $i = 1, \dots, m$ , are given by

$$\lambda_{Ai}(t|\mathcal{H}(t)) = Y_{Ai}(t) \,\alpha_A \,\exp\{\beta_A \left(X_{ABi}(t) + X_{ACi}(t)\right) + \gamma_A t + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(t)\}, \quad t > 0, \ (5.63)$$

$$\lambda_{Bi}(t|\mathcal{H}(t)) = Y_{Bi}(t) \,\alpha_B \,\exp\{\beta_B \,X_{BCi}(t) + \gamma_B t + \boldsymbol{\zeta}' \boldsymbol{z}_{Bi}(t)\}, \quad t > 0, \tag{5.64}$$

and

$$\lambda_{Ci}(t|\mathcal{H}(t)) = Y_{Ci}(t) \,\alpha_C \,\exp\{\beta_C \,X_{CBi}(t) + \gamma_C t + \boldsymbol{\zeta}' \boldsymbol{z}_{Ci}(t)\}, \quad t > 0, \tag{5.65}$$

respectively. Let  $\boldsymbol{\theta} = (\alpha_A, \beta_A, \gamma_A, \boldsymbol{\zeta})$ . The likelihood function  $L(\boldsymbol{\theta})$  of the outcome " $N_{Ai}(\tau_i) = n_{Ai}$  failures of Component A at times  $t_{Ai1} \leq t_{Ai2} \leq \cdots \leq t_{Ain_{Ai}}$  in  $[0, \tau_i]$ " can be written as follows.

$$\prod_{i=1}^{m} \prod_{j=1}^{n_{A}} \alpha_{A} \exp(\beta_{A}(X_{ABi}(t_{Aij}) + X_{ACi}(t_{Aij})) + \gamma_{A}t_{Aij} + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(t_{Aij})))$$

$$\exp\{-\int_{0}^{\tau_{i}} Y_{Ai}(s) \alpha_{A} e^{\beta_{A}(X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(s)} ds\}.$$
(5.66)

Then, log likelihood function  $\ell(\boldsymbol{\theta}) = \log L(\boldsymbol{\theta})$  is given by

$$\ell(\boldsymbol{\theta}) = \sum_{i=1}^{m} n_{Ai} \log \alpha_{A} + \beta_{A} \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} (X_{ABi}(t_{Aij}) + X_{ACi}(t_{Aij})) + \gamma_{A} \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} t_{Aij} + \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} \boldsymbol{\zeta}' \, \boldsymbol{z}_{Ai}(t_{Aij}) - \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, \alpha_{A} \, e^{\beta_{A} \, (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A} s + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(s)} ds.$$

The components of the score vector  $\boldsymbol{U}(\boldsymbol{\theta})$  are given by

$$U_{\alpha_{A}}(\boldsymbol{\theta}) = \frac{\sum_{i=1}^{m} n_{Ai}}{\alpha_{A}} + \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) e^{\beta_{A}(X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(s)} ds, \qquad (5.67)$$

$$U_{\gamma_A}(\boldsymbol{\theta}) = \sum_{i=1}^m \sum_{j=1}^{n_{Ai}} t_{Aij} - \alpha_A \sum_{i=1}^m \int_0^{\tau_i} Y_{Ai}(s) \, s \, e^{\beta_A \left( X_{ABi}(s) + X_{ACi}(s) \right) + \gamma_A s + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(s)} ds, \quad (5.68)$$

$$\boldsymbol{U}_{\boldsymbol{\zeta}}(\boldsymbol{\theta}) = \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} \boldsymbol{z}_{Ai}(t_{Aij}) - \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, \boldsymbol{z}_{Ai}(s) \, \boldsymbol{e}^{\beta_{A}(X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(s)} ds,$$
(5.69)

and

$$U_{\beta_{A}}(\boldsymbol{\theta}) = \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} X_{ABi}(t_{Aij}) + X_{ACi}(t_{Aij}) - \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \left( X_{ABi}(s) + X_{ACi}(s) \right) e^{\beta_{A} \left( X_{ABi}(s) + X_{ACi}(s) \right) + \gamma_{A}s + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(s)} ds.$$
(5.70)

The observed information matrix  $\boldsymbol{I}(\boldsymbol{\theta}) = \partial \ell(\boldsymbol{\theta}) / \partial \boldsymbol{\theta} \partial \boldsymbol{\theta}'$  is given by

$$\boldsymbol{I}(\boldsymbol{\theta}) = \begin{pmatrix} I_{\alpha_A \alpha_A}(\boldsymbol{\theta}) & I_{\alpha_A \gamma_A}(\boldsymbol{\theta}) & I_{\alpha_A \boldsymbol{\zeta}_A}(\boldsymbol{\theta}) & I_{\alpha_A \beta_A}(\boldsymbol{\theta}) \\ I_{\gamma_A \alpha_A}(\boldsymbol{\theta}) & I_{\gamma_A \gamma_A}(\boldsymbol{\theta}) & I_{\gamma_A \boldsymbol{\zeta}_A}(\boldsymbol{\theta}) & I_{\gamma_A \beta_A}(\boldsymbol{\theta}) \\ I_{\boldsymbol{\zeta}_A \alpha_A}(\boldsymbol{\theta}) & I_{\boldsymbol{\zeta}_A \gamma_A}(\boldsymbol{\theta}) & I_{\boldsymbol{\zeta}_A \boldsymbol{\zeta}_A}(\boldsymbol{\theta}) & I_{\boldsymbol{\zeta}_A \beta_A}(\boldsymbol{\theta}) \\ I_{\beta_A \alpha_A}(\boldsymbol{\theta}) & I_{\beta_A \gamma_A}(\boldsymbol{\theta}) & I_{\beta_A \boldsymbol{\zeta}_A}(\boldsymbol{\theta}) & I_{\beta_A \beta_A}(\boldsymbol{\theta}) \end{pmatrix},$$
(5.71)

where

$$\begin{split} I_{\alpha_{A}\alpha_{A}}(\boldsymbol{\theta}) &= \frac{\sum_{i=1}^{m} n_{A_{i}}}{\alpha_{A}^{2}}, \\ I_{\alpha_{A}\gamma_{A}}(\boldsymbol{\theta}) &= \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, s \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\alpha_{A}\gamma_{A}}(\boldsymbol{\theta}) &= \frac{\sum_{i=1}^{m} n_{Ai}}{\alpha_{A}} + \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, \boldsymbol{z}_{Ai}(s) \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\alpha_{A}\beta_{A}}(\boldsymbol{\theta}) &= \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, (X_{ABi}(s) + X_{ACi}(s)) \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\gamma_{A}\gamma_{A}}(\boldsymbol{\theta}) &= \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, s^{2} \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\gamma_{A}\gamma_{A}}(\boldsymbol{\theta}) &= \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, s \, \boldsymbol{z}_{Ai}(s) \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\gamma_{A}\beta_{A}}(\boldsymbol{\theta}) &= \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, s \, \boldsymbol{z}_{Ai}(s) \, s^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\zeta_{A}\beta_{A}}(\boldsymbol{\theta}) &= \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, \boldsymbol{z}_{Ai}(s) \, s \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\zeta_{A}\beta_{A}}(\boldsymbol{\theta}) &= \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, \boldsymbol{z}_{Ai}(s) \, s \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\zeta_{A}\beta_{A}}(\boldsymbol{\theta}) &= \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, \boldsymbol{z}_{Ai}(s) \, (X_{ABi}(s) + X_{ACi}(s)) \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\beta_{A}\beta_{A}}(\boldsymbol{\theta}) &= \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, \boldsymbol{z}_{Ai}(s) \, (X_{ABi}(s) + X_{ACi}(s)) \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds, \\ I_{\beta_{A}\beta_{A}}(\boldsymbol{\theta}) &= \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \, \{X_{ABi}(s) + X_{ACi}(s)\}^{2} \, e^{\beta_{A} (X_{ABi}(s) + X_{ACi}(s)) + \gamma_{A}s + \zeta' \boldsymbol{z}_{Ai}(s)} ds. \end{split}$$

For testing  $H_0$ :  $\beta_A = 0$ , the restricted maximum likelihood estimators of  $\boldsymbol{\theta}_0 = (\alpha_A, \gamma_A, \boldsymbol{\zeta}, 0)$  can be obtained maximizing  $\ell(\boldsymbol{\theta}_0)$ . This can be done by an optimizing software package such as nlm in R. Then, let  $\tilde{\boldsymbol{\theta}}_0 = (\tilde{\alpha}_A, \tilde{\gamma}_A, \tilde{\boldsymbol{\zeta}}, 0)$  denote the restricted maximum likelihood estimator of  $\boldsymbol{\theta}_0$ .

Then the standardized partial score statistic for testing parallel carry over effects is given by  $\Tau$ 

$$Z = \frac{U_{\beta}(\boldsymbol{\theta}_0)}{\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_0))^{\frac{1}{2}}},$$
(5.72)

$$U_{\beta_{A}}(\boldsymbol{\theta}) = \sum_{i=1}^{m} \sum_{j=1}^{n_{Ai}} X_{ABi}(t_{Aij}) + X_{ACi}(t_{Aij}) - \alpha_{A} \sum_{i=1}^{m} \int_{0}^{\tau_{i}} Y_{Ai}(s) \left( X_{ABi}(s) + X_{ACi}(s) \right) e^{\gamma_{A}s + \boldsymbol{\zeta}' \boldsymbol{z}_{Ai}(s)} ds,$$
(5.73)

and  $\widehat{Var}(U_{\beta}(\tilde{\boldsymbol{\theta}}_0))$  is given by

$$I_{\beta_A\beta_A}(\tilde{\boldsymbol{\theta}}_0) - \boldsymbol{I_1}(\tilde{\boldsymbol{\theta}}_0) \, \boldsymbol{I_2}(\tilde{\boldsymbol{\theta}}_0)^{-1} \, \boldsymbol{I_1}(\tilde{\boldsymbol{\theta}}_0)', \tag{5.74}$$

where

$$\boldsymbol{I}_{1}(\tilde{\boldsymbol{\theta}}_{0}) = \left( I_{\beta_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \quad I_{\beta_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \quad I_{\beta_{A}\boldsymbol{\zeta}_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \right),$$
(5.75)

and

$$\boldsymbol{I}_{2}(\tilde{\boldsymbol{\theta}}_{0}) = \begin{pmatrix} I_{\alpha_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\alpha_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\alpha_{A}\boldsymbol{\zeta}_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\gamma_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\gamma_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\gamma_{A}\boldsymbol{\zeta}_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \\ I_{\boldsymbol{\zeta}_{A}\alpha_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\boldsymbol{\zeta}_{A}\gamma_{A}}(\tilde{\boldsymbol{\theta}}_{0}) & I_{\boldsymbol{\zeta}_{A}\boldsymbol{\zeta}_{A}}(\tilde{\boldsymbol{\theta}}_{0}) \end{pmatrix}.$$
(5.76)

## 5.5 Application

The purpose of this section is to illustrate the methods developed in this chapter. Therefore, we use a generated data set for testing the presence of parallel carryover effects in multiple systems with three components: Components A, B and C. We consider the case given in Section 5.2.5, where the number of systems m > 0, and the repair times of Components A, B and C are not negligible; i.e.,  $\Delta_A > 0$ ,  $\Delta_B > 0$  and  $\Delta_C > 0$ . In this setup, Component A is the redundant component and Components B and C are primary components.

As explained in Sections 1.1.1 and 4.4, information and limited availability data on diesel power engines (i.e., power generators) operating in 22 remote communities were provided by a power company. Among those communities, 4 of them have relatively old generators comparing with the other generators operating in the same community; the community number 3, the community number 5, the community number 9 and the community number 15. Since ages of some generators are more than 30 years, it is useful to consider increasing time trends in the failure occurrences of these power generators. Since the availability data, including failure times and return to service times, are very limited, we generated a failure data set for the power generators operating in those 4 communities with old power generators. The values of parameters used in the data generation process were based on the limited availability data set, capacities of the generators and purchase year of the generators. In the following discussion, we use the term "system" to denote the power system operating in a community, and "component" to denote the power generators, and use them interchangeably. Each of those 4 systems includes three components.

We let the oldest generator in each of these systems be the redundant component (Component A). Other two generators (Components B and C) are then the primary generators. Starting time of the followups (i.e., t = 0) of the systems are the purchase date of the redundant component. The end-of-followup time  $\tau$  is September 1, 2015, which is the same for all 4 systems. Since the redundant components are the oldest ones, we consider monotonic time trends only in them. Repair times are based on the distances of the communities from the headquarters of the company. This means that repair times are identical for the components operating in the same system. However, repair times can vary between communities. Therefore, we chose the repair times of the generators as follows:  $\Delta_3 = 30$  days,  $\Delta_5 = 30$  days,  $\Delta_9 = 15$  days, and  $\Delta_{15} = 45$  days for the community numbers 3, 5, 9, and 15, respectively.



Figure 5.13: Dot plots of failures of the redundant generators operating in the community number 3, 5, 9 and 15



Figure 5.14: Plots of cumulative failures of the redundant generators in the community number 3, 5, 9 and 15 versus operating time

Observation periods of the redundant generators in the community number 3, 5, 9 and 15 are 9131, 12783, 9496 and 10592 days, respectively. Also, total number of failures of the redundant generators operating in the community number 3, 5, 9 and 15 over their corresponding observation period are 45, 63, 49 and 44, respectively. The observation periods of the generators B and C in the community number 3, 5, 9 and 15 are different from each others. Dot plots of the failures of the redundant generators are given in Figure 5.13. The cumulative number of failures of redundant components against the operating times are presented in Figure 5.14. The plots in these figures show that there are clustering of failures over time. Furthermore, mild convex shape observed in the plots presented in Figure 5.13 indicates an increasing trend in the failure occurrences in redundant generators.

The absence of parallel carryover effects in a system can be tested by considering the following model. For i = 3, 5, 9, and 15, the model for Component A in the *i*th system is given by

$$\lambda_{Ai}(t|\mathcal{H}(t)) = Y_{Ai}(t) \,\alpha_{Ai} \,\exp\{\beta_{Ai} X_{ABi}(t) + \beta_{Ai} X_{ACi}(t) + \gamma_i t\}, \qquad t > 0, \qquad (5.77)$$

where  $Y_{Ai}(t)$  is the at-risk indicator of Component A in the *i*th system,  $\alpha_{Ai}$ ,  $\beta_{Ai}$  and  $\gamma_i$  are model parameters, and  $X_{ABi}(t)$  and  $X_{ACi}(t)$  are defined in Section 5.2.5.

We present the restricted maximum likelihood estimates  $\tilde{\alpha}_{Ai}$  and  $\tilde{\gamma}_{Ai}$  of  $\alpha_{Ai}$  and  $\gamma_{Ai}$ , respectively, and their standard errors in Table 5.7 when  $\beta_{Ai} = 0$ . We used the nlm package in R to obtain these estimates and their standard errors. We test the null hypothesis  $H_0$ :  $\beta_{Ai} = 0$  against the alternative hypothesis  $H_1$ :  $\beta_{Ai} \neq 0$ . The observed values of the test statistic Z are given in Table 5.8 along with the *p*-values based on the standard normal distribution as well as based on 1000 simulation runs (denoted by  $p^*$ -value). The results in Table 5.8 suggest that there is some evidence against the null hypothesis  $H_0$ :  $\beta_{A_3} = 0$  and  $H_0$ :  $\beta_{A_9} = 0$ . Therefore, we conclude that there is parallel carryover effects on the redundant components operating in community 3 and community 9.

$\tilde{\alpha}_{A_3}$	$0.002501 \ (0.001232)$	$\tilde{\alpha}_{A_5}$	$0.002499 \ (0.000914)$
$\widetilde{\gamma_3}$	$0.000182 \ (0.000050)$	$\tilde{\gamma_5}$	$0.000131 \ (0.000022)$
$\tilde{\alpha}_{A_9}$	0.002504 ( 0.000823)	$\tilde{\alpha}_{A_{15}}$	0.002501(0.000840)
$\widetilde{\gamma_9}$	$0.000144 \ (0.000033)$	$\tilde{\gamma}_{15}$	$0.000119 \ (0.000030)$

Table 5.7: Estimates of  $\alpha_{Ai}$ ,  $\beta_{Ai}$ ,  $\gamma_i$  and  $\alpha_{Ai}$ ,  $\gamma_i$  when  $\beta_{Ai} = 0$  where i = 3, 5, 9, and 15. The numbers in parentheses are the standard errors of the estimates.

i	$\Delta_i$	$U_{\beta_{Ai}}(\tilde{\boldsymbol{ heta}})$	$Var(U_{\beta_{Ai}}(\tilde{\boldsymbol{\theta}}))$	Z	<i>p</i> -value	$p^*$ -value
3	30	4.729459	6.231403	1.894605	0.058	0.069
5	30	1.874224	4.122099	0.923129	0.355	0.355
9	15	3.305245	2.682509	2.018056	0.049	0.058
15	45	1.537437	4.950848	0.690967	0.489	0.507

Table 5.8: Statistic Z and p-values

Figure 5.14 suggests that there are some trends in the number of failures of the redundant generators as we generated data from a trend model. It should be noted that we took  $\gamma = 0.0001$  in the model (5.77) while generating the data. Therefore, the rate function increases very slowly as time increases. We now test of absence of monotonic trend of the systems. For this purpose, we use the Laplace test (Cox and Lewis, 1966), which is given by

$$LA = \frac{\{\sum_{j=1}^{n_i} T_{ij} - n_i(\tau_i)/2\}}{\{n_i(\tau_i)^2/12\}^{1/2}}$$
(5.78)

The observed values of the Laplace statistic (5.78) are presented in Table 5.9. The standard normal approximation gives a two-sided p-value of 0.0003258, 0.0000268, 0.0495566 and 0.1258483. These results show that there is a strong evidence of trend in the redundant components in community 3, 5 and 9.

i	LA	<i>p</i> -value
3	3.593810	0.0003258
5	4.198454	0.0000268
9	1.963771	0.0495566
15	1.530681	0.1258483

Table 5.9: Statistic LA and p-values

# Chapter 6

# Summary and Future Research

In this chapter, we present a summary of the results obtained in this thesis. We also briefly discuss some of the limitations of our approach and future research topics.

# 6.1 Summary and Conclusions

The statistical analysis of failure data from repairable systems has been a major research area in statistics and reliability engineering. In this thesis, we considered a reliability improvement technique called redundancy, which is often applied to power systems. Redundancy can significantly increase the availability of systems. However, implementation of it incurs cost. If the cost of repairs of components in a redundant system is expensive, the reliability program of a company may not be cost-efficient. Therefore, it is important to detect the reasons of failures of components in redundant systems. In some cases, failures of components may result in a temporary increase in the risk of failures in the redundant components while the failed ones are under repair. In such cases, failures may cluster together over time. We referred to this phenomenon as a parallel carryover effect. In this thesis, we developed simple tests for parallel carryover effects and discuss their asymptotic properties in various settings. The tests developed are easy to implement and have good overall power.

It is well known that the data acquisition is notoriously difficult in reliability studies (see, Lawless, 1983; Blischke and Murthy, 2003). In this thesis, we analyzed two randomly generated data sets as explained below (also, see Sections 4.4 and 5.5). The data sets were generated according to the limited information received from a power company. Our purpose with these analyses was to illustrate the methodology developed in the thesis. We did not consider issues related to the data sets. However, the results of these analyses showed that the methods developed can be applied to real life data sets as well.

In Chapter 3, we investigated testing for the presence of parallel carryover effects in redundant systems with two components. We considered the cases in which repair times of the redundant component are negligible and non-negligible. Asymptotic normal approximations of the test statistics given in Sections 3.2.1 and 3.2.2 were discussed analytically as well as by simulations under two different settings; (i) when the observation period increases in a single system, and (ii) when the number of systems approaches infinity for a fixed observation period. Simulation studies showed that the standard normal approximation is adequate in both cases. The results of a simulation study conducted to investigate the power of the tests were presented under various scenarios. We found that the overall power of the test is high in overall.

In Chapter 4, we discussed testing for parallel carryover effects in redundant systems with three components. We considered cases in which repair times of the redundant component are negligible and non-negligible. We presented two settings; (i)when a single system is under observation, and (ii) when multiple systems are under observation. We investigated asymptotic properties of the test statistics through simulations. The results of our simulation studies indicated that the standard normal approximation is accurate in settings with large m and/or  $\tau$  values. An application of the methods was given by analyzing a simulated data set in the context of power systems with multiple generators.

In Chapter 5, we discussed testing for parallel carryover effects in redundant systems with stochastic aging and covariates. The models developed in Chapters 2 and 3 were extended accordingly. We developed partial score tests for the presence of parallel carryover effects in redundant systems with two or three components. In this chapter, we assumed that the components are subject to monotonic trends due to stochastic aging. Once again, we considered two settings; (i) when a single system is under observation, and (ii) when multiple systems are under observation. The adequacy of the standard normal approximations for the test statistics was investigated through simulations under various settings. Finally, we analyzed a generated data set similar to that of the previous chapter, but included a trend in the rate functions of failure occurrences of the redundant generator.

An issue related to our methodology is the choice of repair times  $\Delta$  of a failed

component. The tests developed in this thesis require a fixed value for the repair times of a failed component. This information can be obtained from experts or history data. However, care is needed because too long or too short time specification for the repair times may result in difficulties in the estimation of parallel carryover effects. In an extreme case, when  $\Delta \to \infty$  or  $\Delta \to 0$ , parallel carryover effects are not estimable. This is not an important issue in the context of repairable systems because repair times of failed components are usually not that short or long.

Another issue is the misspecification of  $\Delta$ . We studied this issue through a simulation study. Table B.1 and Table B.2 in Appendix B show that the power of the test statistic Z in (3.21) developed in Section 3.2.1 when  $\Delta_{B_0}$  is misspecified. In Table B.1,  $\Delta_{B_0}$  denotes the true value of the repair times and  $\Delta_B$  is the value of the repair times used in the test. For example, we generated data with  $\Delta_{B_0} = \frac{2}{3}\Delta_B$ , where  $\Delta_B = 7$  time units was used in the test. The factors of the simulation study were  $\Delta_B = 7$ , 14,  $\Delta_{B_0} = \frac{2}{3}\Delta_B, \frac{4}{3}\Delta_B, e^{\beta_A} = 2$ , 4, and  $\tau = 100, 200, 500, and 1000$ . The results of Tables B.1 and B.2 indicates that there is a small loss in the power if repair times specified little smaller or larger than the true value of repair times. However, power is increasing as  $\tau$  increases. We also investigated the power of the test statistic Z in (3.38) developed in Section 3.2.2 when m = 10, 20, 50, 100 and  $\tau = 100$ . In this case, we present the results with the same factors of Tables B.1 and B.2, but  $e^{\beta_A} = 2, 2.5$ . The results are presented in Tables B.3 and B.4, and the conclusions are similar to those obtained from Tables B.1 and B.2.

In this thesis, we considered the settings where parallel carryover effects can be classified as time varying external covariates. This is because of the fact that failures of primary components temporarily change the probabilistic characteristics of failure occurrences in a redundant component, but failures of a redundant component do not affect the probabilistic characteristics of failure occurrences of primary components. As discussed in Chapter 1, in this case, carryover effects can be classified as an external covariate (Kalbfleisch and Prentice, 2002). However, if failures of a redundant component also affects the probabilistic characteristics of failure occurrences on the primary components, then joint modelling of the at-risk indicators and counting processes is needed. This is because the at risk indicator does not evolve independently of the counting process. In this case, the full likelihood based inference may become unmanageable.

### 6.2 Alternating Two-State Processes

In this thesis, we assumed for simplicity that parallel carryover effect periods are constant. In applications, the duration of repair times may vary. Alternating twostate processes are useful when the duration of repair times varies (See Cook and Lawless, 2007; Nakagawa, 2008). An alternating two-state process involves the "up" state and the "down" state, and a process can be either in the up state or the down state at time t. We can specify a counting process model, which is similar to the models developed in this thesis, for "up to down" transitions. Then, this model can be extended to include varying repair times which are generated from a model. Such a method is considered by Hong et al. (2013), where they use a truncated lognormal distribution as a model for event durations. We can develop such a method to investigate the presence of carryover effects with varying repair times. This topic will be investigated in the future.

#### 6.3 Redundant Systems with Imperfect Repairs

Imperfect repairs are common in applications. There are many imperfect repair models proposed for the repairable systems. A model that can incorporate imperfect repairs as well as effects of maintenance activities of repairable systems is proposed by Cigsar (2010). In this model, an internal carryover effect is specified to reflect imperfect repairs or maintenance activities so that it is assumed that after each repair or maintenance activities the risk of a failure temporarily changes for the same process. Our models in this thesis can be extended to include such internal carryover effects as well. For example, with the settings of Section 3.2.1, this can be done by specifying the intensity function as  $\lambda_A(t|\mathcal{H}(t)) = Y_A(t) \alpha_A \exp\{\beta_{A_1}X_A(t) + \beta_{A_2}Z_A(t)\}$ , where  $X_A(t) = I\{N_B(t^-) > 0\}I\{t - t_{BN_B(t^-)} \leq \Delta_B\}$  and  $Z_A(t) = I\{N_A(t^-) >$  $0\}I\{t - t_{AN_A(t^-)} \leq \Delta\}$ . So function  $X_A(t)$  detects parallel carryover effects and function  $Z_A(t)$  detects transient carryover effects. Here,  $\Delta_B$  is the repair times of Component B and  $\Delta$  is specified time of transient carryover of Component A. We will explore such models as a future research.
### Bibliography

- [1] AALEN, O., BORGAN, O., AND GJESSING, H. Survival and event history analysis: a process point of view. Springer Science & Business Media, 2008.
- [2] ANDERSEN, P. K. BORGAN, G. R., AND KEIDING, N. Statistical Models Based on Counting Processes. Springer-Verlag, New York, 1993.
- [3] ASCHER, H. E. Evaluation of repairable system reliability using the "bad-as-old" concept. *IEEE Transactions on Reliability* 17, 2 (1968), 103–110.
- [4] BAKER, R. D. Data-based modeling of the failure rate of repairable equipment. Lifetime Data Analysis 7, 1 (2001), 65–83.
- [5] BILLINTON, R., AND ALLAN, R. N. Reliability evaluation of engineering systems. New York: Plenum press, 1992.
- BOOS, D. D. On generalized score tests. The American Statistician 46, 4 (1992), 327–333.
- [7] BROWN, M., AND PROSCHAN, F. Imperfect repair. Journal of Applied Probability (1983), 851–859.
- [8] CIGSAR, C. Some models and tests for carryover effects and trends in recurrent event processes. Ph.D. thesis, University of Waterloo, 2010.
- [9] CIGSAR, C., AND LAWLESS, J. F. Assessing transient carryover effects in recurrent event processes, with application to chronic health conditions. *The Annals of Applied Statistics* 6, 4 (2012), 1641–1663.
- [10] COOK, R. J., AND LAWLESS, J. F. The statistical analysis of recurrent events. Springer Science & Business Media, 2007.
- [11] Cox, D. The statistical analysis of dependencies in point processes. In Symposium on Point Processes, Ed. P.A.W. Lewis. John Wiley & Sons, New York, 1972b.
- [12] COX, D., AND LEWIS, P. The Statistical Analysis of Series of Events. Monographs on Statistics and Applied Probability. Chapman & Hall, 1966.

- [13] DALEY, D., AND VERE-JONES, D. An introduction to the theory of point processes. Springer, New York, 1988.
- [14] DALEY, D., AND VERE-JONES, D. An introduction to the theory of point processes. Vol 1: Elementary Theory and Methods, Springer, New York, 2003.
- [15] KALBFLEISCH, J. D., AND PRENTICE, R. L. Relative risk (cox) regression models. The Statistical Analysis of Failure Time Data, Second Edition (2002), 95–147.
- [16] KARR, A. F. Point processes and their statistical inference, 2 ed. Marcel Dekker, New York, 1991.
- [17] KINGMAN, J. F. C. Poisson processes. John Wiley & Sons, 1993.
- [18] LAI, C.-D., AND XIE, M. Stochastic ageing and dependence for reliability. Springer Science & Business Media, 2006.
- [19] LAWLESS, J. F. Statistical methods in reliability. *Technometrics 25*, 4 (1983), 305–316.
- [20] LAWLESS, J. F., CIGSAR, C., AND COOK, R. J. Testing for monotone trend in recurrent event processes. *Technometrics* 54, 2 (2012), 147–158.
- [21] LEWIS, P. A., AND SHEDLER, G. S. Statistical analysis of non-stationary series of events in a data base system. *IBM Journal of Research and Development 20*, 5 (1976), 465–482.
- [22] LINDQVIST, B. H. On the statistical modeling and analysis of repairable systems. Statistical Science (2006), 532–551.
- [23] MEEKER, W. Q., AND ESCOBAR, L. A. Pitfalls of accelerated testing. IEEE Transactions on Reliability 47, 2 (1998), 114–118.
- [24] MISRA, K. B. Reliability analysis and prediction: A methodology oriented treatment, vol. 15. Elsevier, 2012.
- [25] NAKAGAWA, T. Advanced reliability models and maintenance policies. Springer Science & Business Media, 2008.
- [26] NAKAGAWA, T. Stochastic processes: With applications to reliability theory. Springer Science & Business Media, 2011.
- [27] PEÑA, E. A. Smooth goodness-of-fit tests for composite hypothesis in hazard based models. The Annals of Statistics 26, 5 (1998), 1935–1971.
- [28] R CORE TEAM. R: A Language and Environment for Statistical Computing. R Foundation for Statistical Computing, Vienna, Austria, 2013.

- [29] RIGDON, S. E. Imperfect repair, counting processes. *Encyclopedia of Statistics* in Quality and Reliability (2007).
- [30] RIGDON, S. E., AND BASU, A. P. Statistical methods for the reliability of repairable systems. John Wiley & Sons, New York, 2000.
- [31] THOMPSON, W. Point process models with applications to safety and reliability. New York: Chapman & Hall, 1988.

## Appendix A

### **Data Sets**

#### A.1 Data 1

Part of data set used in Section 3.4

Comp:	1=Componen	t A, $2=$	Component	B. 3	=Component	С
- ·	- · · · ·	- )	- · · · · ·	, -	- · · · · ·	

					_					
Region	City	Comp	Start	End	_	Region	City	Comp	Start	End
1	1	1	60	90	-	1	1	1	3995	4025
1	1	1	474	504		1	1	1	4076	4106
1	1	1	1092	1122		1	1	1	4342	4372
1	1	1	1296	1326		1	1	1	4680	4710
1	1	1	1445	1475		1	1	2	2187	2217
1	1	1	1770	1800		1	1	2	2350	2380
1	1	1	2137	2167		1	1	2	2807	2837
1	1	1	2224	2254		1	1	2	3589	3619
1	1	1	2879	2909		1	1	2	3757	3787
1	1	1	2968	2998		1	1	2	3863	3893
1	1	1	3006	3036		1	1	2	3936	3966
1	1	1	3362	3392		1	1	2	4200	4230
1	1	1	3638	3668		1	1	2	4297	4327
1	1	1	3641	3671		1	1	3	2756	2786
1	1	1	3711	3741		1	1	3	2890	2920

#### A.2 Data 2

	City 3			City 5			City 9			City 15	
	$\Delta = 30$			$\Delta = 30$			$\Delta = 15$			$\Delta = 45$	
	$\tau = 9131$			$\tau = 12783$			$\tau = 9496$			$\tau = 10592$	
А	, <u>— 5101</u> В	С	А	л <u>– 12</u> ,000 В	С	А	7 <u>— 9490</u> В	С	А	B B	С
1065	4640	1052	358	11444	6156	132	6239	6030	75	7778	4469
2601	4782	2240	649	11524	6417	1026	6410	6195	155	8242	4588
2877	4958	3267	729	11638	6453	1251	6528	6813	685	9418	4831
2908	5122	3310	1038	11674	6813	1502	6804	7001	1344	9495	5210
3091	5259	4569	2223	12364	7138	1930	8139	7693	1490	9588	5331
3147	5367	4845	2604	12610	7507	2338	8312	7807	1547	9806	6170
3371	5625	4040	3672	12010	7943	2500	8305	8400	2130	10122	6831
3553	5736	5171	4005		0007	3107	8449	8617	2103	10122	6008
3808	5848	5871	4033		0887	3170	8513	8638	2220		7201
2004	6282	5065	4903		10005	2521	8612	8038	2092		7611
4632	6663	6240	5274		10596	3778	0135	8825	3473		7806
4032	6729	6717	5300		11027	3014	9133	8844	3710		7044
4720	7201	6941	5252		11/27	2006	9472	8044	4502		7944 9195
5010	7201	7007	5452		11423	4170		0270	4502		0120 9025
5107	8114	7050	5565		19747	4179		0412	5170		8495
5107	0114	1909 9100	5642		12/4/	4245		9413	5026		0420
5220		8202	6127			4211			5230		0844
5339		8293	6107			4330			5344		9044
5462		0094	0197			4373			5479		
5529		8043	5605			4432			5632		
5098		8732	7032			4461			5080		
5718		8703	7130			4510			0998		
5743		8986	7531			4774			6347		
5856			7628			5089			6534		
6368			7754			5140			6793		
6674			7804			5353			6985		
6726			8090			5435			7173		
6860			8595			5472			7449		
6940			8765			6036			7598		
7084			8844			6084			7819		
7326			8897			6659			7897		
7576			9017			6722			8018		
7806			9132			6836			8221		
7853			9204			7128			8351		
8154			9292			7222			8439		
8394			9395			7384			8596		
8463			9621			7404			8908		
8558			9682			7460			8974		
8613			9912			7826			9245		
8640			10158			8129			9716		
8704			10264			8145			9887		
8997			10543			8339			10041		
			10587			8401			10208		
			10633			8448			10274		
			10679			8646			10351		
			10777			8777					
			10871			8855					
			11014			8981					
			11111			9085					
			11180			9489					
			11303								
			11449								
			11506								
			11678								
			11718								
			11769								
			11844								
			11878								
			12196								
			12227								
			12329								
			12373								
			12476								
			12718								

Table A.1: Data set used in Section 4.5, times (in days) are failure times in each component, repair times,  $\Delta$ , are the same among the components within a city

### Appendix B

# Misspecification of $\Delta_B$

$\tau$	$\Delta_{B_0}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 4$	$e^{\beta_A} = 4$
100	$\frac{2}{3}\Delta_B$	0.232	0.230	0.753	0.748
	$\Delta_B$	0.369	0.369	0.996	0.995
	$\frac{4}{3}\Delta_B$	0.237	0.237	0.704	0.700
200	$\frac{2}{3}\Delta_B$	0.401	0.379	0.954	0.958
	$\Delta_B$	0.537	0.518	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.381	0.360	0.925	0.916
500	$\frac{2}{3}\Delta_B$	0.679	0.687	1.000	1.000
	$\Delta_B$	0.885	0.890	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.717	0.736	1.000	1.000
1000	$\frac{2}{3}\Delta_B$	0.901	0.898	1.000	1.000
	$\Delta_B$	0.991	0.991	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.924	0.920	1.000	1.000

Table B.1: Results of the power study when  $\Delta_{B_0}$  is misspecified, where  $\Delta_B = 7$  is assumed

au	$\Delta_{B_0}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 4$	$e^{\beta_A} = 4$
100	$\frac{2}{3}\Delta_B$	0.195	0.199	0.657	0.660
	$\Delta_B$	0.290	0.293	0.995	0.995
	$\frac{4}{3}\Delta_B$	0.198	0.205	0.499	0.503
200	$\frac{2}{3}\Delta_B$	0.333	0.345	0.926	0.928
	$\Delta_B$	0.519	0.535	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.276	0.299	0.790	0.802
500	$\frac{2}{3}\Delta_B$	0.610	0.610	1.000	1.000
	$\Delta_B$	0.882	0.882	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.576	0.577	0.988	0.988
1000	$\frac{2}{3}\Delta_B$	0.884	0.885	1.000	1.000
	$\Delta_B$	0.994	0.994	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.832	0.832	1.000	1.000

Table B.2: Results of the power study when  $\Delta_{B_0}$  is misspecified, where  $\Delta_B = 14$  is assumed

m	$\Delta_{B_0}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 2.5$	$e^{\beta_A} = 2.5$
10	$\frac{2}{3}\Delta_B$	0.887	0.884	0.992	0.992
	$\Delta_B$	0.988	0.988	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.912	0.912	0.994	0.994
20	$\frac{2}{3}\Delta_B$	0.993	0.992	1.000	1.000
	$\Delta_B$	1.000	1.000	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.996	0.996	1.000	1.000
50	$\frac{2}{3}\Delta_B$	1.000	1.000	1.000	1.000
	$\Delta_B$	1.000	1.000	1.000	1.000
	$\frac{4}{3}\Delta_B$	1.000	1.000	1.000	1.000
100	$\frac{2}{3}\Delta_B$	1.000	1.000	1.000	1.000
	$\Delta_B$	1.000	1.000	1.000	1.000
	$\frac{4}{3}\Delta_B$	1.000	1.000	1.000	1.000

Table B.3: Results of the power study when  $\Delta_{B_0}$  is misspecified, where  $\Delta_B = 7$  is assumed

m	$\Delta_{B_0}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$	$\hat{Pr}\{Z > 1.645\}$	$\hat{Pr}\{Z > \hat{Q}_{0.950}\}$
		$e^{\beta_A} = 2$	$e^{\beta_A} = 2$	$e^{\beta_A} = 2.5$	$e^{\beta_A} = 2.5$
10	$\frac{2}{3}\Delta_B$	0.884	0.884	0.996	0.996
	$\Delta_B$	0.990	0.990	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.792	0.792	0.955	0.955
20	$\frac{2}{3}\Delta_B$	0.995	0.995	1.000	1.000
	$\Delta_B$	1.000	1.000	1.000	1.000
	$\frac{4}{3}\Delta_B$	0.972	0.972	0.997	0.997
50	$\frac{2}{3}\Delta_B$	1.000	1.000	1.000	1.000
	$\Delta_B$	1.000	1.000	1.000	1.000
	$\frac{4}{3}\Delta_B$	1.000	1.000	1.000	1.000
100	$\frac{2}{3}\Delta_B$	1.000	1.000	1.000	1.000
	$\Delta_B$	1.000	1.000	1.000	1.000
	$\frac{4}{3}\Delta_B$	1.000	1.000	1.000	1.000

Table B.4: Results of the power study when  $\Delta_{B_0}$  is misspecified, where  $\Delta_B = 14$  is assumed